

7719 2019

Original Version: July 2019 This Version: April 2023

Labour Markets, Trade and Technological Progress. A Meta-Study

Nikos Terzidis, Steven Brakman, Raquel Ortega-Argiles



Impressum:

CESifo Working Papers ISSN 2364-1428 (electronic version) Publisher and distributor: Munich Society for the Promotion of Economic Research - CESifo GmbH The international platform of Ludwigs-Maximilians University's Center for Economic Studies and the ifo Institute Poschingerstr. 5, 81679 Munich, Germany Telephone +49 (0)89 2180-2740, Telefax +49 (0)89 2180-17845, email office@cesifo.de Editor: Clemens Fuest https://www.cesifo.org/en/wp An electronic version of the paper may be downloaded • from the SSRN website: www.SSRN.com

- from the RePEc website: <u>www.RePEc.org</u>
- from the CESifo website: <u>https://www.cesifo.org/en/wp</u>

Labour Markets, Trade and Technological Progress. A Meta-Study

Abstract

Theoretical models, supported by empirical evidence, suggest that technological progress and trade are two essential factors to explain recent labor market developments in many OECD countries; technology can make jobs obsolete, and import competition can drive firms out of business. Both causes are often mentioned in tandem, but their relative contribution is unclear. This meta-analysis disentangles the interplay between technology and trade regarding recent labor market developments. Using a sample of some 623 technology and 1094 trade elasticities from 91 studies, our meta-analysis first reveals that despite small publication selection, technology and trade benefit both wages and employment in a statistically significant and economically meaningful way. Nevertheless, the multivariate meta-regression analysis indicates that this conclusion is conditional on several research dimensions. In the most prominent outcome, we document that the skill-bias impact from technology is concentrated on employment, where high-skilled workers benefit relatively more compared to low-skilled ones. In contrast, trade effects expand over both wages and employment, but mainly benefit high-skilled workers in industrialized labor markets.

JEL-Codes: F160, J310, O110.

Keywords: labor market, trade, technological progress, meta-study.

Nikolaos Terzidis* Faculty of Economics & Business University of Groningen PO Box 800 The Netherlands – 9700 AV Groningen n.terzidis@rug.nl

Steven Brakman Faculty of Economics & Business University of Groningen PO Box 800 The Netherlands – 9700 AV Groningen s.brakman@rug.nl Raquel Ortega-Argiles Regional Economic Development The University of Manchester Oxford Road United Kingdom – Manchester, M13 9PL

*corresponding author

Updated April 2023

1. Introduction

Since the 1980s, it became evident that in industrialized countries, employment of low-skilled workers relative to high-skilled ones declined. Also, the wage of unskilled workers compared to high-skilled labor fell (Katz et al., 1995). These stylized facts initiated a debate on the causes of these developments. Two possible explanations characterize the discussion: international trade and domestic technological progress.⁴

Trade-related explanations stress the growth of import competition from low-wage economies, causing unemployment in import-competing sectors and decreasing wages. Domestic technological growth often has similar effects. If automation replaces workers, the consequences are similar to trade effects and empirically look the same. This raises the question of what factor is the most important.

In the 1980s and 1990s, the dominant explanation was that automation was the main cause. Labor market data, mainly in the US and EU countries demonstrated increasing demand for high-skilled workers, primarily motivated by the introduction of computers and the deepening of capital in production (Jorgenson et al., 2008). Trade only had a smaller impact as exports from low-wage countries were too small relative to GDP to explain the labor market developments (see Krugman, 2008 for a reflection on these developments in the 1990s). The consensus – with qualifications – was that although trade was growing; automation was the main cause of the labor markets' changes in the 1980s and 1990s.⁵

In the early 2000s, stimulated by the Chinese membership of the WTO and the subsequent steep increase in imports from China, the discussion revisited the role of trade on labor market developments. Import competition became more important as an explanation for the transformation of employment. In a seminal paper, Autor et al. (2013) take a closer look at the 1990-2007 period and note that, especially after 2001, imports from China to the US, and also to other industrialized countries increased enormously. Acemoglu et al. (2016) also conclude that import competition is a major source of labor market disruptions, compared to domestic technological progress as a cause. In addition, the technology argument is not supported by

⁴ There are many surveys of the discussion that took place during the 1990s on the effects of globalization and technological change on labor markets; see for instance Lawrence & Slaughter (1993), Wood (1995), Krugman, Cooper & Srinivasan (1995), Sachs & Shatz (1996), Bernard & Wagner (1997), Borjas, Freeman & Katz (1997). This discussion is summarized in Helpman (2018).

⁵ Although there was some consensus on the most important cause of the changes on labor markets, there was debate as to what model is best suited to analyse the developments, see for instance the discussion between Krugman (2000) and Leamer (2000). This discussion is evaluated in Feenstra (2016).

evidence. Gordon (2016) extensively documents that for the US, technological progress has substantially slowed down from a peak in the 1950s to 2014. This slowdown in productivity growth has also been documented for other industrialized countries (see e.g., Cette et al., 2016). Studies like these changed the view on what drives labor market developments, and motivated a shift in the consensus from technology to trade.

In line with the theoretical debate, a growing body of evidence emerged that investigates the labor market effects of automation and trade. The current meta-study takes stock of the extant literature, synthesizes the evidence base and assesses the validity of competing theoretical arguments. Our main analysis builds on elasticities, which measure the responsiveness of labor demand, reflected on either wages or employment, on various measures of technology and trade. The collected samples comprise 644 technology and 1073 trade elasticities from 91 primary studies. Based on those and following Stanley and Doucouliagos (2012), we estimate multivariate weighted least squares meta-regression models (WLS-MRA), which is the most reliable and unbiased methodology to evaluate the empirical evidence in an empirically robust and economically meaningful way.

The current meta-study confirms that publication selection is a major issue (Stanley & Doucouliagos, 2014). Publication selection is the tendency to report significant results, which are in line with the conventional theory. Therefore, our analysis controls for publication bias.

In general, we document uniform positive effects from trade and technology on both wages and employment. The key outcome of our analysis sheds further light on the skill-biased nature of globalization. We reveal that technology benefits the employment of high-skilled workers relatively more compared to low- and medium-skilled ones. In contrast, the skill-biased nature of trade expands on both wages and employment. Specifically, we conclude that trade benefits the wages and the employment of high-skilled workers more compared to the low-skilled ones. Furthermore, we find that the differential effects of trade and automation also depend on various other dimensions, such as the period of analysis, the geography, the sector, the spatial unit and various methodological choices.

The remainder of the paper is organized as follows. Section 2 briefly discusses the theoretical background of the various channels linking trade and technology with labor market developments, which motivates the empirical part of the current meta-analysis. Section 3 explains the way we summarized the literature, separated into the selection of the primary

studies and the functional forms of the estimated equations from where we extracted the technology and trade elasticities. In Section 4, we explore the total variation in the collected samples disaggregated into its main categories while we also provide preliminary evidence of publication selection. Section 5 discusses the empirical setting of the multivariate meta-regression analysis and reports the empirical outcomes. Finally, Section 6 concludes.

2. Labor market effects from trade and technological progress: theoretical approaches.

Trade theory traditionally relies on the Stolper-Samuelson (S-S) theorem to explain relative factor price movements. The theorem states that a relative price increase benefits the production factor that is intensively used in producing that commodity. If this production factor is high-skilled labor, then a price increase in high-skilled intensive products raises the real wage of high-skilled workers and reduces the real wage of low-skilled workers; wage adjustments are larger than price ones, which is the so-called *magnification effect* (see Jones, 1965). In principle, this theorem explains the change in the skilled workers' real wage premium in industrialized countries. The flip side of this reasoning is that a wage premium for low-skilled workers develops in low-skilled-worker-abundant countries.

According to the Heckscher-Ohlin-Samuelson (HOS) model, these price changes also have consequences for the relative factor use; sectors economize on the production factors that have become more expensive. In industrialized countries, the share of high-skilled relative to low-skilled workers should decline, and the opposite should happen in low-skilled abundant countries. However, the reverse happened in many countries; high-skilled workers' employment relative to low-skilled ones increased in the 1980s, despite higher relative wages, pointing towards a higher demand for skilled labor. A possible explanation for this phenomenon is that because of technological advances, the demand for high-skilled workers increased because production became more skill-intensive. Feenstra (2016, p.99), however, forcefully points out that in the late 1990s, developments of relative employment of high- versus low-skilled workers are consistent with what one would expect from trade theory; a relative decline of high-skilled employment. Therefore, the combined evidence concerning relative employment developments in the 1980s and 1990s is mixed. On the one hand, it points towards

a higher demand for skilled workers due to technological progress. On the other hand, it is consistent with trade-related development, as Feenstra (2016) pointed out⁶.

For the HOS model to work, however, the embodied factor supplies in trade flows should be substantial enough to have economy-wide effects. For industrialized countries, this implies that the low-skilled labor embodied in imports should be considerable. Borjas et al. (1997) and Krugman (2000) found that in the 1980s and 1990s, the import flows were too small to explain developments in the US labor market. Berman et al. (1998) refer to similar findings and indicate the importance of technological innovations as an alternative explanation for the labor market developments. Together, the evidence does not point to trade as the primary explanation. During the 1980s and early 1990s, the consensus developed that technological change was the main driver to explain relative changes in high-skilled workers' position versus low-skilled ones in labor markets.

The reasoning behind technological change and developments in labor markets is straightforward. A typical effect of technological change in most high-income countries is that when occupations become more skill-intensive, there is an increased demand for higher skilled workers. If demand increases more rapidly than the supply of high-skilled workers, high-skilled workers' relative wages increase. Furthermore, technological progress might make low-skilled jobs obsolete. Therefore, the wages of skilled workers increase as well as the share of skilled employment. This reasoning depends on the type of technological progress (see Helpman, 2009 for a survey)⁷. The discussion between Krugman (2000) and Leamer (2000) on the different consequences of factor-related versus sector-biased technological progress is a case in point. Leamer (2000) assumes the small country case where prices are fixed and the sector bias of technological progress determines relative wages, whereas Krugman (2000) discusses the

⁶ Helpman (2018) evaluates the role of modern developments in trade theory on increasing income inequality. These models are based on the concepts of assignment and matching between firms and workers (see Costinot and Vogel, 2010). Trade re-shuffles workers and improves matching. In skill-abundant countries, this benefits skilled workers.

⁷ Technological progress is *high-skill augmenting* if it makes high-skilled workers more productive. This is comparable to an increase of high-skilled workers at the old level of technology, similarly, for *low-skill augmenting* technological progress. If technological progress augments both factors equally, it is *neutral technological progress*. Depending on which type is stronger, it can change relative factor rewards. Technological progress can also replace production factors (*process innovation*). For example, a reorganization of the production process can economize on low-skilled workers while keeping output constant. Technological progress can also have a *sector* dimension, benefiting one sector (including all factors of production) more than other sectors. The consequences for each type of production factor thus depend on the type of technological progress.

large country case, where prices change and only the factor bias of technological change matters. Empirically it is challenging to differentiate between these two cases.

Furthermore, it is not easy to distinguish offshoring from automation. The theoretical model of Grossman and Rossi-Hansberg (2008) illustrates this. In this model, offshoring of low-skilled activities has three effects: first, it decreases production costs and lowers prices as it is low-skilled worker saving and more low-skilled workers become available on the labor market; the model also predicts a productivity-enhancing effect because, overall, the remaining tasks are higher skill intense. The first two effects are similar to the ones discussed above concerning trade and technological progress. The third effect is a new element in their model. According to this, offshoring acts like a low-skilled biased technical change that makes workers more productive and increases wages. However, the excess supply of workers acts in the opposite direction. Wright (2014) finds a small net reduction in employment due to offshoring (see also Hummels et al., 2018).

Overall, technological progress can add, reverse, or substitute for the effects of trade on factor markets, which explains to some extent the conflicting evidence on the relative importance of trade versus technological progress (see also Jones, 1965). The conclusion of Feenstra (2016, p.90) is, therefore, something to be expected. After reviewing the evidence for the 1980s and 1990s, he concludes that estimates are '*quite sensitive to the data used and the specification of the regression*.' However, the literature favors technology above trade as the primary driver of adverse labor market developments in the 1980s and early 1990s.

The focus of the literature changed when the first empirical evidence became available on the impact of China joining the WTO in 2001. The seminal article of Autor et al. (2013) analyses the consequences of the China shock on the US labor market. They found that an increase in import exposure of \$1000 per worker over a decade reduces manufacturing employment (working population) by 0.596 percentage points. Between 1990 and 2000, the import penetration changed by \$1140 per worker and between 2000 and 2007 by \$1839 per worker. In the first period, manufacturing employment fell by 0.68 percentage points and in the second period by 1.1 percentage points. Acemoglu et al. (2016), using a different methodology, found that between 1999 and 2011, job losses in the US amounted to 2.4 million. They estimated that if import penetration from China had not increased after 1999, the number of manufacturing jobs would have been 560.000 larger (p.144). Dauth et al. (2014) find similar but smaller effects

for Germany. Again, as the review of Hummels et al. (2018) points out, the trade effects are difficult to disentangle from technology.

As briefly discussed, the evidence is mixed and changes over time. This illustrates the usefulness of conducting a meta-study, to resolve the theoretical conundrum and the conflicting empirical evidence. The following section describes how we construct a representative sample of the evidence base. The subsequent analysis evaluates the sample, and reveals whether technology and/or trade impose significant labor market outcomes beyond publication selection. Furthermore, we establish 'best practice estimates' and identify how various aspects of study design shape the reported outcomes.

3. Labor market effects from trade and technological progress: systematizing the empirical evidence

The current section discusses the process of summarizing the literature. First, we outline the selection of studies, based on a set of pre-determined criteria (section 3.1). Second, in section 3.2, we describe the functional forms that are used in the primary studies where we extract the labor market elasticities from technology and trade.

3.1 Selection of studies

Figure 1 illustrates four phases of our literature search (Appendix A for a detailed outline of the search strategy). We follow the PRISMA flow diagram methodology and the recommended reporting guidelines as suggested in the *Meta-Analysis of Economics Research Network's* (MAER-Net), (see Moher et al., 2009 and Havranek et al., 2020, for further descriptions).⁸

In the identification stage, we conducted a full-text search of multiple search queries (Appendix A1 – Table A1.1) encompassing an extensive vector of relevant terms in three cross-disciplinary databases (EconLit, Web of Science and Google Scholar). To overcome the concerns of literature searches such as academic databases' biases towards studies published in peer-reviewed journals, we added two steps (Petticrew & Roberts, 2008). First, we used snowballing techniques by scanning relevant literature reviews (Klein et al., 2003; Ugur et al., 2018).

⁸ The PRISMA flow diagram shows the details of information flow in each stage of literature search in metaanalysis, including the number of studies identified, screened and deemed eligible. Furthermore, the current metaanalysis complies with the most recent revision of MAER-Net reporting guidelines, obtained from: https://www.maer-net.org/post/revision-of-reporting-guidelines.



Figure 1. Overview of the literature search

Secondly, we extended our search to reputable working paper series (among others, NBER, IZA, CESifo) and the grey literature (departmental working paper series) following notifications on our search queries on the most recently available studies. We concluded the identification stage of the literature search January 2022 with a total of 1747 primary studies.

In the second stage, we screened the selected studies as follows. First, inspection of the titles and abstracts verified that the selected studies were empirically investigating the labor market effects (employment or wages) of technology, trade or both in a high-income economy.⁹ Next,

⁹ High-income countries are determined based on their gross national income (*Country and Lending Groups taxonomy*, The World Bank), which defines high-income countries as those with per capita GNI more than \$12.056.

a full-text search established that the estimated elasticities were based on well-established econometric techniques determining correlation or causation and that the authors reported all the necessary information (detailed empirical model, estimated effect, measures of precision, number of observations etc.).

In the third phase, we considered the eligibility of the selected studies based on our predetermined selection criteria (Appendix A1 - Table A1.2). After removing duplicates and adding 2 additional studies based on recommendations from external experts, we finalized the list of 91 primary studies, which formed the basis of our quantitative analysis. Out of the 91 selected studies (Appendix A2), 73 are published in peer-reviewed journals, including *The Quarterly Journal of Economics, American Economic Review, Review of Economics and Statistics* and *Journal of Political Economy*. In addition, 17 studies are working papers, while 1 is a book chapter.¹⁰ Publication years span from 1991 to 2022 with the median study published in 2013. So, half of the studies are published within the last 10 years, indicating that our metaanalysis covers an active research area.

To minimize selection bias and to use all the available information, we followed standard meta-analytic recommendations and extracted the maximum possible number of estimates from each study (Doucouliagos & Stanley, 2009).¹¹ On average, we collected 19 estimates from each primary study, referring to a period of analysis from 1974 to 2020 and a large selection of industrialized countries (the majority referring to the US, the UK, Germany and Sweden, possibly reflecting data availability).

The current study draws on a comprehensive set of (semi-) elasticities, obtained from double-log models (Section 3.2 for a detailed discussion on the identification of the elasticities). Elasticities are typical measures of economic effects and capture the percentage change in either employment or wages, stimulated by a percentage change in the technology or trade practice. Elasticities are directly comparable; therefore, they are standard measures in meta-analyses (Doucouliagos & Stanley, 2009; Melo et al., 2009). Nevertheless, to increase the degrees of freedom for the multivariate MRA, we also include semi-elasticities, which might raise

¹⁰ To test whether leading journals report more precise estimates and thus they should only be included in the metaanalysis, we followed Stanley and Doucouliagos (2012) and regressed the precision of the collected elasticities on the indicator of journal quality used in the MRA (results are reported in Appendix B – Table B1). We found no evidence that journal quality is an important conditioning factor of the precision of the estimates, thus validating our choice to include elasticities from working papers and the grey literature.

¹¹ On the downside, multiple estimates from the same data are not independent from each other, biasing results (Weichselbaumer & Winter-Ebmer, 2005). To correct for this, we cluster the standard errors by primary study.

comparability concerns. Sensitivity analyses (Appendices D2 and D3) illustrate the robustness of the main results across different samples, while in Appendix F, we construct partial correlation coefficients, which ensure perfect comparability and re-iterate the core analysis.

3.2 Identifying the labor market elasticities from technology and trade

This section discusses the functional forms of the empirical specifications in the primary studies whose parameters are the extracted labor market elasticities from technology and trade. Empirical evidence at the firm level typically follows a labor demand approach (Hijzen, 2005). This reflects a partial equilibrium framework, where labor supply is assumed to be perfectly elastic and the representative firm decides on the optimal labor demand (L_D) for a given level of output and wages, by minimizing a quasi-fixed cost function (Eq. 1).¹²

$$C_w = L_D(w, x, z) \tag{1}$$

Firm-level costs (C_w) are a function of factor prices, such as wages (w), output and fixed input quantities (x) as well as a vector of demand shifters such as technological capital and international trade (z). Applying Shephard's lemma, and assuming labor mobility across firms and firm-specific capital, gives the conditional factor demand in log-linear form (Eq. 2):

$$LnL_{it} = \alpha_0 + \beta_1 lnY_{it} + \beta_2 lnw_{it} + \beta_3 lnK_{it} + \beta_4 lnTechn_{it} + \beta_5 lnTrade_{it} + X'_{it}\gamma + \varepsilon_{it}$$
(2)

where LnL_{it} is the logarithm of total employment, lnY_{it} is the logarithm of output, lnw_{it} is the logarithm of the wage rate, lnK_{it} is a measure of capital, $lnTechn_{it}$ represents a cluster of technology variables reflecting inputs, such as R&D expenditure or R&D personnel, or a proxy of output, such as patents, see also Blechinger et al., (1998). Furthermore, $lnTrade_{it}$ includes imports or exports of final products, or intermediates. X_{it} is the vector of control variables, consistent with Mincerian wage equations (Mincer, 1974). Coefficients β_4 and β_5 reflect the conditional labor demand elasticities from technology and trade respectively, since they depend on a certain level of output (Crino, 2009). Winkler (2010) notes that Eq. (2) only considers the

¹² Alternatively, by assuming that firms minimize their costs for the production of output, a similar (ordinary) labor demand equation is derived from maximizing profits from a constant elasticity of substitution (CES) production function, augmented with some exogenous technology and trade measures (see Van Reenen, 1997) for the detailed derivation). The main difference between the conditional and the ordinary labor demands is that price effects in the former just capture substitution effects, while in the latter they also capture the effect on the optimal output level (Aguirregabiria & Alonso-Borrego, 2001)

productivity and substitution effects from technology and trade. To allow for scale effects, the authors often drop output; thereby estimating the unconditional labour demand equation.

Equation (2) is a standard version of employment, but often variants are estimated (Hijzen, 2005). First, industry-level studies drop the notion of the representative firm. Industries differ with respect to technology adoption or trade dependence. Second, the supply of labor is not perfectly elastic. A shock in labor demand, due to technology or trade, will be reflected in: changes in the relative output, the relative wages or labor utilization. The elasticity with respect to each channel depends on the flexibility of the labor market; in case wages and output do not fully adjust to clear the labor market, a demand shock will be reflected via a change in labor utilization.

On the empirical front, applied economists often use a translog functional form. This further departs from the standard labor demand approach, since labor demand is substituted by the cost share of labor, derived from a GDP function, as the dependent variable (Berman et al., 1994, and Feenstra, 2016). Such models (Eq. 3) observe relative labor demand changes after wages and employment have adjusted.¹³

$$LnWSH_{it} = \alpha_0 + \beta_1 lnY_{it} + \beta_2 lnw_{it} + \beta_3 lnK_{it} + \beta_4 lnTechn_{it} + \beta_5 lnTrade_{it} + X'_{it}\gamma + \varepsilon_{it}$$
(3)

Where $LnWSH_{it}$ is the logarithm of the wage share of some category of workers (i.e., low-, medium- or high-skilled) in the total wage bill, while all the other variables are similar to Eq. (2). In the above setting, $\beta_4 > 0$ ($\beta_5 > 0$) is tantamount to technology (trade) shifting outwards the relative labor demand for the investigated group of workers.

Utilizing the translog framework, both changes in the factor prices and/or factor quantities explain changes in the wage share. So, applications with the wage share as the dependent variable are often used in more flexible labor markets (like the US) where wages are considered the main adjustment mechanism. In contrast, authors often switch to relative employment as the dependent variable in more rigid labor markets (like Continental European countries), as in Eq. (4) (Strauss-Kahn, 2004). The latter approach has the additional merit of solving potential wage endogeneity concerns (Piva et al., 2005).

¹³ In many cases, authors differentiate between wages for unskilled, semi-skilled and skilled workers, which results in a system of labor demand equations, one for each type of labor (see f.i., Ekholm & Hakkala, 2006).

$$LnEmplSh_{it} = \alpha_0 + \beta_1 lnY_{it} + \beta_2 lnw_{it} + \beta_3 lnK_{it} + \beta_4 lnTechn_{it} + \beta_5 lnTrade_{it} + X'_{it}\gamma + \varepsilon_{it} \quad (4)$$

Where $LnEmplSh_{it}$ is the logarithm of the employment share of some category of workers (i.e., the high-skilled ones) and all the other regressors are similar to Eq. (2). Like before, technology and trade benefit the relative labor demand of the investigated category of workers if the estimated parameters β_4 and β_5 are positive.

Finally, Feenstra & Hanson (1999) estimate the long-run wage effects from structural variables such as technology or trade through the inspiring *mandated wage regressions* framework, which exploits the industry-specific zero-profit conditions. This two-stage methodology starts by regressing the chosen approximations of technological capital and outsourcing (structural variables) on industry prices and productivity. Notably, the latter are treated as endogenous, thereby allowing for a large-open economy setting. In the second stage¹⁴, the components of the price and productivity changes are regressed against the factor shares, resulting in a price regression where the estimated coefficients reflect the changes in factor prices (such as wages), assuming that the structural variables were the only sources of changes in value added and productivity. Such mandated wage regressions are estimated for each structural variable separately.

Having described the selection of the primary studies and the identification of the extracted elasticities, the following section offers two bodies of preliminary evidence. Section 4.1 disentangles the overall variation in our samples into its two main components: *sampling error* and *excess heterogeneity*. Section 4.2 evaluates the presence of publication selection.

4. What explains the differences in the elasticities of technology and trade

4.1 Summary statistics

Table 1 presents some (unweighted) descriptive statistics. In fact, we assume the absence of excess heterogeneity between the collected estimates, besides the one due to sampling error. Figure 2 illustrates the probability density functions of the labor market elasticities from technology and trade.

¹⁴ The second stage regression is a modified version of the price regression (Baldwin & Hilton, 1984; Leamer, 1994, 2010), where the change in industry prices is regressed against the cost share of each factor. Nevertheless, Feenstra & Hanson (1999) argue that -once fully specified- the price regression turns into an identity; therefore, it is unable to predict any changes in the factor prices besides the ones that actually materialized (see Feenstra & Hanson, 1999 for a detailed discussion).

	Obs.	Mean	St. Dev.	Iqr	Skewn.	Kurt.	I^2	Q-test
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Technology elasticities	623	-0.021	0.815	0.157	1.855	18.21	100%	0.000***
• Wage elasticities	231	-0.211	0.912	0.797	-0.282	8.236	100%	0.000***
• Employment elasticities	392	0.091	0.732	0.088	4.673	29.42	100%	0.000***
Trade elasticities	1094	0.003	7.203	0.105	-7.521	183.7	99.9%	0.000***
• Wage elasticities	459	0.128	10.63	0.066	-5.789	92.35	99.9%	0.000***
• Employment elasticities	635	-0.088	2803	0.208	6.367	86.39	100%	0.000***

Table 1. Summary statistics

Note: Column 4 reports the interquartile range (the difference between the 75th and the 25th percentile). Column 7 reports the percentage of the between-estimate heterogeneity, besides the one attributed to sampling error. Column 8 reports the p-value for the χ^2 Cochran's test under the null hypothesis of no between-estimate heterogeneity. * for p < 0.1, ** for p < 0.05 and ** * for p < 0.0

Figure 2. Kernel density distributions



Interestingly, preliminary comparisons reveal differential labor market effects from technology and trade. Specifically, Table 1 (Column 2) illustrates that technology imposes negative effects on average (-0.021) as opposed to positive ones from trade (0.003); nevertheless, both effects are small, based on the partitions suggested by Doucouliagos (2011)¹⁵.

¹⁵ Doucouliagos (2011) suggests that effects smaller than +/- 0.07 can be regarded as small, effects between +/- 0.07 and +/- 0.33 are moderate, while effects larger than +/- 0.33 indicate strong association.

Once we distinguish between employment and wage elasticities, we conclude that the overall negative impact of technology includes two opposing components; a negative wage elasticity (-0.211) and a much smaller in absolute value positive employment elasticity (0.091). Regarding trade, the average positive impact includes a positive effect on wages (0.128) and a negative – as well as smaller in absolute value- effect on employment (-0.088).

Furthermore, the collected technology and trade elasticities exhibit substantial dispersion, as verified by their large standard deviations compared to the average effects, the very high Kurtosis values and the heavy tails in the kernel density distributions (Figure 2). Nevertheless, the kernel density distributions exhibit high peaks around zero, which is called the genuine effect¹⁶ (Fisher, 1932; Pearson, 1904). Taken together, the skewness coefficients from Table 1 and the kernel density figure reveal strongly asymmetrical distributions, both positive and negative. This corroborates our expectations that the theory predicts results of both signs, which are also observed in the empirical findings.

More detailed, weighted analysis of the underlying parameters of interest, using the inverse of the variance as the optimal weights (Hedges & Olkin, 1985) is reported in the last two columns of Table 1 (weighted averages based on fixed and random effects models are reported in Appendix B – Table B2). The very high I^2 coefficient illustrates that excess heterogeneity explains almost the entirety of the overall variation. Similarly, Cochran's Q test always rejects the null hypothesis of within-estimate homogeneity. Taken together, the above highlight the importance of excess heterogeneity (Higgins et al., 2003) and the minimal role of sampling error in the economics literature (Disdier & Head, 2008). Therefore, a research-driven meta-analysis should provide evidence based on multivariate meta-regression models, controlling for the precision of the estimates and augmented with multiple moderator variables, reflecting the numerous sources of variation (Stanley & Doucouliagos, 2012). In the remainder of this section, we discuss the coded sources of variation, together with some simple averages. These will be used in the analysis that will uncover the impact from heterogeneity, which is the focus of Section 5.2.

¹⁶ The term *genuine* or *overall* effect comes from R.A. Fisher and Karl Pearson who independently developed statistical methods to summarize the average effect from multiple tests, where simple hypothesis testing investigates the null hypothesis that the overall effect is zero (Stanley & Doucouliagos, 2012). However, rejecting the above null hypothesis is not sufficient evidence of a genuine effect; instead, it implies either publication bias or excess heterogeneity. The value added of the current multivariate meta-regression analysis (Section 5.2) lies in ascertaining the presence of a genuine effect once we control for publication selection and use a comprehensive set of moderator variables to approximate excess heterogeneity.

To account for the excess heterogeneity in our samples, we code a series of characteristics of the estimated elasticities. The so-called moderator variables distinguish between the *methodological* (Table 2) and the *structural* (Table 3) dimensions of the study design. The former include the data, estimation and publication characteristics like -among others- the functional form of the econometric model, the estimation technique or the data aggregation while the latter mainly refers to the country, the period or the spatial unit of the analysis¹⁷.

Columns 1-2 show statistics related to technology, whereas Columns 3-4 present those for trade.¹⁸ There are several noteworthy outcomes. Regarding technology, we first show that product innovation tends to impose positive effects on average as opposed to negative ones from process innovation. Somewhat surprisingly, trade in intermediates seems to benefit wages and employment in the industrialized countries, even though it is mostly related to vertical specialization patterns. This potentially reflects earlier findings associating trade in intermediates with employment growth in the US (Amiti & Wei, 2009). In contrast, the negative average wage and employment elasticities of trade in final goods are in line with the job destruction outcomes (Biscourp & Kramarz, 2007).

Furthermore, Table 2 offers preliminary evidence on the skill-biased nature of technological innovations (Autor et al., 2003). As verified by Figure 3, the average elasticities are negative for low- and medium- skilled workers, and positive for high-skilled ones. A similar pattern is largely visible on trade elasticities (Figure 4, small discrepancies on the mean values are explained by the exclusion of values outside the boxplots in the Figures, for better illustration). The positive effects from trade seem to be concentrated on high-skilled labor. As a result, the preliminary analysis reveals considerable overlap in the skill-biased nature of technology and trade.

¹⁷ Besides the discussed characteristics of the primary study, the labor market effects from automation and trade also depend on country-specific conditions, such as GDP per capita, unemployment or employment protection regulation. Although the impact from such factors is included in the multiple fixed effects we are using, we augment our multivariate MRA models with relevant variables, thus increasing the value added of the current meta-study (Stanley & Doucouliagos, 2012). Nevertheless, some of the collected elasticities refer to multiple countries or go back over 40 years, resulting in many missing observations in the additional variables, which decreases the sample size and threatens the validity of the analysis. Furthermore, the added variables are typically collectively insignificant. As such, they are eliminated in the general-to-specific approach (see Section 5.1 for details). Those models are not reported for brevity.

¹⁸ The tables also show the structure of the moderator variables. The continuous variables included in the regression analysis (number of explanatory variables, the ten-year recursive impact factor, publication year and number of years in the dataset) refer to the entire sample and therefore are omitted for brevity.

Table 2. Methodological heterogeneity

		Technology ef	fects (N = 623)	Trade effects (N = 1094)	
		Wage	Employment	Wage	Employment
Variable	Description	Mean	Mean	Mean	Mean
		(st. dev.)	(st. dev.)	(st. dev.)	(st. dev.)
		(1)	(2)	(3)	(4)
Panel A - Data cha	vracteristics				
Nature of innovatio	n (reference category: process innovation)				
Product	The technology variable refers to product innovation	0.066 (0.81)	0.194 (0.78)	n.a.	n.a.
Process	The technology variable refers to process innovation	-0.422 (0.93)	-0.095 (0.53)	n.a.	n.a.
Nature of trade poli	cy (reference category: trade in final goods)				
Intermediates	The trade variable refers to trade in intermediate products	n.a.	n.a.	0.322 (14.6)	0.035 (3.69)
Final	The trade variable refers to trade in final products	n.a.	n.a.	-0.092 (0.65)	-0.258 (0.95)
Data aggregation (r	eference category: worker-level)				
Industry	The effects are estimated at the industry level	0.078 (1.04)	-0.010 (0.57)	0.055 (17.7)	-0.040 (3.30)
Firm	The effects are estimated at the firm level	0.056 (0.13)	0.224 (0.88)	3.341 (8.24)	-0.130 (0.62)
Worker	The effects are estimated at the worker level	-0.400 (0.80)	0.006 (0.15)	-0.044 (0.55)	-1.759 (3.36)
Sectoral impact (rej	ference category: economy-wide)				
Manufacturing	The estimates refer to the manufacturing sector	0.001 (0.58)	0.174 (0.80)	0.171 (2.36)	-0.225 (2.63)
Services	The estimates refer to the services sector	0.092 (0.09)	0.042 (0.14)	0.001 (0.05)	0.007 (0.21)
Economy-wide	The estimates refer to the entire economy (no sectoral disaggregation)	-0.445 (1.13)	-0.057 (0.53)	0.178 (24.6)	0.634 (6.81)
Skill-level of the w	orkforce (reference category: varied)				
Low	The estimates refer to low-skilled workers	-0.205 (0.49)	-0.204 (0.67)	0.829 (7.30)	-0.864 (3.02)
Medium	The estimates refer to medium-skilled workers	-0.220 (0.34)	-0.266 (0.28)	-0.588 (13.0)	-0.940 (3.30)
High	The estimates refer to high-skilled workers	0.371 (0.78)	0.565 (1.12)	0.292 (19.7)	1.311 (6.73)
Varied	The estimates refer to workers of various skill levels	-0.603 (1.04)	0.091 (0.67)	-0.076 (0.63)	-0.189 (0.88)
Panel B - Estimation	on characteristics				
Estimated model (r	eference category: technology only or trade only)				
Technology	The estimated model includes only technology measures	-0.156 (0.86)	0.008 (0.12)		

Trade	The estimated model includes only trade measures	0 226 (0 02)	0.207(1.00)	0.174(2.44)	-0.194 (1.91)			
Type of analysis (ref	the estimated model metudes both technology and trade measures -0.220 (0.92) 0.207 (1.09) 0.101 (17.0) 0.110							
Time series	The estimates are based on time-series analysis	-0 130 (0 96)	0 133 (0 68)	0 208 (2 15)	-0.091 (0.48)			
Cross-sectional	The estimates are based on cross-sectional analysis	-0.264 (0.87)	-0.036(0.81)	-0.28 (20.8)	-0.089 (8.31)			
Time dependence (<i>re</i>	ference category: contemporaneous)	0.201 (0.07)	0.000 (0.01)	0.20 (2010)				
Lagged	The estimates are based on lagged explanatory variables	0.051 (0.06)	0.030 (0.10)	-0.005 (0.14)	-0.046 (0.52)			
Contemporaneous	The explanatory and dependent variables are contemporaneous	-0.143 (0.99)	0.191 (0.84)	0.262 (2.41)	-0.103 (0.47)			
Form of the dependent	nt variable (reference category: first-differenced dependent variable)							
Level	The dependent variable is in levels	0.061 (0.75)	0.205 (0.85)	0.225 (12.0)	0.060 (2.71)			
First-differences	The dependent variable is first differenced	-0.786 (0.94)	-0.050 (0.46)	-0.189 (1.02)	-0.237 (2.89)			
Estimation method (reference category: non-endogenous)								
Endogeneity	The endogenous nature of technology and trade is addressed	-0.101 (0.77)	0.084 (0.59)	-0.039 (17.2)	-0.011 (3.61)			
Non-endogenous	The endogenous nature of technology and trade is not addressed	-0.244 (0.95)	0.096 (0.79)	0.259 (2.54)	-0.200 (0.90)			
Type of labor demand	d (reference category: absolute)							
Relative	The dependent variable is in shares	0.077 (0.73)	-0.088 (0.58)	0.082 (21.2)	-0.007 (4.70)			
Absolute	The dependent variable is not in shares	-0.447 (0.97)	0.165 (0.75)	0.167 (2.20)	-0.107 (2.26)			
Type of employment	(reference category: number of workers)							
Hours	Employment is measured in terms of hours worked	n.a.	-0.131 (0.70)	n.a.	0.950 (12.2)			
Workers	Employment is measured in terms of the number of workers	n.a.	0.139 (0.71)	n.a.	-0.117 (2.05)			
Inclusiveness of the e	estimated model (reference category: non comprehensive)							
Comprehensive	The estimated model includes any of the following: capital, output,	-0.113(0.92)	0.129(0.77)	0.138(10.9)	-0.091(2.88)			
Comprehensive	value-added, lagged dependent variable or fixed effects	-0.113 (0.92)	0.129 (0.77)	0.156 (10.7)	-0.071 (2.00)			
Non comprehensive	The estimated model omits important variables or fixed effects	-0.951 (0.36)	-0.164 (0.28)	-0.029 (0.16)	-0.145 (0.71)			
Panel C - Publication	n characteristics							
Type of primary stud	y (reference category: working paper)							
Peer reviewed	The primary study is published in peer-reviewed journal	-0.403 (0.92)	0.107 (0.74)	0.166 (2.35)	-0.133 (2.32)			
Working paper	The primary study is a working paper	0.102 (0.79)	0.033 (0.61)	0.105 (18.3)	0.062 (4.11)			

Table 3.	Structural	heterogeneit	ty
----------	------------	--------------	----

		Technology eff	ects (N = 623)	Trade effects (N = 1094)		
		Wage	Employment	Wage	Employment	
Variable	Description	Mean	Mean	Mean	Mean	
		(st. dev.)	(st. dev.)	(st. dev.)	(st. dev.)	
		(1)	(2)	(3)	(4)	
Panel A – Domestic cou	untry characteristics					
Spatial unit of analysis (reference category: national-level)					
Local	The effects are estimated at the regional level	-1.16 (0.91)	-0.142 (0.79)	-0.592 (1.52)	-1.482 (2.61)	
National	The effects are estimated at the national level	0.038 (0.73)	0.158 (0.24)	0.205 (11.2)	-0.045 (2.80)	
Geography classification	n (reference category: US, UK, Canada or Japan)					
Europe	The estimates refer to European economies	0.058 (0.76)	0.065 (0.43)	0.014 (12.5)	-0.091 (2.56)	
US-UK-Canada-Japan	The estimates refer to the US, the UK, Canada or Japan	-0.464 (0.97)	0.118 (0.95)	0.429 (3.42)	-0.096 (3.27)	
Panel B – Destination c	ountry characteristics					
Development status of th	ne trade partner (reference category: not specified trade partner,)				
Developed	The trade effects refer to trade with developed countries	n.a.	1.495 (0.05)	0.130 (6.19)	-0.028 (1.97)	
Developing	The trade effects refer to trade with developing countries	-0.901 (1.05)	0.440 (1.68)	0.362 (18.2)	-0.347 (6.29)	
Not specified	The development status of the trade partner is not specified	-0.048 (0.79)	0.019 (0.36)	0.010 (0.43)	-0.031 (0.44)	
Panel C – Time charact	eristics					
Midpoint of the analysis	(reference category: before 2001)					
After 2001	The midpoint of the dataset is after 2001	-0.124 (0.44)	0.066 (0.43)	-0.090 (0.78)	-0.084 (2.88)	
Before 2001	The midpoint of the dataset is before 2001	-0.236 (1.01)	0.107 (0.87)	0.221 (12.7)	-0.102 (2.81)	



Figure 3. Technology elasticities by skill type





(b) Employment elasticities

Note: Authors' derivation. Values outside the boxplot are excluded for better illustration.



Figure 4. Trade elasticities by skill type





(b) Employment elasticities *Note:* Authors' derivation. Values outside the boxplot are excluded for better illustration. Regarding the estimation characteristics (Table 2 - Panel B), we conclude that -on averagethe reported wage and employment elasticities from both technology and trade are positive when the dependent variable is in levels and negative when first-differenced. First differentiation deals with the omitted variable bias by removing the unobserved, time-invariant component of the error term (Wooldridge, 2015). In contrast, they are often associated with attenuation bias (Ugur et al., 2018). The different outcomes with respect to levels and first differences indicate that the unobserved, time-invariant component of the error term tends to bias the reported elasticities upwards.

Finally, the publication type (Table 2 - Panel C) influences both the sign and the size of the reported elasticities. First, estimates from peer-reviewed journals tend to be larger in absolute value compared to those reported in working papers. Nevertheless, the preliminary analysis indicates both negative and positive elasticities reported in peer-reviewed journals, as opposed to only positive ones published on working papers. This indicates that publication bias is less concerning in our samples.

Regarding the various categories of structural heterogeneity (Table 3), the analysis highlights the importance of geography (Table 3 – Panel A). First, regional effects are on average negative in all the collected samples, potentially reflecting that technology and trade tend to disfavor employment and wages at the regional level. In contrast, at the national-level the evidence tends to be positive, with the notable exception of the employment elasticities from trade. However, these national effects are much smaller in absolute value than regional ones; technology and trade effects tend to be more harmful at the regional level.

Second, -with the exception of wage effects from technology- the average effects from technology and trade point to the same direction even in countries with very different characteristics, such as the relatively rigid European labor markets and more flexible ones, like the UK, the US, Canada or Japan (Geishecker, 2008). Beyond the similarity in the direction of the effects, the average elasticities tend to be larger in the more flexible labor markets. This is consistent with the wage elasticities, since wages are expected to be the main adjustment mechanism in more flexible labor markets.

Finally, Table 3 – Panel C, reveals similarities in the evolution of the estimated effects over time. The technology elasticities –either negative (wage) or positive (employment)- seem to increase in absolute value over time. The same happens in the employment effects from trade.

Nevertheless, the average wage elasticities from trade are negative till the early 2000s, when they turn positive.

The data presented in this section indicate many sources of heterogeneity in the collected estimates. Section 5.2 reports the outcomes from the multivariate meta-regression analysis, which is more robust evidence regarding the above preliminary trends.

4.2 Publication bias

A well-known problem in the economics literature is possible publication selection (Doucouliagos & Stanley, 2013). This reflects the referees' or authors' tendency to report significant results consistent with theory while suppressing unfavorable outcomes (Stanley, 2008). Within our context, publication bias is less concerning, since both positive and negative labor market effects from technology and trade are theoretically possible and thus reported. Nevertheless, publication bias is a possible threat to the statistical inference and should be taken on board (Stanley & Doucouliagos, 2012).

The current study analyzes publication bias as follows. First, we display suggestive evidence employing funnel plots, which are scatter diagrams of the empirical estimates against their corresponding measure of precision (st. error). The basic idea is straightforward. High precision elasticities with low st. error (in the bottom part of the graphs) should reflect the weighted average (solid vertical line) and be concentrated around it. In contrast, less precise estimates (in the top part of the graphs) are expected to be relatively equally spread around the weighted means, providing an overall funnel shape reflecting the absence of a systematic relationship between the estimated effect and its' standard error. Second, we perform the funnel asymmetry test (Egger et al., 1997) using weighted MRA models to analyze the relationship between the empirical estimates and their st. errors.

Figure 5 illustrates the funnel plots for the wage (Panel a) and the employment (Panel b) elasticities from technology, while Figure 6 presents similar evidence for two trade samples. Simple inspection reveals that in all samples there is considerable variation in the evidence base, with both positive and negative reported effects. Notably, although there is a tendency for more precise estimates to concentrate around the weighted averages, this is not always the case. As a result, we cannot offer conclusive evidence as to the presence of publication selection by simply eyeballing the scatterplots.



Figure 5. Funnel plots of technology elasticities

(a) Wage elasticities

Source: Authors' calculations

Note: Hollow circles are statistically insignificant elasticities, while full circles are statistically significant. The solid blue line indicates the weighted mean (0.001) (fixed effects estimator) and the dashed line indicates the unweighted mean (-0.211)



(b) Employment elasticities

Source: Authors' calculations

Note: Hollow circles are statistically insignificant elasticities, while full circles are statistically significant. The solid blue line indicates the weighted mean (0.287) (fixed effects estimator) and the dashed line indicates the unweighted mean (0.091)



Figure 6. Funnel plots of trade elasticities

(a) Wage elasticities

Source: Authors' calculations

Note: Hollow circles are statistically insignificant elasticities, while full circles are statistically significant. The solid blue line indicates the weighted mean (0.025) (fixed effects estimator) and the dashed line indicates the unweighted mean (0.128)



(b) Employment elasticities

Source: Authors' calculations

Note: Hollow circles are statistically insignificant elasticities, while full circles are statistically significant. The solid blue line indicates the weighted mean (0.020) (fixed effects estimator) and the dashed line indicates the unweighted mean (-0.088)

More robust evidence is obtained by performing the funnel asymmetry test (Eq. 5) which determines publication selection by a regression of the elasticities (e_i) against their corresponding st. errors (SE_{ei}) .

$$e_i = \beta_0 + \beta_1 S E_{ei} + u_i \tag{5}$$

Based on Eq. (5), we use the *funnel asymmetry* and *precision effect* tests (FAT-PET). Egger et al. (1997) argue that a conventional t-test of β_1 is the funnel asymmetry test (FAT) which ascertains the presence, the direction and the magnitude of publication selection. Publication bias is considered to be modest if $|\widehat{\beta_1}| < 1$, substantial if $1 < |\widehat{\beta_1}| < 2$ and severe if $|\widehat{\beta_1}| > 2$ (see also Doucouliagos & Stanley, 2013). Furthermore, testing for $\beta_0 = 0$ is the precision effect test (PET), which tests for a statistically significant genuine effect, after controlling for publication selection. Based on this, Doucouliagos (2011) suggests that when $\beta_0 < 0.07$, the estimated genuine effect is small.

However, Eq. (5) is heteroscedastic, since the explanatory variable is a sample estimate of the standard deviation of the dependent variable (Havranek & Irsova, 2011). Thus, we estimate Eq. (5) using weighted least squares (WLS). Stanley & Doucouliagos (2012) indicate that the inverse variance $(1/SE_{ei}^2)$ is the optimal weight for the st. errors.¹⁹ Furthermore, to account for within-study dependence, arising from the inter-dependence between the multiple estimates collected per primary study, we estimate Eq. (5) using robust st. errors, clustered by primary study.

The outcomes of simple FAT-PET-MRA models are reported in Appendix C (Table C1). This preliminary evidence indicates that publication bias is not a major concern for our analysis. Furthermore, the univariate models fail to establish significant genuine effects, once we account for publication selection. Nevertheless, Stanley (2008) stresses the various statistical problems of the FAT-PET meta-regression framework, which tends to be too conservative in estimating the genuine effect, therefore prone to committing Type I error. As a result, the test outcomes should be interpreted with some caution. Furthermore, Stanley & Doucouliagos (2012) argue that univariate FAT-PET-MRA tests should be expanded to allow for greater complexity,

¹⁹ Similarly, weighting Eq. (5) with the inverse of the st. error is equal to dividing it with SE_{ei} , thus transforming it to: $t_i = \beta_0(1/SE_{ei}) + \beta_1 + \varepsilon i$ where the dependent variable is the t-statistic of each estimated effect and $\varepsilon_i = u_i/SE_{ei}$. The transformed equation can be estimated by ordinary least squares.

reflecting the variation in the literature and large-sample misspecification biases. Therefore, in what follows, we discuss how we use the coded sources of variation to build the multivariate MRA models (Section 5.1) and investigate the presence of publication selection and genuine effects based on the outcomes (Section 5.2).

5. Multivariate meta-regression analysis

5.1 Empirical strategy

The current meta-analysis applies multivariate meta-regression models to investigate the labor market effects of technology and trade. Specifically, we add to the univariate Eq. 5 the sources of methodological and structural variation that are discussed in Section 4.1. Following Stanley & Doucouliagos (2012), we replace the genuine effect (β_0) with the term ($\beta_0 + \sum \beta_k Z_{ki}$) to model both excess heterogeneity (*Z*) and large sample or misspecification biases. Furthermore, we substitute the univariate term for publication selection ($\beta_1 SE_i$) with a multivariate one ($\beta_1 SE_i + \sum (\delta_j K_{ji})SE_i$), resulting in the following equation (Eq. 6):

$$e_i = \beta_0 + \sum \beta_k Z_{ki} + \beta_1 S E_i + \sum (\delta_j K_{ji}) S E_i + \varepsilon_i$$
(6)

The K_{ji} variables reflect the observed aspects of the quality of the estimation method that affect the propensity to report an estimate. A main advantage of Eq. (6) is that the Z- and Kvariables can be employed to control for the effects of research quality both on the magnitude of the actual empirical effect (Z) and the preference that an estimate is reported and published (K) in the first place. Based on the wide set of explanatory variables and the number of observations in the technology and trade samples, it is practical to limit the number of Kvariables (Stanley & Doucouliagos, 2012). We assume that K-variables reflect publication quality (*peer-reviewed journal, publication year and 10-year impact factor*). Furthermore, based on the discussion in Section 4.1, we split the moderator variables (Z) into two parts, reflecting the methodological (M) and the structural (S) variation. Equation (7) reports the model that is estimated.

$$e_{i} = \beta_{0} + \beta_{1}SE_{i} + \sum \beta_{k}M_{ki} + \sum \beta_{n}S_{ni} + \sum (\delta_{m}K_{mi})SE_{i} + \gamma'p_{i} + \theta'c_{j}y_{t} + \lambda'c_{j} + \mu'y_{t} + \varepsilon_{i}$$

$$(7)$$

Where the dependent variable (e_i) is the collected elasticity, SE_i is the corresponding st. error, M_{ki} is the vector of the k variables reflecting methodological heterogeneity and S_{ni} is the vector including the n aspects of structural variation. In addition, we include primary study

fixed effects (p_i) which control for study-invariant measures and country by midpoint $(c_j y_t)$ fixed effects which control for unobserved heterogeneity related to the country and the average period of analysis, while the latter are also used independently $(c_j \text{ and } y_t)$. Finally, ε_i stands for the error term.

Similarly, as in Section 4.2, the multivariate meta-regression model (Eq. 7) also suffers from heteroscedasticity. We use weighted least squares (WLS) with the inverse variance as the analytic weights to deal with this.²⁰ This strategy reduces the effects of (implausibly) large elasticities, which typically have large st. errors. To account for within study dependence, we estimate Eq. (8) using robust st. errors, clustered by primary study.

The most meaningful way to report meta-analytic results (Stanley & Doucouliagos, 2012) is to start from an all-inclusive WLS model, that is, a general-to-specific approach (backwards selection) and report results of more parsimonious models, where all variables are significant at least to the 10% level. The merit of backwards selection is that "... *model construction proceeds from a general model to a more structured, in an ordered and statistically valid fashion, and in this way avoids the worst of data mining*" (Charemza & Deadman, 1997, p.78). In our case, this is particularly appropriate given the multitude of the moderator variables, which explodes the number of possible MRA models well above the number of observations. The resulting Type I error implies that some research dimensions in the all-inclusive models might be statistically significant even if the results are only based on random noise (Sala-i-Martin, 1997). Furthermore, even when there are sufficient degrees of freedom, such all-inclusive MRA models often suffer from multicollinearity and low statistical power, which undermine their predictive power.

A common methodological debate in meta-analysis refers to the treatment of outliers, which might bias the meta-regression outcomes. Despite the extreme values, outliers reflect important information; therefore, we report the meta-regression outcomes for the entire samples in the main results. To verify that outliers do not drive the reported outcomes, Appendix D1 reports the MRA results excluding the outliers. First, we filter out extreme values by trimming the first

²⁰ Alternatively, the inverse st. error is also suggested as an analytic weight, which would transform Eq. (7) into: $t_i = \beta_1 + \beta_0 \left(\frac{1}{SE_i}\right) + \sum \beta_k M_{ki} \left(\frac{1}{SE_i}\right) + \sum \beta_n Sni(\frac{1}{(SE_i)}) + \sum \delta_m K_{mi} + \gamma' p_i + \theta' c_j y_t + \lambda' c_j + \mu' y_t + v_i$ However, simulations indicate that the variance performs better as the analytic weight in WLS-MRA models (Stanley & Doucouliagos, 2012).

and the last 5 percentiles from the distributions of the collected elasticities and secondly, we winsorize the first (last) 20 percentiles of each distribution with the value of the 20th (80th) percentile. The main difference between the two methods is that the latter retains the same number of estimates, while the former eliminates 46 technology and 55 trade elasticities.

As a sensitivity analysis, we compare the abovementioned preferred approach with random (REE) and fixed effects (FEE) meta regression models (reported in Appendix E). Note, however, that Moreno et al. (2009) and Stanley et al. (2010) illustrate that both REE and FEE models are biased in the presence of publication selection. Furthermore, our dataset lacks a clear panel structure, further justifying our choice for the cluster-robust WLS-MRA models as our preferred analysis.

Equation (7) provides a flexible framework to account for the vast heterogeneity in the collected samples and potential publication selection. Nevertheless, the relationship between the standard error and the propensity to report an estimate might not be linear. Instead, simulations have indicated the appropriateness of a squared term (SE_i^2) as a corrected estimate in a multivariate MRA. The resulting *precision effect test corrected for st. error* (PEESE) is obtained from substituting the SE_i with its squared term, as in Eq. (8), which will be estimated using the inverse variance as the analytic weight:

$$e_{i} = \beta_{0} + \beta_{1}SE_{i}^{2} + \sum \beta_{k}M_{ki} + \sum \beta_{n}S_{ni} + \sum (\delta_{m}K_{mi})SE_{i}^{2} + \gamma'p_{i} + \theta'c_{j}y_{t} + \lambda'c_{j} + \mu'y_{t} + \varepsilon_{i}$$

$$(8)$$

In what follows, we report the empirical results from estimating Eq. (7) and (8) for the labor market elasticities from technology and trade.

5.2 Results

The current section reports the outcomes from our empirical analysis distinguishing between the wage and employment elasticities from technology and trade. For each of the four subsamples (wage or employment elasticities from technology and trade), we report the outcomes from the WLS-MRA models (Eq. 7), first with the entire set of moderator variables; then the WLS-MRA general-to-specific model, which is our preferred estimation and finally we estimate the PEESE model (Eq. 8). To illustrate the growing importance of trade with China, we complement the trade analysis with WLS-MRA models restricting trade only with China. As outlined above, the discussion is predominantly based on the WLS general-to-specific models, which typically replicate the outcomes of the all-inclusive ones.

Wage elasticities from technology

Table 4 reports the estimation results for the technology elasticities. Regarding the wage elasticities (Columns 1-3), the model with all the moderator variables (Column 1) and the more parsimonious general-to-specific model (Column 2) provide similar outcomes, verifying the robustness of our analysis to alternative model specifications. In more detail, Column 2 indicates the significance of 10 Z- and 2 K-variables as well as the coefficient of the $SE(\beta_1)$ and the intercept (β_0).

As explained above we have to deal with possible publication biases. The average multivariate publication bias is calculated from the average estimated value of: $\beta_1 SE + \sum (\gamma_i K_i)SE$, as opposed to $\beta_1 SE$ for the univariate MRA. Using an F-test, we verify the joint significance of the 2 K-variables (*publication year* and *10-year impact factor* each one interacted with the st. error) and the coefficient of the st. error (F_{3,20} = 24.03, p < 0.00001).

Plugging in the estimated beta coefficients and the average value of the st. error (0.234), we conclude that the average publication selection in the wage elasticities from automation equals -0.214. This contrasts the evidence from the univariate MRA (Appendix C) which indicated the absence of publication selection in the sample of wage elasticities from technology (assuming significance of the regression coefficient, calculating $\beta_1 SE$ would result on an average publication bias of 0.013), which highlights the importance of the multivariate MRA. Subtracting the calculated publication bias from the average wage elasticity from technology (-0.211 - Table 1), yields an overall effect equal to zero (0.003); thus, providing evidence that the negative average wage elasticity is due to the impact from publication bias. Once we properly account for it, the average wage elasticity becomes marginally positive.

Along similar lines, the genuine effect is a multivariate combination of the 10 significant Z-variables (*manufacturing, services, technology and trade, time series, levels, endogeneity, 10-year impact factor, trade with developing countries, after 2001* and *data span*). An F-test ($F_{6,21}$ = 64088.92, p < 0.00001) verifies the presence of genuine systematic patterns in the reported wage elasticities from automation. Assuming that all the Z-dummy variables are equal

to $zero^{21}$ and substituting the continuous ones with their sample means, we estimate a genuine wage effect equal to 0.440, which is more robust evidence that the average wage elasticity is positive and large (Doucouliagos, 2011), once we control for publication bias.

The PET is often biased downward when there is a genuine effect (Stanley, 2008). As a result, we focus on the PEESE test (Column 3) for a less biased, corrected estimate of the actual effect. Column 3 reports almost identical results with Column 2 (only 2 Z- and 1 K-variables differ ether in sign or significance). In particular, 10 Z-variables are significant determinants of the genuine effect ($F_{6,20} = 2900.02$, p < 0.00001). Following the same concept as before (setting all Z-dummy variables are equal to zero and replacing the continuous ones with their sample means), in line with our expectations we conclude on a larger (0.738) genuine effect. Reassuringly, -once again- the multivariate nature of our analysis converts the negative unweighted average wage elasticity into positive.

The estimated genuine effect from the PET allows for the calculation of 'best practice' estimates. Given their controversial nature (Doucouliagos & Stanley, 2009), identifying the 'best practice' estimates requires some professional judgement. First, we filter out publication selection by setting SE = 0. Then, we can safely argue that robust econometric techniques are based on time series analysis, after 2001 with first differenced dependent variables, which account for unobserved heterogeneity. Furthermore, the estimated model should include both measures of technology and trade, while the estimation technique should address the endogenous nature of technology. Translating the above choices into appropriate values for the respective significant Z-dummy variables and substituting the two continuous ones (10-year impact factor and dataspan) with their sample means, yields an overall wage elasticity of -0.104. The negative outcome indicates that our subjective judgement on what constitutes 'best practice' transforms the positive genuine effect into negative. This is by no means a limitation of the analysis. Instead, it clearly indicates the strong impact from the increased variation in the literature (reflecting for instance the period of the analysis, or the numerous methodological choices regarding the estimated model). Furthermore, the best practice estimate is sensitive to the choice of the reference category in the applied dummy structure. As such, we place more emphasis on the genuine effect, as verified by both the multivariate PET and the PEESE tests.

²¹ This implies that the estimated elasticity is at the macroeconomic level, before 2001. Also, the estimated econometric model does not include a trade measure, it is based on cross-sectional analysis with a first-differenced dependent variable, while the estimation method is not controlling for endogeneity.

	Wage elasticities			Employment elasticities			
Variable	WLS	WLS	PEESE	WLS	WLS	PEESE	
		general to			general to		
		specific			specific		
	(1)	(2)	(3)	(4)	(5)	(6)	
FAT – PET – PEESE tests	1.015	0.551.000		0 ==1		0.001.4	
Intercept (β_0) (PET) or (γ_0) (PEESE)	-1.815	0.551**	0.678**	0.751	0.706*	0.231*	
	[3.622]	[0.161]	[0.188]	[0.523]	[0.358]	[0.131]	
St. error (β_1) or st. error ² (γ_1) (FAT)	-1.807	-1.474**	2.018	-1.540	-1.296	0.780	
	[1.640]	[0.567]	[1.848]	[1.645]	[0.951]	[0.625]	
Type of innovation (ref. process innovation	n)			0.00111	0.00411	0.00111	
Product innovation	0.005	omitted	omitted	0.001**	0.001**	0.001**	
	[0.003]	onnica	ommea	[0.0003]	[0.0005]	[0.0003]	
Data aggregation (ref. worker-level)							
Industry-level	0.200**	omitted	omitted	0.100	0.167**	0.190**	
	[0.086]	ommed	ommed	[0.105]	[0.078]	[0.063]	
Firm-level	0.925	omitted	omitted	0.982^{***}	1.117***	1.128***	
	[2.126]	omned	omnea	[0.137]	[0.066]	[0.061]	
Sectoral Ievel (ref. economy-wide)							
Manufacturing	-0.277**	-0.225*	-0.790***	0.227	0.221	0.200	
Wandracturing	[0.086]	[0.111]	[0.108]	[0.151]	[0.139]	[0.128]	
Services	-0.270***	-0.218*	-0.783***	0.076**	0.068**	0.038***	
Services	[0.086]	[0.112]	[0.108]	[0.017]	[0.013]	[0.005]	
Skill-level (ref. varied skill)							
Low-skill	-0.003	omitted	omittad	0.061*	0.050**	0.012	
Low-skiii	[0.004]	omitieu	omitieu	[0.033]	[0.023]	[0.023]	
Madium skill	-0.002	omittad	omitted	0.069**	0.060**	0.029	
wiedluiii-Skiii	[0.004]	omilied	omitied	[0.028]	[0.018]	[0.017]	
High shill	-0.001	omittad	omitted	0.195***	0.187***	0.160***	
High-skill	[0.002]	omilied	omilied	[0.021]	[0.014]	[0.005]	
I							

Table 4. Labor market effects from technology (Dependent variable: e_i)

Other methodological aspects						
Technology and trade	-0.008**	-0.008**	-0.017**	0.009	omitted	omitted
(ref. only technology)	[0.003]	[0.002]	[0.002]	[0.026]	omittea	omittea
Time series	-0.010	-0.010*	-0.008***	-0.142***	-0.146***	-0.160***
(ref. cross-sectional)	[0.007]	[0.005]	[0.0004]	[0.016]	[0.008]	[0.00007]
Lagged independent variable	0.001	omitted	omitted	-0.007	omitted	omitted
(ref. contemporaneous ind. var.)	[0.001]	omitieu	omilieu	[0.015]	omittea	omitteu
Dependent variable in levels	-0.445**	-0.391***	-0.872***	0.148***	0.152***	0.161***
(ref. first-differenced dep. var)	[0.122]	[0.086]	[0.108]	[0.016]	[0.006]	[0.00006]
Controlling for endogeneity	-0.015**	-0.019**	-0.040***	0.029	omitted	omitted
(ref. no IV approach)	[0.005]	[0.005]	[0.105]	[0.019]	omnieu	omitica
Wages in shares	2.864	omitted	omitted			
(ref. wages not in shares)	[4.528]	omnica	omnica			
Hours worked				-0.0001***	0.0001***	0.0001***
(ref. number of workers)				$[8.55*10^{-8}]$	[6.80*10-11]	[6.43*10 ⁻¹¹]
Publication characteristics (K-)						
Peer reviewed paper	1.546	omitted	omitted	-0.782**	-0.848**	-0.307*
(ref. working paper)	[2.130]	omnieu	ommed	[0.250]	[0.406]	[0.176]
Peer reviewed paper * st. error	0.155	0.155 omitted	omitted	0.534	omitted	omitted
	[1.577]	ommed	ommed	[1.639]	ommou	ommed
Publication year	-0.781	-0.127	-0.934**	0.602***	0.676***	0.650***
i donioution your	[0.766]	[0.389]	[0.349]	[0.133]	[0.021]	[0.022]
Publication year * st. error	-1.896***	-2.045***	-1.070*	-1.355**	-1.282**	-2.682
	[0.375]	[0.423]	[0.597]	[0.580]	[0.565]	[2.712]
10-Year impact factor	-2.031	1.130**	0.713	0.244	omitted	omitted
	[3.048]	[0.478]	[0.424]	[0.829]	ommou	ommed
10-Year impact factor * st_error	-4.030**	-3.764**	-2.793	0.881	omitted	omitted
To Tour impact fuctor st. error	[1.765]	[1.407]	[3.919]	[1.169]	ommed	omnea
Geography (ref. US-UK-Canada-Japan)						
Europe	-0.375	omitted	omitted	-0.0008	omitted	omitted
Zuropo	[0.456]			[0.005]		
Spatial unit (ref. national-level analysis)						

Regional – level analysis	0.114 [0.113]	omitted	omitted	0.492** [0.217]	0.474** [0.173]	0.769*** [0.155]
Development status (ref. not-specified)						
Developed	omitted	omitted	omitted	0.580 [0.358]	0.888*** [0.202]	0.547*** [0.050]
Developing	0.557*** [0.071]	0.575*** [0.063]	1.014*** [0.050]	0.198** [0.079]	0.211** [0.072]	0.325*** [0.044]
Time (ref. midpoint before 2001)						
After 2001	-0.307** [0.080]	-0.279*** [0.047]	-0.330*** [0.077]	0.025 [0.075]	omitted	omitted
Other controls						
Dataspan	-0.349*** [0.011]	-0.349*** [0.008]	-0.385*** [0.010]	0.241*** [0.005]	0.240*** [0.006]	0.252*** [0.0006]
Number of RHS variables	1.1*10 ⁻⁶ [0.00006]	omitted	omitted	0.0008 [0.005]	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important controls or</i> <i>fixed effects</i>)	-0.049 [0.059]	omitted	omitted	-0.095** [0.045]	-0.098** [0.038]	-0.118 [0.092]
Observations	231	231	231	392	392	392
Adjusted R ²	0.626	0.622	0.480	0.985	0.985	0.985
Primary study fixed effects	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes
Country by Midpoint fixed effects	yes	yes	yes	yes	yes	yes

Note: Robust st. errors, clustered by primary study are reported in brackets. All models are weighted by the inverse of the variance of the collected elasticities. Columns 2 and 5 are based on the respective WLS-MRA models (Columns 1 and 4), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories is significant at the 10% level. * for p<0.1, ** for p<0.05 and *** for p<0.01.

Besides the central findings discussed so far, our multivariate regression analysis reveals how the complex structure in economics research potentially shapes the reported wage elasticities from technology. First, we uncover that the average positive wage elasticity, calculated for economy-wide studies, decreases by a magnitude of 0.225 for the manufacturing and 0.218 for the service sector. This conclusion supports the relevant arguments that possible negative effects from automation at the sectoral level are often compensated by positive effects on other sectors (inter-sectoral linkages – Blechinger et al., 1998) which tend to increase the economy-wide wage elasticities from automation.

Notably, our regression analysis highlights the evolution patterns of the wage effects from technology. In particular, more recent wage elasticities from technology are typically lower by an order of 0.279. Furthermore, the inclusion of trade measures in the estimated model tends to decrease the wage elasticity from technology by an order of 0.008. Nevertheless, when the trade measure refers to trade with developing countries, the wage elasticity from automation tends to be larger by an order of 0.575. Assuming that more recent estimations include measures of both technology and trade, the above effects uncover the significant overlap between the effects from technology and trade, often discussed in the theory (Feenstra, 2016; Hummels et al. 2018).

Along similar lines, our WLS-MRA illustrates the importance of methodological choices; in that, the positive genuine effect decreases by 0.010 for time series analyses and by 0.391 in models where the dependent variable is in levels. In addition, less positive effects are reported in empirical models which address the endogenous nature of technology (-0.019) and estimations based on a larger number of years (-0.349), which reflect better data accessibility. The above characteristics are aspects of more robust and sophisticated modern econometric techniques. Similarly, elasticities published in high-quality journals (which are typically based on robust methodologies) tend to be larger (1.130 per 1 s.d. of the standardized 10-year impact factor included in the analysis). Taken together, the more detailed results provide new insights regarding the diverse impact from technical research choices on the reported elasticities.

Employment elasticities from technology

Table 4 (Columns 4-6) report the outcomes of our analysis on the employment elasticities from technology. Starting from the WLS-MRA model with all the coded sources of variation (Column 4), we estimate the general-to-specific model (Column 5), where 17 Z-variables are significant at least for $\alpha = 0.1$. First, we calculate publication selection based on the formula:
$\beta_1 SE + \sum (\gamma_i K_i) SE$. Given the joint significance of the relevant terms (F_{2,36} = 2.68, p < 0.10), the multivariate publication selection is equal to -0.193. Negative publication bias was the outcome of the univariate funnel asymmetry test ($\beta_1 SE$); albeit much larger in magnitude (-4.830) (Appendix C). Once we subtract the calculated publication bias from the unweighted mean (Table 1), we conclude on a positive average effect from technology on employment (0.284).

Once we account for publication selection, the genuine effect is a multivariate function of the intercept and all the significant Z-variables (*product innovation, industry- / firm-level data, services, low-/ medium- / high-skilled, time-series, levels, hours, peer-reviewed, publication year, regional, developed, developing, dataspan, including controls*). Given their joint significance ($F_{9,36} = 3.4 * 10^5$, p < 0.00001) and filtering out publication selection ($SE_i = 0$), the genuine effect is calculated by setting all the relevant dummy indicators equal to zero and using the mean sample values of the continuous ones (*dataspan, publication year*). As a result, the genuine effect is 0.496. Based on the PEESE test (Column 6) and after verifying the joint significance of all the Z-variables ($F_{9,36} = 4.1 * 10^5$, p < 0.00001), the same reasoning yields an average effect of: 0.306. Taken together, we provide robust evidence that the genuine effect of technology on employment is positive and often exceeds the limit for a strong association (Doucouliagos, 2011), even after accounting for publication selection and excess heterogeneity.

Based on Column 5, we estimate best practice estimates concerning the employment elasticities from technology. The calculations are subjective though and based on the significant covariates and our choices about what constitutes '*best practice*'. Here, we opt for the more direct firm-level effects, published in peer-reviewed journals. Also, the time-series analysis controls for the endogenous nature of technology, while the dependent variable is first-differenced and the estimated model includes important control variables. Translating those choices into the appropriate values for the respective binary indicators and substituting the continuous ones (*dataspan, publication year*) with their mean values, we calculate a best practice estimate equal to 0.796. As a result, our subjective argumentation on what constitutes '*best practice*', results in a larger positive impact from technology on employment compared to the calculated genuine effect (0.496).

Column 5 also indicates several noteworthy patterns regarding the effects from the various sources of heterogeneity. A key outcome highlights the skill-biased nature of technology on employment (Autor et al., 2003). In particular, the average positive effect from

technology, which is calculated for workers of various skill levels, is further increased for low-, medium- and high-skilled workers. Nevertheless, the estimated term for high-skilled workers (0.187) is much larger compared to the ones for low- (0.050) and medium-skilled workers $(0.060)^{22}$. As a result, our analysis confirms that the beneficial employment effects from technology are mainly accrued to high-skilled labor.

Table 4 also indicates that methodological choices may also shape the overall employment elasticities from technology. First, industry- and firm-level estimates are typically larger from worker-level ones, by an order of 0.167 and 1.117 respectively. Similarly, elasticities for the service sector are larger from economy-wide ones by a magnitude of 0.068. Furthermore, time-series analyses tend to report lower employment elasticities from technology by an order of 0.146, while not first-differencing the estimated model increases the reported elasticities by a magnitude of 0.152.

Shifting to the various aspects of structural variation, the WLS-MRA illustrates that employment elasticities from technology published in peer-reviewed journals are lower by an order of 0.848 compared to those on working papers; thus adding to the discussion of publication selection. As to the evolution of the reported effects, the *publication year* variable points that the employment elasticities are more positive by 0.676 for every s.d. of the standardized variable. Regarding geography, the coefficient of our regional variable indicates that analyses at the regional level tend to report larger positive elasticities by a factor of 0.474. Similarly, studies comprising larger time periods tend to report larger positive outcomes by an order of 0.240 for every additional year in the dataset.

Taken together, the analysis so far points towards a number of noteworthy outcomes regarding the labor market effects from technology. First, we document that negative publication selection potentially contaminates both the wage and the employment elasticities literature. Once we account for publication selection, we conclude that –on average- automation benefits both wages and employment. Notably, we indicate that the skill-biased nature of technology is primarily displayed in employment, where high-skilled worker benefit disproportionally more, compared to low-skilled ones; the analysis did not trace any similar effect on wages. Furthermore, we revealed that the wage effects from technology are less positive in more recent

²² Based on the estimated terms from Column 5, comparing the effects from technology on low- and high-skilled workers (0.546 and 0.683 respectively), we conclude that the differential impact from technology on skilled workers is almost 25% higher than the impact on unskilled ones.

studies; the opposite happens to the employment effects, which tend to increase over time. Finally, the analysis uncovered how several methodological choices (adopting an IV approach, performing a time-series analysis, using industry-level data etc.) shape the labor market effects from automation.

Wage elasticities from trade

Table 5 (Columns 1-4) reports the meta-regression results for the wage elasticities from trade. The discussion is based on the general-to-specific model (Column 2), which is very similar to the model that controls for all sources of variation (Column 1).

The general-to-specific WLS model indicates that all the K-variables are statistically insignificant; this is also the case for the β -coefficient of the st. error. As a result, we conclude on the absence of publication selection in our sample of the wage elasticities from trade, which challenges the negative and significant bias (-2.285) estimated from the univariate FAT (Appendix C). Furthermore, Column 2 points to the significance of 13 Z-variables, which are also jointly significant ($F_{11,33} = 7.7 \times 10^6$, p < 0.00001). To calculate a plausible value for the genuine effect, we set all the dummy Z-variables equal to zero and substitute the continuous ones (publication year, 10-year impact factor, dataspan) with their average values. Utilizing also the intercept and the corresponding beta coefficients, we point towards a large positive genuine effect of 1.685, which is in line with the positive effect indicated from the univariate PET test (0.026). To verify the robustness of the multivariate analysis, we rely on the multivariate PEESE test (Column 3). Re-iterating the above calculations yields a similar genuine effect (0.916), which indicates strong association and is significant in all the conventional levels. In sum, the analysis so far found no evidence of publication selection in the wage elasticities from trade literature coupled with robust evidence of a positive genuine effect of trade on wages.

The multivariate analysis is particularly useful for calculating 'best practice' estimates. Like the technology samples, the above estimates are based on the significant variables in our preferred specification (Column 2) and some reasonable assessment. We assume that trade refers to both intermediate and final products, estimated for a firm in Europe, operating in the manufacturing sector. Moreover, we hypothesize that the estimate is obtained from a more recent study (after 2001), published on a peer- reviewed journal. To obtain a 'best practice' estimate, we convert the abovementioned choices into appropriate values in the dummy structure of our control variables. Then, we filter out publication selection by setting the st. error equal to zero ($SE_i = 0$) and we substitute the significant continuous Z-variables (*publication year*, 10-year impact factor, dataspan) with their mean values. The above calculations provide a 'best practice' estimate equal to 0.775.

Interestingly, the multivariate MRA indicates how the multiple sources of variation shape the estimated genuine effect (1.685). In particular, we show that the positive average wage effects from trade are larger for high-skilled workers by a magnitude of 0.013, which indicates that workers above a skill threshold experience more beneficial effects on their wages because of trade. In addition, based on the negative estimated term for trade in intermediates (-0.002), our analysis points to smaller positive wage effects from vertical specialization (Hijzen, 2005). This outcome is complemented by the estimated term for the *developed* Z-variable (0.008), which shows that trade with developed countries (typically of the intra-industry type (Debaere et al., 2006)) results in larger positive average effects.

Furthermore, the reported elasticities are conditional on various methodological choices. First, in line with the literature indicating that firms engaging in international trade will likely benefit, we show that the wage effects at the firm level are larger compared to worker-level ones by a magnitude of 0.016. Regarding the sectoral level of analysis, the WLS-MRA (Column 2) challenges the popular opinion that the negative effects from trade are predominantly manifested in the manufacturing sector (Blechinger et al., 1998) Specifically, we show that wage elasticities in the manufacturing are larger by an order of 0.014 compared to economy-wide ones. Nevertheless, this outcome possibly reflects productivity effects and does not include the negative employment impact, which we ascertain in the analysis of the employment elasticities from trade.

The significant *dataspan* variable reveals that the wage elasticities based on large periods of analysis are lower on average by order of 0.408. Furthermore, the negative term for the *wage in shares* variable (-2.613) indicates that the wage elasticities are smaller when the authors investigate relative labor demand, thereby expressing the dependent variable as a share of the wages against the total wage bill.

Regarding the importance of publication quality, our analysis provides mixed evidence. First, we show that the wage elasticities extracted from studies published in peer-reviewed journals are typically less positive by order of 0.948. In contrast, the positive estimated term for the *10*-

year impact factor Z-variable (1.516), indicates that high-quality studies tend to report larger positive wage elasticities from trade. As to their evolution over time, our analysis indicates that the wage elasticities published after 2001 are typically less positive by an order of 0.716. This effect is partially mediated by the impact from the *publication year* variable, which shows that more recent elasticities are typically larger by order of 0.130. Finally, our analysis highlights the important role of geography. In particular, the positive effect of trade on wages is a bit smaller in the more rigid European labor markets by an order of 0.008, compared to the more flexible labor markets (like the US, the UK, Canada or Japan) where wages are expected to be the main adjustment mechanism.

To address the central role of China in global trade since the early 2000s, Column 4 estimates the general-to-specific model (Column 2) when China is the only trade partner. The limited number of observations (64) implies that the discussed findings should be taken with a grain of salt. Nevertheless, the insignificance of the β -coefficients for the st. error and all the Kvariables indicates the absence of significant publication selection. Furthermore, utilizing the intercept and after verifying the joint significance of all the significant Z-variables (F_{3,7} = $7.8*10^6$, p < 0.00001), we calculate a genuine effect of -0.031. Interestingly, we conclude that although trade benefits wages in the industrialized countries, trade with China results in negative effect, which is marginally below the limit for large effects (Doucouliagos, 2011).

The above caveat notwithstanding, it is worth noting that the analysis of the wage effects from trade with China largely verifies many of the outcomes discussed so far. In particular, trade with China corroborates the already established negative effects from trade in intermediates (-0.003) as well as the positive ones for the wage elasticities in the manufacturing sector (0.019) and the wages of high-skilled workers (0.003). Similarly, industry-level studies report larger elasticities by 1.917, which can reverse the negative genuine effect into positive. Regarding publication quality, Column 4 illustrates that the elasticities published in peer-reviewed journals are further decreased by 1.955, an effect partially mitigated by the positive impact for the outcomes reported in high-quality studies (0.466), as indicated by the *10-year impact factor* variable. As to the impact from time, our analysis shows that evidence after 2001 is typically lower (-0.088); however, this is contrasted by the positive term for the *publication year* variable (0.457), which points to larger wage elasticities in more recently published studies.

Employment elasticities from trade

Table 5 (Columns 5-8) reports the outcomes from our multivariate MRA regarding the employment elasticities from trade. The discussion is based on the general-to-specific model (Column 6), which almost replicates the sign and significance patterns of the comprehensive model (Column 5), indicating the significance of 10 Z- and 1 K-variable. The average publication bias is based on the joint significance of the coefficient for the st. error and the significant K-variable ($F_{2,40} = 6.2$, p < 0.05) and is equal to -0.049, significant at the 5% level. This contrasts the insignificant bias estimated in the univariate analysis (Appendix C). Subtracting the calculated bias from the average employment elasticity from trade (-0.088) results in a less negative average elasticity from trade on employment (-0.039).

To estimate a reasonable value for the genuine effect, we first verify the joint significance of the intercept and the 10 significant Z-variables ($F_{8,48} = 1.4*10^8$, p < 0.00001). Then, we substitute all the dummy Z-variables with zero and the two continuous ones (*publication year* and *dataspan*) with their mean values. Based on the above, the calculated effect is equal to 3.187. As such, the multivariate analysis points to a large positive genuine effect beyond publication selection, which contrasts the insignificant one from the univariate analysis (Appendix C). Given the downward bias of the PET test, we also estimate the genuine effect based on the PEESE results (Column 7). After verifying the joint significance of the relevant variables ($F_{8,40} = 2.3*10^8$, p < 0.00001), similar calculations yield a more conservative genuine effect: 1.193. Overall, our multivariate approach provides some evidence of negative publication selection, coupled with more robust evidence of a positive and strong (Doucouliagos, 2011) genuine effect once publication selection is accounted for.

Like before, we estimate best practice estimates concerning the employment elasticities from trade. Our calculations are based on the general-to-specific WLS model and refer to more recent (after 2001) estimates for the manufacturing sector, reported in peer-reviewed journals. Plugging the respective values into the corresponding dummy variables and substituting the continuous one (*publication year* and *dataspan*) with their mean values, we calculate a best practice estimate equal to 2.167; further verifying the positive employment effects from trade.

	Wage elasticities				Employment elasticities			
Variable	WLS	WLS general to specific	PEESE	WLS general to specific - China	WLS	WLS general to specific	PEESE	WLS general to specific - China
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
FAT – PET – PEESE tests								
Intercept (β_0) (PET) or (γ_0) (PEESE)	2.064* [1.077]	1.701** [0.665]	0.899*** [0.118]	0.009 [0.243]	-13.987*** [1.337]	3.561** [1.558]	1.339 [2.101]	-2.312 [6.614]
St. error (β_1) (FAT) or (γ_1)	0.176 [0.318]	0.375 [0.310]	0.0006 [0.002]	-0.752 [0.982]	0.963 [0.750]	-0.058 [0.314]	-0.059 [0.211]	-5.068 [3.248]
Type of trade (ref. trade in final goods)								
Trade in intermediates	-0.002** [0.0008]	-0.002*** [0.0006]	-0.003*** [0.0006]	-0.003** [0.0009]	-0.040 [0.087]	omitted	omitted	omitted
Data aggregation (ref. worker-level)								
Industry-level	-0.004 [0.004]	-0.009 [0.007]	-0.008 [0.006]	1.917* [0.967]	10.793*** [0.885]	omitted	omitted	omitted
Firm-level	0.014* [0.007]	0.016** [0.006]	0.016*** [0.006]	omitted	11.139*** [0.904]	omitted	omitted	omitted
Sectoral Ievel (ref. economy-wide)								
Manufacturing	0.014** [0.003] -0.002	0.014*** [0.001] -0.003	0.015*** [0.00004] -0.0005	0.019** [0.004] -0.037	-0.149*** [0.033] -0.136**	-0.121*** [0.020] -0.099	-0.088*** [0.014] -0.060**	1.379* [0.612] 1.616**
	[0.010]	[0.009]	[0.009]	[0.049]	[0.036]	[0.019]	[0.018]	[0.633]
Skill-level (ref. varied skill)	0.004	0.001	0.007	0.0004	0.000	0.000	0.004	0.007
Low-skill	0.006 [0.004]	0.004 [0.001]	0.005 [0.004]	-0.0001 [0.003]	-0.003 [0.002]	-0.003* [0.001]	-0.001 [0.001]	-0.085 [0.048]
Medium-skill	0.007* [0.004]	0.005 [0.003]	0.006* [0.002]	omitted	-0.040** [0.019]	-0.036** [0.016]	-0.028* [0.0016]	omitted
High-skill	0.015**	0.013**	0.013***	0.003**	-0.0006	0.002**	0.001*	0.175

Table 5. Labor market effects from trade (Dependent variable: e_i)

	[0.005]	[0.004]	[0.004]	[0.001]	[0.0007]	[0.0006]	[0.0007]	[0.100]
Other methodological aspects								
Technology and trade (ref. only trade)	0.0001 [0.001]	omitted	omitted	omitted	-0.001 [0.002]	omitted	omitted	omitted
Time series (<i>ref. cross-sectional</i>)	0.007 [0.008]	omitted	omitted	omitted	-0.322** [0.096]	omitted	omitted	omitted
Lagged independent variable (<i>ref. contemporaneous ind. var.</i>)	0.0009 [0.001]	omitted	omitted	omitted	-0.0005 [0.0006]	omitted	omitted	omitted
Dependent variable in levels (ref. first-differenced dep. var)	0.018 [0.048]	omitted	omitted	omitted	0.004 [0.003]	omitted	omitted	omitted
Controlling for endogeneity (ref. no IV approach)	0.007 [0.005]	omitted	omitted	omitted	0.007 [0.006]	omitted	omitted	omitted
Wages in shares (ref. wages not in shares)	-2.987** [1.269]	-2.613** [0.867]	-1.564*** [0.128]	omitted				
Hours worked <i>(ref. number of workers)</i>					0.272*** [0.043]	0.275*** [0.041]	0.199*** [0.037]	0.147 [0.159]
Publication characteristics (K-)								
Peer reviewed paper	-1.146**	-0.948**	-0.558***	-1.955*	-0.180***	-0.970**	-0.411***	1.386
(ref. working paper)	[0.530]	[0.325]	[0.056]	[0.988]	[0.042]	[0.337]	[0.500]	[1.901]
Peer reviewed paper * st. error	0.275 [0.506]	omitted	omitted	omitted	-1.556* [0.970]	omitted	omitted	omitted
Publication year	0.114**	0.130***	0.124***	0.457**	0.181***	5.048**	1.950	-1.293
r ubileation year	[0.039]	[0.005]	[0.004]	[0.220]	[0.038]	[2.214]	[2.985]	[7.736]
Publication year * st. error	-0.011 [0.572]	omitted	omitted	omitted	1.195** [0.433]	1.148** [0.326]	omitted	5.603 [4.746]
10-Year impact factor	1.851* [0.991]	1.516** [0.598]	0.791*** [0.106]	0.466*** [0.032]	-0.567 [0.114]	omitted	omitted	omitted
10-Year impact factor * st. error	0.243 [0.433]	omitted	omitted	omitted	-0.769 [0.719]	omitted	omitted	omitted
Geography (ref. US-UK-Canada-Japan	2)							
Europe	-0.008** [0.003]	-0.008*** [0.002]	-0.011*** [0.0009]	-0.041 [0.084]	3.334*** [0.351]	omitted	omitted	omitted

Spatial unit (ref. national-level analysis	s)							
Regional – level analysis	0.018 [0.071]	omitted	omitted	omitted	2.829*** [0.264]	3.090*** [0.197]	2.780*** [0.021]	1.472* [0.709]
Development status (ref. not-specified)								
Developed	0.004 [0.003]	0.008** [0.004]	0.009* [0.005]	omitted	-0.022* [0.012]	omitted	omitted	omitted
Developing	-0.025 [0.018]	-0.023 [0.018]	-0.022 [0.018]	omitted	-0.026 [0.018]	omitted	omitted	omitted
Time (ref. midpoint before 2001)								
After 2001	-0.915 [0.576]	-0.716** [0.328]	-0.315*** [0.062]	-0.088*** [0.0005]	-0.109*** [0.020]	-0.128*** [0.016]	-0.098*** [0.0009]	0.262** [0.070]
Other controls								
Dataspan	-0.407*** [0.037]	-0.408*** [0.019]	-0.386*** [0.002]	-0.345** [0.059]	0.052*** [0.009]	0.058*** [0.008]	0.042*** [0.0008]	0.042 [0.028]
Number of RHS variables	0.0004 [0.0003]	omitted	omitted	omitted	-0.0004 [0.001]	omitted	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important</i> controls or fixed effects)	0.015*** [0.003]	0.018*** [0.002]	0.017** [0.0002]	0.015** [0.002]	0.008 [0.035]	omitted	omitted	omitted
Observations	459	459	459	64	635	635	635	60
Adjusted R ²	0.904	0.906	0.905	0.621	0.992	0.992	0.991	0.767
Primary study fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Country x Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes

Note: Robust st. errors, clustered by primary study are reported in brackets. All models are weighted by the inverse of the variance of the collected elasticities. Columns 2 and 6 are based on the respective weighted-least-squares models (Columns 1 and 5), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories are significant at the 10% level. * for p<0.05 and *** for p<0.01.

Regarding more nuanced effects from the coded sources of variation, the current analysis advances the discussion regarding the skill-biased nature of globalization of production. A key outcome highlights that the positive genuine effect is decreased for low- (-0.003) and medium-skilled workers (-0.035), while it is marginally larger for high-skilled ones (0.002). Despite the small differences, the estimated terms add to the results from the analysis of the wage elasticities from trade and imply a more beneficial effect from trade on high-skilled workers.

Notably, the WLS-MRA illustrates how numerous aspects of the study design shape the average effect. First, we indicate that elasticities for the manufacturing sector are typically lower by 0.121. In addition, employment elasticities collected from studies published in peer-reviewed journals are less positive, as indicated by the estimated term (-0.970). Considering the methodological choices, Column 6 indicates that when employment is approximated by the hours worked as opposed to the number of workers, the average elasticity is larger by order of 0.275. Regarding the impact from geography, the significant *regional* coefficient, illustrates that regional level analyses typically report larger employment effects (3.090).

The analysis highlights the importance of the time dimension in two ways. Similar to the wage elasticities from trade, the employment ones referring to the period after 2001 are smaller (-0.128). Nevertheless, this is contrasted by the positive estimated term for publication year (5.048) which reveals that more recent studies tend to report larger employment elasticities. Finally, our analysis shows that elasticities based on longer time periods (*dataspan*) tend to be more positive by order of 0.058.

Similar to the wage sample, we address the elevated role of China in world trade by reestimating the general-to-specific model (Column 6) when China is the only trade partner (Column 8). Note, however, the low number of observations (60). This limitation notwithstanding, the analysis shows the absence of publication selection. At the same time, we verify the joint significance of the relevant Z-variables ($F_{4,8} = 57211.12$, p < 0.00001) and use them to calculate a negative genuine effect (-1.031) beyond publication selection. However, the reported elasticities are conditional on a number of factors. Specifically, sectoral-level analyses typically report less negative elasticities by an order of 1.379 for the manufacturing and 1.616 for the service sector. Similarly, evidence after 2001 is less negative on average (0.262), which also holds for regional-level analyses (1.472).

Taken together, the multivariate analysis of the labor market effects from trade determined that only the literature of the employment elasticities is contaminated by publication selection. Nevertheless, once we account for this bias, the PET and PEESE tests indicated that trade –on average- benefits both wages and employment. Therefore, our results imply that the beneficial effects from trade expand on both wages and employment. Nevertheless, the positive effects are reversed when industrialized countries trade with China. Interestingly, we also indicated that the skill-biased nature of the internationalization of production is reflected in both wages and employment. High-skilled workers enjoy larger positive wage and employment effects from trade compared to low- and medium-skilled ones. Furthermore, differential effects between wages and employment were uncovered for elasticities in the manufacturing sector. In contrast, our analysis concluded that both the wage and the employment elasticities in studies after 2001 tend to be smaller.

6. Conclusions and discussion

In the last four decades, high-income countries have experienced a relative decline in manufacturing employment, accompanied by a decline in the wage share of medium- and low-skilled employees. The explanations point towards domestic technological progress and international trade as the principal causes for these developments. However, theory stresses that there are winners and losers from globalization and automation. A meta-analysis helps to synthesize the evidence base, appraise competing claims in the theoretical literature and pinpoint what factors determine positive outcomes and what factors determine the negative effects. The present meta-study evaluates the labor market effects from technology and trade by analyzing a sample of 1717 elasticities extracted from 91 studies. The study design differs along many dimensions. Some studies focus on wages others on employment. Some studies differentiate between skill levels of the workforce, while the data aggregation often varies between industry- versus firm- or worker-level ones. We codify these aspects of heterogeneity into control variables. To offer a genuine understanding of contemporary and much debated topics in economics, we apply robust, multivariate MRA techniques in the form of weighted least squares models, which address both the within- and the between-study variation.

Our meta-analysis confirms the consensus that publication selection is a concern in the economics literature. In our case, we uncover significant negative publication bias in the samples of the wage and employment elasticities from technology and the sample of the employment elasticities from trade. At the same time, the analysis indicates that only the literature of wage elasticities from trade is not contaminated by publication selection. Nevertheless, we document significant, economically meaningful and often overlapping

genuine effects once we account for publication bias. On average, technology and trade benefit both wages and employment.

More importantly, our comprehensive approach provided robust evidence regarding how the granularity in the methodology and the study design influences the reported elasticities. Several results stand out. First, we uncover that the skill-bias effects from technology are concentrated on employment, where technology benefits high-skilled workers relatively more compared to low-skilled ones. Similarly, trade benefits the employment of high-skilled workers more than low-skilled ones. However, the differential impact from trade expands also on wages, where it benefits the wages of skilled workers relatively more compared to unskilled ones. Taken together, the analysis reveals both overlapping and unique effects from technology and trade, while they jointly contribute to the skill-biased nature of modern labor market developments.

Our analysis also indicates that the wage effects from technology decrease over time, while the employment ones are larger at the regional level. Considering trade, we reveal relatively similar evolution between the wage and the employment effects; both tend to be less positive after the early 2000s. Furthermore, we also uncover that employment elasticities from trade are on average larger at the regional level, implying more beneficial local effects, thus contrasting a popular view in the literature that globalization of production is stigmatized as the main culprit explaining the decay of local labor markets (Rodrik, 2018). Finally, we reveal that trade between developed countries benefits the wages in the industrialized countries.

The current work increases our understanding of which factors drive the labor market consequences of automation and trade.

References

- Acemoglu, D., Autor, D., Dorn, D., Hanson, G. H., & Price, B. (2016). Import competition and the great US employment sag of the 2000s. *Journal of Labor Economics*, 34 (S1), S141–S198.
- Aguirregabiria, V., & Alonso-Borrego, C. (2001). Occupational structure, technological innovation, and reorganization of production. *Labour Economics*, 8 (1), 43–73.
- Amiti, M., & Wei, S.J. (2009). Service offshoring and productivity: Evidence from the US. *World Economy*, *32* (2), 203–220.
- Autor, D. H., Dorn, D., & Hanson, G. H. (2013). The China syndrome: Local labor market effects of import competition in the United States. *American Economic Review*, *103* (6), 2121–68.

- Autor, D. H., Levy, F., & Murnane, R. J. (2003). The skill content of recent technological change: An empirical exploration. *The Quarterly Journal of Economics*, *118* (4), 1279-1333.
- Baldwin, R. E., & Hilton, R. S. (1984). A technique for indicating comparative costs and predicting changes in trade ratios. *The Review of Economics and Statistics*, 105–110.
- Becker, S. O., Ekholm, K., & Muendler, M.-A. (2013). Offshoring and the onshore composition of tasks and skills. *Journal of International Economics*, 90 (1), 91–106.
- Berman, E., Bound, J., & Griliches, Z. (1994). Changes in the demand for skilled labor within us manufacturing: Evidence from the annual survey of manufactures. *The Quarterly Journal of Economics*, 109 (2), 367–397.
- Berman, E., Bound, J., & Machin, S. (1998). Implications of skill-biased technological change: International evidence. *The Quarterly Journal of Economics*, *113* (4), 1245–1279.
- Bernard, A. B., & Wagner, J. (1997). Exports and success in German manufacturing. *Weltwirtschaftliches Archiv*, 133 (1), 134–157.
- Biscourp, P., & Kramarz, F. (2007). Employment, skill structure and international trade: Firmlevel evidence for France. *Journal of International Economics*, 72 (1), 22–51.
- Blechinger, D., Kleinknecht, A., Licht, G., & Pfeiffer, F. (1998). *The impact of innovation on employment in Europe: An analysis using CIS data*. ZEW-Dokumentation No. 98-02, Manheim.
- Borjas, G. J., Freeman, R. B., & Katz, L. F. (1997). How much do immigration and trade affect labor market outcomes? *Brookings papers on economic activity*, (1), 1–90.
- Cette, G., Fernald, J., & Mojon, B. (2016). The pre-Great Recession slowdown in productivity. *European Economic Review*, 88, 3–20.
- Charemza, W. W., & Deadman, D. F. (1997). *New directions in econometric practice*. Edward Elgar Publishing.
- Costinot, A., & Vogel, J. (2010). Matching and inequality in the world economy. *Journal of Political Economy*, *118* (4), 747–786.
- Crino, R. (2009). Offshoring, multinationals and labour market: A review of the empirical literature. *Journal of Economic Surveys*, 23 (2), 197–249.
- Dauth, W., Findeisen, S., & Suedekum, J. (2014). The rise of the East and the Far East: German labor markets and trade integration. Journal of the European Economic Association, 12 (6), 1643–1675.
- De Groot, H. L., Poot, J., & Smit, M. J. (2016). Which agglomeration externalities matter most and why? *Journal of Economic Surveys*, 30 (4), 756–782.
- Debaere, P., Lee, H., & Lee, J. (2006). *Does where you go matter? The impact of outward foreign direct investment on multinationals' employment at home*. Centre for Economic Policy Research, Discussion Paper, No. 5737.

- Disdier, A.-C., & Head, K. (2008). The puzzling persistence of the distance effect on bilateral trade. *The Review of Economics and Statistics*, 90 (1), 37–48.
- Doucouliagos, H. (2011). *How large is large? Preliminary and relative guidelines for interpreting partial correlations in economics*. Deakin University, Department of Economics.
- Doucouliagos, H., & Stanley, T. D. (2009). Publication selection bias in minimum-wage research? A meta-regression analysis. *British Journal of Industrial Relations*, 47 (2), 406–428.
- Doucouliagos, H., & Stanley, T. D. (2013). Are all economic facts greatly exaggerated? Theory, competition and selectivity. *Journal of Economic Surveys*, 27 (2), 316–339.
- Egger, M., Smith, G. D., Schneider, M., & Minder, C. (1997). Bias in meta-analysis detected by a simple, graphical test. *BMJ*, *315* (7109), 629–634.
- Ekholm, K., & Hakkala, K. (2006). *The effect of offshoring on labour demand: Evidence from Sweden*. Centre for Economic Policy Research, Discussion Paper, No. 5648.
- Feenstra, R. C. (2016). *Advanced international trade: Theory and evidence*. Princeton University Press.
- Feenstra, R. C., & Hanson, G. H. (1999). The impact of outsourcing and high-technology capital on wages: Estimates for the united states, 1979–1990. *The Quarterly Journal of Economics*, 114 (3), 907–940.
- Fisher, R. A. (1932). Statistical methods for research workers. London: Oliver & Boyd.
- Geishecker, I. (2008). The impact of international outsourcing on individual employment security: A micro-level analysis. *Labour Economics*, 15 (3), 291–314.
- Gordon, R. J. (2016). The rise and fall of American growth. Princeton University Press.
- Grossman, G. M., & Rossi-Hansberg, E. (2008). Trading tasks: A simple theory of offshoring. *American Economic Review*, *98* (5), 1978–97.
- Havranek, T., & Irsova, Z. (2011). Estimating vertical spillovers from FDI: Why results vary and what the true effect is. *Journal of International Economics*, 85 (2), 234–244.
- Havranek, T., Stanley, T., Doucouliagos, H., Bom, P., Geyer-Klingeberg, J., Iwasaki, I., Reed, W. R., Rost, K., & Van Aert, R. (2020). Reporting guidelines for meta-analysis in economics. *Journal of Economic Surveys*, 34 (3), 469–475.
- Hedges, L. V., & Olkin, I. (1985). Statistical methods for meta-analysis. Academic Press.
- Helpman, E. (2009). The mystery of economic growth. Harvard University Press.
- Helpman, E. (2018). Globalization and inequality. Harvard University Press.
- Higgins, J. P., Thompson, S. G., Deeks, J. J., & Altman, D. G. (2003). Measuring inconsistency in meta-analyses. *BMJ*, 327 (7414), 557–560.

- Hijzen, A. (2005). A bird's eye view of international outsourcing: Data, measurement and labour demand effects. *Economie Internationale*, (4), 45–63.
- Hummels, D., Munch, J. R., & Xiang, C. (2018). Offshoring and labor markets. *Journal of Economic Literature*, *56* (*3*), 981–1028.
- Jones, R. W. (1965). The structure of simple general equilibrium models. *Journal of Political Economy*, 73 (6), 557–572.
- Jorgenson, D. W., Ho, M. S., & Stiroh, K. J. (2008). A retrospective look at the US productivity growth resurgence. *Journal of Economic Perspectives*, 22 (1), 3–24.
- Katz, L. F., Loveman, G. W., & Blanchflower, D. G. (1995). A comparison of changes in the structure of wages in four OECD countries. Differences and changes in wage structures (pp. 25–66). University of Chicago Press.
- Klein, M. W., Schuh, S., & Triest, R. K. (2003). *Job creation, job destruction, and international competition*. WE Upjohn Institute.
- Krugman, P. (2000). Technology, trade and factor prices. *Journal of International Economics*, 50 (1), 51–71.
- Krugman, P. (2008). Trade and wages, reconsidered. *Brookings Papers on Economic Activity*, (1), 103–154.
- Krugman, P., Cooper, R. N., & Srinivasan, T. (1995). Growing world trade: Causes and consequences. *Brookings Papers on Economic Activity*, (1), 327–377.
- Lawrence, R. Z., & Slaughter, M. J. (1993). International trade and American wages in the 1980s: Giant sucking sound or small hiccup? *Brookings Papers on Economic Activity*. *Microeconomics*, (2), 161–226.
- Leamer, E. E. (1994). *Trade, wages and revolving door ideas (tech. rep.)*. National Bureau of Economic Research Cambridge, USA.
- Leamer, E. E. (2000). What's the use of factor contents? *Journal of International Economics*, 50 (1), 17–49.
- Leamer, E. E. (2010). In search of Stolper-Samuelson linkages between international trade and lower wages. *Imports, Exports, and the American Worker*, 141-203.
- Melo, P. C., Graham, D. J., & Noland, R. B. (2009). A meta-analysis of estimates of urban agglomeration economies. *Regional Science and Urban Economics*, 39 (3), 332–342.
- Mincer, J. (1974). Schooling, experience, and earnings. *Human Behavior & Social Institutions, No. 2.*
- Moher, D., Liberati, A., Tetzlaff, J., Altman, D. G., & Group, P. (2009). Preferred reporting items for systematic reviews and meta-analyses: The prisma statement. *Annals of Internal Medicine*, 151 (4), 264–269.

Moreno, S. G., Sutton, A. J., Ades, A., Stanley, T. D., Abrams, K. R., Peters, J. L., & Cooper,

N. J. (2009). Assessment of regression-based methods to adjust for publication bias through a comprehensive simulation study. *BMC Medical Research Methodology*, 9 (1), 1-17.

- Pearson, K. (1904). Report on certain enteric fever inoculation statistics. *British Medical Journal*, *2*, 1243–1246.
- Petticrew, M., & Roberts, H. (2008). *Systematic reviews in the social sciences: A practical guide*. John Wiley & Sons.
- Piva, M., Santarelli, E., & Vivarelli, M. (2005). The skill bias effect of technological and organizational change: Evidence and policy implications. *Research Policy*, *34* (2), 141–157.
- Rodrik, D. (2018). Populism and the economics of globalization, *Journal of International Business Policy*, *1*, 12-33.
- Sachs, J. D., & Shatz, H. J. (1996). *International trade and wage inequality in the United States: Some new results* (Development Discussion Papers, No. 524). Harvard Institute for International Development.
- Sala-i-Martin, X. (1997). I just ran two million regressions. *American Economic Review*, 87 (2), 178–183.
- Stanley, T. D. (2008). Meta-regression methods for detecting and estimating empirical effects in the presence of publication selection. Oxford Bulletin of Economics and statistics, 70 (1), 103–127.
- Stanley, T. D., & Doucouliagos, H. (2012). *Meta-regression analysis in economics and business*. Routledge Advances in Research Methods.
- Stanley, T. D., & Doucouliagos, H. (2014). Meta-regression approximations to reduce publication selection bias. *Research Synthesis Methods*, 5 (1), 60–78.
- Stanley, T. D., Jarrell, S. B., & Doucouliagos, H. (2010). Could it be better to discard 90% of the data? A statistical paradox. *The American Statistician*, 64 (1), 70–77.
- Strauss-Kahn, V. (2004). The role of globalization in the within-industry shift away from unskilled workers in France. Challenges to Globalization: Analyzing the Economics (pp. 209–234). University of Chicago Press.
- Ugur, M., Awaworyi Churchill, S., & Solomon, E. (2018). Technological innovation and employment in derived labour demand models: A hierarchical meta-regression analysis. *Journal of Economic Surveys, 32 (1)*, 50–82.
- Van Reenen, J. (1997). Employment and technological innovation: Evidence from UK manufacturing firms. *Journal of Labor Economics*, 15 (2), 255–284.
- Weichselbaumer, D., & Winter-Ebmer, R. (2005). A meta-analysis of the international gender wage gap. *Journal of Economic Surveys*, 19 (3), 479–511.

Winkler, D. (2010). Services offshoring and its impact on productivity and employment:

Evidence from Germany, 1995-2006. The World Economy, 33(12), 1672-1701.

- Wood, A. (1995). How trade hurt unskilled workers. *The Journal of Economic Perspectives*, 9 (3), 140–168.
- Wooldridge, J. M. (2015). *Introductory econometrics: A modern approach*. Cengage learning, 3rd Edition, Thomson/South-Western, Mason.
- Wright, G. C. (2014). Revisiting the employment impact of offshoring. *European Economic Review*, *66*, 63–83.

List of figures:

- Figure 1. Overview of the literature search
- Figure 2. Kernel density distributions
- Figure 3. Technology elasticities by skill type
- Figure 4. Trade elasticities by skill type
- Figure 5. Funnel plots of technology elasticities
- Figure 6. Funnel plots of trade elasticities

Online Appendix

Labor markets, trade and technological progress. A meta-study

Nikolaos Terzidis, Steven Brakman and Raquel Ortega-Argiles

Table of contents

Appendix A. Literature search and selected studies
Appendix A1. Literature search
Appendix A2. Reference list of selected studies
Appendix B. Preliminary evidence12
Appendix C. Publication bias (univariate FAT-PET-PEESE analysis)
Appendix D. Robustness analysis14
Appendix D1. Outlier analysis14
Appendix D2. Estimation outcomes excluding semi-elasticities
Appendix D3. Estimation outcomes excluding calculated elasticities and semi-elasticities
Appendix D4. Estimation outcomes excluding elasticities where the dependent variable is expressed in shares
Appendix D5. Estimation outcomes excluding semi-elasticities, calculated elasticities and elasticities where the dependent variable is expressed in shares and winsorizing
Appendix E. Random and fixed effects analysis 50
Appendix E1. Random effects meta-regression analysis
Appendix E2. Fixed effects meta-regression analysis (MRA)
Appendix F. Partial correlation coefficients. Theoretical principles and empirical results
Appendix F1. Construction and summary statistics of partial correlation coefficients
Appendix F2. Meta-regression analysis outcomes based on partial correlation coefficients
Appendix G. Collective robustness analysis (summary tables, only WLS general-to-specific estimation outcomes)
Appendix H. Best practice estimates
Reference list

Appendix A. Literature search and selected studies

Appendix A1. Literature search

Figure 1 in the main text reports the outcomes of the literature search based on multiple search queries including a wide set of relevant terms (Table A1.1) to three multi-disciplinary databases (*EconLit, Web of Science* and *Google Scholar*). Six research terms were related to technology and 6 to international trade, combined with Boolean operators (AND, OR) and wild card symbols (*) to approximate alternative word endings. We confined the literature search to English studies published between 1970 and 2022.

General terms	Technology	International Trade
employment, wages, elasticity,	automation, computerization,	Trade, outsourcing,
regression analysis	innovation, research and	offshoring, foreign direct
	development, technical	investment (FDI), export
	change, robotization	(import) competition
regression analysis	innovation, research and development, technical change, robotization	offshoring, foreign direct investment (FDI), export (import) competition

Table A1.1 Search Terms

Considering EconLit, each search string resulted in 19.000 studies on average. After inspecting the outcomes for the first two queries and realizing that no new records were retained from the second search string, we inspected the first 3.000 studies in each of the following queries, after which the relevance dropped sharply. Due to its inter-disciplinary nature, *Web of Science* provided on average 22.000 results for each research string. Similarly to the *EconLit* database, we inspected the first technology and trade queries in detail and then confined ourselves to the first 3.000 studies. Finally, *Google Scholar* generated many references (even exceeding 2.000.000 studies in the broader research string). Therefore, we restricted the identification stage to the first 3.000 studies by research query.

In the screening stage, we excluded several studies that failed to satisfy the inclusion criteria (Table A1.2). The latter are designed to ensure that equilibrium employment or wage effects in developed countries are obtained by means of robust regression analysis identifying causal effects. Notably, the dependent variable should be a continuous measure of employment or wages. We excluded studies on productivity, probability of losing one's job or plant survival studies, where the independent variable is expressed in ratios (skilled-to-unskilled), studies based on dummy indicators of engaging in technological innovations or participating in the

INCLUDED

- Studies on equilibrium employment or wage effects (in levels or changes)
- Studies using robust econometric techniques (regression analysis establishing causation or correlation)
- Studies where automation and/or trade are approximated by a continuous variable
- Labor market effects in developed countries

NOT INCLUDED

- Studies on productivity, employment security (probability of becoming jobless), wage or employment differential (e.g. dependent variable is the skilled-tounskilled wage or employment ratio), organizational change (decentralization, delayering of managerial practices etc.)
- Trade exposure measured in monetary terms (e.g. import prices)
- Studies where automation and/or trade are approximated by a dummy or count variables (e.g. based on responses regarding the number of technologies adopted)
- Estimates from interaction terms (e.g. outsourcing*high-skill dummy)
- Studies based on non-parametric / descriptive / match sampling etc. techniques
- Quantile or binary (probit and logit) regression outcomes
- Effects on developing countries

international markets and estimates from interaction terms. Furthermore, we excluded studies based on the mean or matching sampling comparisons because they do not provide an immediate effect of technology and/or trade. In contrast, such studies mostly compare the characteristics of different firms/sectors due to differential exposure to technology and trade. Finally, we excluded estimates based on quantile regression analysis or binary regression outcomes (probit or logit).

Appendix A2. Reference list of selected studies

- Acemoglu, D., Autor, D., Dorn, D., Hanson, G. H. and Price, B. (2016). Import competition and the great US employment sag of the 2000s. *Journal of Labor Economics*, 34(1 - pt 2), S141 - S198.
- Acemoglu, D. and Restreppo, P. (2020). Robots and jobs: Evidence from the US labor markets. *Journal of Political Economy*, 128(6): 2188-2244.
- Adams, J.D. (1999). The structure of firm R&D, the factor intensity of production and skill bias. *The Review of Economics and Statistics*, *81(3)*, 499-510.
- Agovino, M., Aldieri, L. Garofalo, A. & Vinci, C.P. (2018). R&D spillovers and employment. Evidence from European patent data. *Empirica*, 45(2), 247-260.
- Aguirregabiria, V. & Alonso-Borego, C. (2001). Occupational structure, technological innovation and reorganization of production. *Labour Economics*, *8*(1), 43-73.
- Aldieri, L. & Vinci, C.P. (2018). Innovation effects on employment in high-tech and low-tech industries: Evidence from large international trends within the triad. *Eurasian Business Review*, 8(2), 229-243.
- Allen, S.G. (2001). Technology and the wage structure. *Journal of Labor Economics*, 19(2), 440-483.
- Amiti, M. & Wei, S.J. (2005). Service offshoring, productivity and employment: Evidence from the United States. IMF Working Paper, 05/238.
- Amiti, M. & Wei, S.J. (2009) Service offshoring and productivity: Evidence from the US. World Economy, 32(2): 203-220.
- Amiti, M., Wei, S.J., Haskel, J. & Auriol, E. (2005). Fear of service offshoring. Is it justified? *Economic Policy*, 20(42), 307-347.
- Anderton, B., & Brenton, P. (1999). Outsourcing and low-skilled workers in the UK. *Bulletin* of Economic Research, 51 (4), 267–286.
- Anderton, B., Brenton, P., & Oscarsson, E. (2002). *Outsourcing and inequality*. CEPS Working Document No. 187, October 2002.
- Ariu, A., Hakkala, K. N., Jensen, J. B., & Tamminen, S. (2019). Service Imports, Workforce Composition, and Firm Performance: Evidence from Finnish Microdata (February 18, 2022). Available at SSRN: https://ssrn.com/abstract=4044315.
- Ashournia, D., Munch, J. R., & Nguyen, D. (2006). The impact of Chinese import penetration on Danish firms and workers. Institute for the Study of Labor (IZA) Discussion Papers, DP No. 10033.

- Autor, D. H., Dorn, D., & Hanson, G. H. (2013). The China syndrome: Local labor market effects of import competition in the United States. *American Economic Review*, 103(6), 2121–68.
- Autor, D. H., Katz, L. F., & Krueger, A. B. (1998). Computing inequality: Have computers changed the labor market? *The Quarterly Journal of Economics*, *113*(4), 1169–1213.
- Balsvik, R., Jensen, S., & Salvanes, K. G. (2015). Made in China, sold in Norway: Local labor market effects of an import shock. *Journal of Public Economics*, *127*, 137–144.
- Bamieh, O., Bripi, F., & Fiorini, M. (2022). Services trade and labor market outcomes: Evidence from Italian firms. *Review of International Economics*, *30*(*3*), 673–701.
- Barbieri, L., Piva, M., & Vivarelli, M. (2019). R&D, embodied technological change, and employment: Evidence from Italian microdata. *Industrial and Corporate Change*, 28(1), 203–218.
- Bartel, A. P., & Lichtenberg, F. R. (1991). The age of technology and its impact on employee wages. *Economics of Innovation and New Technology*, *1*(*3*), 215–231.
- Bartel, A. P., & Sicherman, N. (1999). Technological change and wages: An interindustry analysis. *Journal of Political Economy*, *107*(2), 285–325.
- Baumgarten, D., Geishecker, I., & Gorg, H. (2013). Offshoring, tasks, and the skill-wage pattern. *European Economic Review*, *61*, 132–152.
- Bella, M., & Quintieri, B. (2000). The effect of trade on employment and wages in Italian industry. *Labour*, *14*(2), 291–310.
- Bernard, A. B., Jensen, J. B., & Schott, P. K. (2006). Survival of the best fit: Exposure to lowwage countries and the (uneven) growth of US manufacturing plants. *Journal of International Economics*, 68(1), 219–237.
- Berndt, E. R., & Morrison, C. J. (1995). High-tech capital formation and economic performance in US manufacturing industries. An exploratory analysis. *Journal of Econometrics*, 65 (1), 9–43.
- Bertoli, S. (2008). The impact of material offshoring on employment in the Italian manufacturing industries: The relevance of inter-sectoral effects. Centro Studi Luca d'Agliano Development Studies Working Paper No 244.
- Bloom, N., Draca, M., & Van Reenen, J. (2016). Trade induced technical change? the impact of Chinese imports on innovation, IT and productivity. *The Review of Economic Studies*, 83(1), 87–117.
- Bogliacino, F. (2014). Innovation and employment: A firm level analysis with European R&D scoreboard data. *EconomiA*, *15*(2), 141–154.

- Bogliacino, F., & Pianta, M. (2010). Innovation and employment: A reinvestigation using revised Pavitt classes. *Research Policy*, 39(6), 799–809.
- Bogliacino, F., Piva, M., & Vivarelli, M. (2012). R&D and employment: An application of the LSDVC estimator using European microdata. *Economics Letters*, *116*(*1*), 56–59.
- Bogliacino, F., & Vivarelli, M. (2010). The job creation effect of R&D expenditures. *Australian Economic Papers*, *51*(2), 96-113.
- Broccolini, C., Lo Turco, A., Presbitero, A. F., & Staffolani, S. (2011). Individual earnings, international outsourcing and technological change: Evidence from Italy. *International Economic Journal*, 25(1), 29–46.
- Buerger, M., Broekel, T., & Coad, A. (2012). Regional dynamics of innovation: Investigating the co-evolution of patents, research and development (R&D), and employment. *Regional Studies*, 46(5), 565–582.
- Cadarso, M.A., Gomez, N., Lopez, L.A. & Tobarra, M.A. (2008). The EU enlargement and the impact of outsourcing on industrial employment in Spain, 1993-2003. *Structural Change and Economic Dynamics*, 19, 95-108.
- Ciarli, T., Marzucchi, A., Salgado, E., & Savona, M. (2018). The effect of R&D growth on employment and self-employment in local labour markets. Science Policy Reserch Unit Working Paper Series, Vol SWPS 2018-08.
- Crino, R. (2012). Service offshoring and the skill composition of labour demand. *Oxford Bulletin of Economics and Statistics*, *74*(1), 20–57.
- Criscuolo, C., & Garicano, L. (2010). Offshoring and wage inequality: Using occupational licensing as a shifter of offshoring costs. *American Economic Review*, *100*(2), 439–43.
- Dauth, W., Findeisen, S., Suedekum, J., & Woessner, N. (2018). Adjusting to robots: Workerlevel evidence. Opportunity and Inclusive Growth Institute Working Papers, Paper No. 13.
- Dixon, J., Hong, B., & Wu, L. (2020). *The employment consequences of robots: Firm-level evidence*. Statistics Canada Ontario.
- Doms, M., Dunne, T., & Troske, K. R. (1997). Workers, wages, and technology. *The Quarterly Journal of Economics*, *112(1)*, 253–290.
- Donoso, V., Martin, V., & Minondo, A. (2015). Do differences in the exposure to Chinese imports lead to differences in local labour market outcomes? An analysis for Spanish provinces. *Regional Studies*, 49(10), 1746–1764.
- Dosi, G., Piva, M., Virgillito, M. E., & Vivarelli, M. (2021). Embodied and disembodied

technological change: The sectoral patterns of job-creation and job-destruction. *Research Policy*, *50*(*4*), 104–199.

- Dunne, T., Haltiwanger, J., & Troske, K. R. (1997). Technology and jobs: Secular changes and cyclical dynamics. Carnegie-Rochester Conference Series on Public Policy, 46, 107– 178.
- Ebenstein, A., Harrison, A., McMillan, M., & Phillips, S. (2014). Estimating the impact of trade and offshoring on American workers using the current population surveys. *The Review of Economics and Statistics*, *96*(*4*), 581–595.
- Ekholm, K., & Hakkala, K. (2008). *The effect of offshoring on labour demand: Evidence from Sweden*. Centre for Economic Policy Research (CEPR) Discussion Paper No. 5648.
- Eppinger, P. S. (2019). Service offshoring and firm employment. *Journal of International Economics*, 117, 209–228.
- Falk, M., & Koebel, B. M. (2004). The impact of office machinery, and computer capital on the demand for heterogeneous labour. *Labour Economics*, *11*(*1*), 99–117.
- Falk, M., & Wolfmayr, Y. (2005). Employment effects of trade in intermediate inputs with the EU member states and asia. *Aussenwirtschaft*, 60(4), 441–461.
- Feenstra, R. C., Ma, H., & Xu, Y. (2019). US exports and employment. *Journal of International Economics*, 120, 46–58.
- Fluckiger, M., & Ludwig, M. (2015). Chinese export competition, declining exports and adjustments at the industry and regional level in Europe. *Canadian Journal of Economics/Revue Canadienne d' Économique*, 48(3), 1120–1151.
- Fuster, B., Lillo-Banuls, A. & Martinez-Mora, C. (2019). The effects of service offshoring on employment. *Structural Change and Economic Dynamics*, 51, 529–538.
- Garcia, A., Jaumandreu, J., & Rodriguez, C. (2005). Innovation and jobs: Evidence from manufacturing firms. Part of the project "Innovation and employment in European Firms" supported by the European Commission.
- Geishecker, I., & Gorg, H. (2008). Winners and losers: A micro-level analysis of international outsourcing and wages. *Canadian Journal of Economics / Revue Canadienne* d'Economique, 41(1), 243–270.
- Geishecker, I., & Gorg, H. (2013). Services offshoring and wages: Evidence from micro data. *Oxford Economic Papers*, 65(1), 124–146.
- Geishecker, I., Gorg, H., & Munch, J. R. (2010). Do labour market institutions matter? microlevel wage effects of international outsourcing in three European countries. *Review of World Economics*, 146(1), 179–198.

- Gera, S., Gu, W., & Lin, Z. (1999). *Technology and the demand for skils*. *An industry-level analysis*. Industry Canada Research Publications Programme, Working Paper No. 28.
- Gorg, H., & Hanley, A. (2005). Labour demand effects of international outsourcing: Evidence from plant-level data. *International Review of Economics & Finance*, *14*(*3*), 365–376.
- Greenaway, D., Hine, R. C., & Wright, P. (1999). An empirical assessment of the impact of trade on employment in the United Kingdom. *European Journal of Political Economy*, 15(3), 485–500.
- Greenhalgh, C., Longland, M., & Bosworth, D. (2001). Technological activity and employment in a panel of UK firms. *Scottish Journal of Political Economy*, 48(3), 260–282.
- Hansson, P., & Lundin, N. N. (2004). Exports as an indicator on or promoter of successful
 Swedish manufacturing firms in the 1990s. *Review of World Economics*, 140(3), 415–445.
- Harrison, A., & McMillan, M. (2011). Offshoring jobs? Multinationals and US manufacturing employment. *The Review of Economics and Statistics*, *93*(*3*), 857–875.
- Head, K., & Ries, J. (2002). Offshore production and skill upgrading by Japanese manufacturing firms. *Journal of International Economics*, *58*(*1*), 81–105.
- Helg, R., & Tajoli, L. (2005). Patterns of international fragmentation of production and the relative demand for labor. *The North American Journal of Economics and Finance*, 16 (2), 233–254.
- Hijzen, A., Gorg, H., & Hine, R. C. (2005). International outsourcing and the skill structure of labour demand in the United Kingdom. *The Economic Journal*, 115(506), 860–878.
- Hummels, D., Jørgensen, R., Munch, J., & Xiang, C. (2014). The wage effects of offshoring: Evidence from Danish matched worker-firm data. *American Economic Review*, 104(6), 1597–1629.
- Konings, J., & Murphy, A. P. (2006). Do multinational enterprises relocate employment to lowwage regions? Evidence from European multinationals. *Review of World Economics*, 142(2), 267-286.
- Kramarz, F. (2008). *Offshoring, wages, and employment: Evidence from data matching imports, firms, and workers*. Centre de Recherche en 'Economie et Statistique.
- Lachenmaier, S., & Rottmann, H. (2011). Effects of innovation on employment: A dynamic panel analysis. *International Journal of Industrial Organization*, 29(2), 210–220.
- Lassmann, A., & Spinelli, F. (2020). Services trade and labour market outcomes in the United Kingdom. OECD Trade Policy Papers, No. 243.
- Liu, R., & Trefler, D. (2008). Much ado about nothing: American jobs and the rise of service

outsourcing to China and India. National Bureau of Economic Research, Paper No. 14061.

- Liu, R., & Trefler, D. (2019). A sorted tale of globalization: White collar jobs and the rise of service offshoring. *Journal of International Economics*, *118*, 105–122.
- Lo Turco, A., & Maggioni, D. (2012). Offshoring to high and low income countries and the labor demand. Evidence from Italian firms. *Review of International Economics*, 20(3), 636–653.
- Magli, M. (2021). The direct and indirect effect of services offshoring on local labour market outcomes. CESifo Working Papers, WP. 8413-2020.
- Malgouyres, C. (2017). The impact of Chinese import competition on the local structure of employment and wages: Evidence from France. *Journal of Regional Science*, 57(3), 411–441.
- Martins, P. S., & Opromolla, L. D. (2009). Exports, imports and wages: Evidence from matched firm-worker-product panels. Institute for the Study of Labor (IZA) Discussion Papers, DP No. 4646.
- Michel, B. & Rycx, Fr. (2012). Does offshoring of materials and business services affect employment? Evidence from a small open economy. *Applied Economics*, 44(2), 229-251.
- Munch, J. R., & Skaksen, J. R. (2009). Specialization, outsourcing and wages. *Review of World Economics*, 145(1), 57–73.
- Nedelkoska, L., & Wiederhold, S. (2010). Technology, outsourcing, and the demand for heterogeneous labor: Exploring the industry dimension. Jena Economic Research Papers (No. 2010, 052).
- Neves Sequeira, T., Garrido, S., & Santos, M. (2021). Robots are not always bad for employment and wages. *International Economics*, *167*, 108–119.
- Nilsson Hakkala, K., & Huttunen, K. (2016). *Worker-level consequences of import shocks*. Institute for the Study of Labor (IZA) Discussion Papers, DP No. 10033.
- Pereira, T. (2016). *The effect of developing countries' competition on regional labour markets in Portugal*. Gabinete de Estrategia e Estudos, Ministerio da Economia, Number 58.
- Piva, M., & Vivarelli, M. (2018). Technological change and employment: Is Europe ready for the challenge? *Eurasian Business Review*, 8(1), 13–32.
- Schank, T., Schnabel, C., & Wagner, J. (2007). Do exporters really pay higher wages? First evidence from German linked employer–employee data. *Journal of International Economics*, 72(1), 52–74.

- Senses, M. Z. (2010). The effects of offshoring on the elasticity of labor demand. *Journal of International Economics*, *81(1)*, 89–98.
- Troske, K. R. (1999). Evidence on the employer size-wage premium from workerestablishment matched data. *Review of Economics and Statistics*, 81(1), 15–26.
- Van Reenen, J. (1997). Employment and technological innovation: Evidence from UK manufacturing firms. *Journal of Labor Economics*, 15(2), 255–284.
- Van Roy, V., Vertesy, D., & Vivarelli, M. (2018). Technology and employment: Mass unemployment or job creation? Empirical evidence from European patenting firms. *Research Policy*, 47(9), 1762–1776.
- Vivarelli, M., Evangelista, R., & Pianta, M. (1996). Innovation and employment in Italian manufacturing industry. *Research Policy*, 7(25), 1013–1026.
- Winkler, D. (2010). Services offshoring and its impact on productivity and employment: Evidence from Germany, 1995–2006. *The World Economy*, 33(12), 1672–1701.
- Wright, G. C. (2014). Revisiting the employment impact of offshoring. *European Economic Review*, 66, 63–83.
- Yun, L. (2008). Technical progress and labour demand in Swedish manufacturing firms. Journal of Industry, *Competition and Trade*, 8(2), 147–167.

Appendix B. Preliminary evidence

	Tech	nology	Trade		
	Wage	Employment	Wage	Employment	
	effects	effects	effects	effects	
	(1)	(2)	(3)	(4)	
Ton your impact factor	16231.94	-2519794	-76780.46	-32709.2	
Ten year impact factor	[41653.14]	[2458908]	[76839.82]	[26344.2]	
Constant	557500.6	$3.4*10^{7}$	1367618	488613.2	
Constant	[505528.2]	$[3.2*10^7]$	[1289454]	[327327.9]	
N [K]	231 [21]	392 [37]	459 <i>[34]</i>	635 [41]	
F-test (p-value)	0.700	0.312	0.325	0.222	

Table B1 Relationship between the collected elasticities and journal quality

Note: OLS regressions, dependent variable is the precision of the collected estimate, defined as its inverse st. error and the independent variable is the 10-year recursive impact factor of the primary study. N refers to the number of elasticities and K refers to the number of primary studies. Robust st. errors, clustered by primary study. * for p<0.1, ** for p<0.05 and *** for p<0.01

Table B2. Sample means and preliminary evidence on heterogeneity

			FEE		RI	EE	
	Observ.	Unweighted	Weighted	Weighted	T^2	I^2	Q-test
		Mean	Mean	Mean			(p-val)
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Panel A – Technology elasticities							
Entire sample	623	-0.021	0.282***	-0.038**	0.130	100%	0.000***
• Wage elasticities	231	-0.211	0.001***	-0.130**	0.423	100%	0.000***
• Employment elasticities	392	0.091	0.287***	-0.006	0.032	100%	0.000***
Panel B – Trade elasticities							
Entire sample	1094	0.003	0.023***	-0.027***	0.023	99.9%	0.000***
Wage elasticities	459	0.128	0.025***	0.000	0.002	99.9%	0.000***
• Employment elasticities	635	-0.088	0.020***	-0.053***	0.048	100%	0.000***

Note: Column 2 repeats the unweighted means from Table 1 (main text), to facilitate comparisons. Column 3 reports the outcomes from estimating the WLS form of the univariate FAT-PET model without a constant, where the weight is the inverse variance (fixed effects estimator). Columns 4-7 report the outcomes of similar random effects models. Column 5 reports the variation of the true genuine effect, column 6 reports the percentage of the between-estimate heterogeneity, besides the one attributed to sampling error. Column 7 reports the χ^2 value for Cochran's Q-test under the null hypothesis of no between-estimate heterogeneity. Similar outcomes for columns 3-7 are also obtained by means of the *meta summarize* command, as well as from WLS regressions of the estimated effect on their st. errors without constant, weighted by the inverse variance, based on robust st. errors clustered by primary study. * for \$p<0.1, ** for \$p<0.05 and *** for \$p<0.01

Appendix C. Publication bias (univariate FAT-PET-PEESE analysis)

Table C1 reports the outcomes of the univariate FAT-PET and PEESE analysis. The former is based on the WLS estimations of Equation C1, while the latter is based on the estimation of Equation C2.

$$e_i = \beta_0 + \beta_1 S E_{ei} + u_i \tag{C1}$$

$$e_i = \beta_0 + \beta_1 S E_{ei}^2 + u_i \tag{C2}$$

	Tech	nology	Trade		
	Wage	Employment	Wage	Employment	
	(1)	(2)	(3)	(4)	
Panel A – FAT-PET outcomes					
SE _i β_l (FAT)	0.057	-39.273***	-3.351***	-1.827	
	[0.932]	[9.194]	[0.807]	[3.748]	
Intercept β_0 (PET)	0.0005	0.287***	0.026***	0.021	
	[0.0003]	[0.0007]	[0.002]	[0.042]	
N [K]	231 [21]	392 [37]	459 [34]	635 [41]	
Panel B – PEESE outcomes					
Intercept γ_0 (PEESE)	-0.997	0.270	27.941***	-8.486	
	[0.737]	[0.363]	[0.015]	[6.540]	
N [K]	231 [21]	392 [37]	459 [34]	635 [41]	

Table C1. Univariate FAT-PET and PEESE analysis

Note: The table reports the outcomes of weighted least squares estimations (WLS), using the inverse of the variance as the optimal weight. N refers to the number of observations (elasticities) and K refers to the number of primary studies. Robust st. errors, clustered by primary study are reported in brackets. * for p<0.1, ** for p<0.05 and *** for p<0.01.

The univariate FAT-PET analysis (Panel A) uncovers evidence of significant publication selection only in the sample of employment elasticities from technology (Column 2) and the sample of wage elasticities from trade (Column 3). In both cases, the literature is contaminated by negative publication selection. Nevertheless, once we account for this publication bias, the analysis reveals evidence of significant and positive genuine effects. The PEESE test (Panel B) verifies the above positive genuine effect only in the wage elasticities from trade literature. However, the above models potentially suffer from omitted variable bias; therefore, more solid conclusions are based on the multivariate analysis reported in the main text.

Appendix D. Robustness analysis

Appendix D1. Outlier analysis

Table D1.1. Labor market effects from technology (Dependent variable: e_i)

	Wage elasticities			Employment elasticities			
Variable	WLS	WLS general to	PEESE	WLS	WLS general to	PEESE	
	(1)	(2)	(3)	(4)	(5)	(6)	
FAT – PET – PEESE tests	(_)		(-)	(-)		(1)	
Intercept (β_0) (PET) or (γ_0) (PEESE)	-2.198 [3.965]	0.421* [0.212]	-0.317** [0.131]	0.182 [0.634]	-0.298 [0.265]	-0.067 [0.368]	
St. error (β_1) or st. error ² (γ_1) (FAT)	-1.432 [1.840]	-1.936** [0.824]	0.956 [0.795]	-0.091 [1.689]	-0.544 [0.740]	-3.995 [3.462]	
Type of innovation (ref. process innovation	i)						
Product innovation	0.001 [0.001]	omitted	omitted	0.004** [0.002]	0.003** [0.001]	0.003** [0.001]	
Data aggregation (ref. worker-level)							
Industry-level	0.144 [0.181]	omitted	omitted	0.143 [0.105]	0.184** [0.074]	0.184** [0.076]	
Firm-level	0.982 [2.348]	omitted	omitted	0.937*** [0.081]	0.985*** [0.062]	1.013*** [0.073]	
Sectoral Ievel (ref. economy-wide)							
Manufacturing	-0.237 [0.155]	-0.287 [0.200]	-0.504*** [0.028]	0.241 [0.163]	0.243 [0.161]	0.240 [0.162]	
Services	-0.230 [0.155]	-0.279 [0.201]	-0.498*** [0.028]	[0.020]	[0.019]	[0.021]	
Skill-level (ref. varied skill)							
Low-skill	-0.006** [0.003]	omitted	omitted	0.086*** [0.022]	0.093*** [0.022]	0.084*** [0.022]	
Medium-skill	-0.004	omitted	omitted	0.104^{***}	0.108^{***}	0.100***	

	[0.003] -0.006			[0.021] 0 224***	[0.021] 0.225***	[0.020] 0.225***
High-skill	[0.006]	omitted	omitted	[0.024]	[0.023]	[0.024]
Other methodological aspects						
Technology and trade	-0.009*	-0.009**	-0.010***	0.018	omitted	omitted
(ref. only technology)	[0.005]	[0.004]	[0.0007]	[0.017]	ommeu	omitieu
Time series	-0.015	-0.013**	-0.007***	-4.147**	0.780***	0.548*
(ref. cross-sectional)	[0.011]	[0.005]	[0.0004]	[1.137]	[0.222]	[0.315]
Lagged independent variable	-0.009*	omitted	omitted	-0.014	omitted	omitted
(ref. contemporaneous ind. var.)	[0.005]	omnica	omnea	[0.019]	omnieu	omnica
Dependent variable in levels	-0.387*	-0.462**	-0.614***	0.042***	0.048^{***}	0.051***
(ref. first-differenced dep. var)	[0.193]	[0.176]	[0.028]	[0.008]	[0.003]	[0.003]
Controlling for endogeneity	-0.002	-0.002	-0.007**	0.001	omitted	omitted
(ref. no IV approach)	[0.008]	[0.002]	[0.002]	[0.005]	ommed	omnica
Wages in shares	2.383	omitted	omitted			
(ref. wages not in shares)	[4.930]	ommed	ommed			
Hours worked				0.0001***	0.0001***	0.0001***
(ref. number of workers)				$[5.2*10^{-10}]$	$[2.4*10^{-10}]$	$[3*10^{-10}]$
Publication characteristics (K-)						
Peer reviewed paper	1.886	omitted	omitted	-0.152	-0.771***	-0.765***
(ref. working paper)	[2.278]			[0.105]	[0.178]	[0.176]
Peer reviewed paper * st. error	-0.600	omitted	omitted	-1.402	omitted	omitted
	[2.044]	0.000	0.050	[1.327]		
Publication year	-0.767	-0.298	-0.973**	0.234*	0.614***	0.602***
5	[0.835]	[0.405]	[0.330]	[0.118]	[0.030]	[0.029]
Publication year * st. error	-1.298	-1.427*	-8.012***	-1.256**	-1.212*	-5.996
5	[0.905]	[0.724]	[0./83]	[0.570]	[0.659]	[5.072]
10-Year impact factor	-1.988	0.686	0.090	-7.952***	omitted	omitted
*	[3.301]	[0.654]	[0.315]	[1.516]		
10-Year impact factor * st. error	-3.133	-3.485*	-21.261***	-2.068	omitted	omitted
	[2.828]	[1.9/3]	[4.107]	[3.192]		
Geograpny (rej. US-UK-Canaaa-Japan)						

Europe	0.005 [0.771]	omitted	omitted	0.003 [0.005]	omitted	omitted
Spatial unit (ref. national-level analysis)						
Regional – level analysis	0.162 [0.707]	omitted	omitted	-0.591 [0.552]	-0.171 [0.208]	-0.486 [0.363]
Development status (ref. not-specified)						
Developed	omitted	omitted	omitted	omitted	omitted	omitted
Developing	0.619** [0.159]	0.631*** [0.128]	0.873*** [0.038]	0.198** [0.078]	0.228** [0.065]	0.277*** [0.056]
Time (ref. midpoint before 2001)						
After 2001	-0.323** [0.093]	-0.292*** [0.044]	-0.320*** [0.043]	0.070 [0.069]	omitted	omitted
Other controls						
Dataspan	-0.355*** [0.019]	-0.354*** [0.014]	-0.358*** [0.003]	0.256*** [0.004]	0.256*** [0.005]	0.259*** [0.008]
Number of RHS variables	0.00006	omitted	omitted	-0.0009 [0.002]	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important controls or</i> <i>fixed effects</i>)	-0.062 [0.083]	omitted	omitted	-0.092** [0.027]	-0.099* [0.051]	-0.111 [0.080]
Observations	193	193	193	328	328	328
Adjusted R^2	0.613	0.634	0.576	0.623	0.624	0.619
Primary study fixed effects	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes
Country by Midpoint fixed effects	yes	yes	yes	yes	ves	yes

Note: The table replicates the main analysis, after trimming the first and the last 5 percentiles from the distribution of the elasticities and the st. errors, to remove unreasonable values in the estimated effects and their st. errors (outliers). Robust st. errors, clustered by primary study are reported in brackets. All models are weighted by the inverse of the variance of the collected estimates. Columns 2 and 6 are based on the respective weighted-least-squares models (Columns 1 and 5), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories are significant at the 10% level. * for p<0.1, ** for p<0.05 and *** for p<0.01.

	Wage elasticities			Employment elasticities			
Variable	WLS	WLS	PEESE	WLS	WLS	PEESE	
		general to specific			general to specific		
	(1)	(2)	(3)	(4)	(5)	(6)	
FAT – PET – PEESE tests							
Intercept (β_0) (PET) or (γ_0) (PEESE)	-0.280 [3.095]	0.437* [0.212]	0.410** [0.154]	-0.078 [0.144]	0.125*** [0.032]	0.139** [0.030]	
St. error (β_1) or st. error ² (γ_1) (FAT)	-1.037 [1.186]	-1.237* [0.648]	-1.433 [1.484]	0.138 [0.400]	0.170 [0.252]	0.105 [0.108]	
Type of innovation (ref. process innovation	n)						
Product innovation	0.004 [0.004]	omitted	omitted	0.001*** [0.0002]	0.001*** [0.0002]	0.001*** [0.0002]	
Data aggregation (ref. worker-level)							
Industry-level	-0.027 [0.099]	omitted	omitted	-0.056 [0.048]	-0.021 [0.029]	-0.042* [0.025]	
Firm-level	0.120 [1.796]	omitted	omitted	0.237*** [0.041]	0.239*** [0.024]	0.239*** [0.023]	
Sectoral Ievel (ref. economy-wide)							
Manufacturing	-0.288** [0.087]	-0.268* [0.133]	-0.680*** [0.087]	0.077 [0.053]	0.075 [0.048]	0.074 [0.047]	
Services	-0.281** [0.087]	-0.259* [0.133]	-0.673*** [0.087]	0.033*** [0.003]	0.031*** [0.004]	0.029*** [0.002]	
Skill-level (ref. varied-skill)							
Low-skill	-0.011 [0.007]	omitted	omitted	0.017 [0.016]	0.016 [0.017]	0.012 [0.017]	
Medium-skill	-0.008 [0.006]	omitted	omitted	0.033** [0.011]	0.033** [0.011]	0.027** [0.011]	
High-skill	-0.009 [0.006]	omitted	omitted	0.140*** [0.006]	0.139*** [0.005]	[0.003]	
Other methodological aspects							

Table D1.2. Labor market effects from technology (Dependent variable: e_i)

Technology and trade	0.007**	0.006**	0.0007	0.014	amitta d	omitted
(ref. only technology)	[0.003]	[0.002]	[0.002]	[0.011]	omitiea	
Time series	-0.015**	-0.013**	-0.008***	-0.050***	-0.045***	-0.045***
(ref. cross – sectional)	[0.007]	[0.006]	[0.0006]	[0.011]	[0.002]	[0.00001]
Lagged independent variable	0.002	omitted	omitted	-0.007	omitted	omitted
(ref. contemporaneous)	[0.001]			[0.008]		
Dependent variable in levels	-0.387**	-0.392**	-0.725***	0.052	0.046***	-0.045***
(ref. first-differenced dep. var.)	[0.116]	[0.120]	[0.086]	[0.011]	[0.001]	[0.00001]
Controlling for endogeneity	-0.021**	-0.025***	-0.040***	0.009	omittad	omitted
(ref. no-IV approach)	[0.005]	[0.005]	[0.002]	[0.007]	omitieu	
Wages in shares	0.622	omitted	omitted			
(ref. wages not in shares)	[3.766]	omilieu	omilieu			
Hours worked				0.0001***	0.0001***	0.0001***
(ref. number of workers)				$[1.3*10^{-8}]$	$[3.5*10^{-12}]$	$[2.4*10^{-12}]$
Publication characteristics						
Peer reviewed paper	0.682	omitted	omittad	-0.193**	-0.161***	-0.165***
(ref. working paper)	[1.819]	omitieu	omilieu	[0.071]	[0.005]	[0.005]
Peer reviewed paper * st. error	-0.614	omitted	omitted	-0.251	omitted	omitted
	[1.267]			[0.486]	ommed	omittea
Publication year	-0.120	0.088	-0.720**	0.191***	0.187***	0.175***
	[0.646]	[0.361]	[0.326]	[0.044]	[0.015]	[0.015]
Publication year * st. error	-1.010*	-1.273**	-0.433	-0.414**	-0.378*	-0.872*
	[0.497]	[0.583]	[0.500]	[0.185]	[0.194]	[0.474]
10-Year impact factor	-0.672	0.118	0.046	0.554**	omitted	omitted
	[2.565]	[0.485]	[0.357]	[0.177]	omineu	
10-Year impact factor * st. error	-1.943	-1.843	-2.159	0.067	omitted	omitted
	[1.678]	[1.563]	[3.202]	[0.432]		omittea
Geography (ref. non-OECD)						
Furone	0.085	omitted	omitted	0.007**	omitted	omitted
Lutope	[0.468]	onnica	ommed	[0.002]	ommed	ommed
Spatial unit (ref. national-level analysis)						
Regional – level analysis	-0.057	omitted	omitted	0.081	0.068	0.078**

	[0.112]			[0.084]	[0.050]	[0.0313]
Development status (ref. not specified)						
Developed	omitted	omitted	omitted	0.021 [0.078]	-0.018 [0.052]	0.009 [0.009]
Developing	0.397*** [0.088]	0.405*** [0.089]	0.711*** [0.034]	0.041* [0.022]	0.055** [0.017]	0.065*** [0.008]
Time (ref. midpoint before 2001)						
After 2001	0.081 [0.050]	0.083** [0.022]	0.035 [0.052]	-0.020 [0.017]	omitted	omitted
Other controls						
Dataspan	0.038** [0.011]	0.036** [0.009]	0.009*** [0.007]	0.011*** [0.002]	0.012*** [0.001]	0.013*** [0.001]
Number of RHS variables	-0.0001 [0.0001]	omitted	omitted	-0.003 [0.004]	omitted	omitted
Including important controls and / or fixed effects (ref. missing important controls)	0.0002 [0.0005]	omitted	omitted	-0.007* [0.002]	-0.011* [0.006]	-0.014 [0.010]
Observations	231	231	231	392	392	392
Adjusted R^2	0.716	0.676	0.570	0.944	0.944	0.944
Primary study fixed effects	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes
Country x Midpoint fixed effects	yes	yes	yes	yes	yes	yes

Note: The table replicates the main analysis, after winsorizing the first and the last 20 percentiles from the wage and the employment samples, to remove unreasonable values (outliers). Robust st. errors, clustered by primary study are reported in brackets. All models are weighted by the inverse of the variance of the collected estimates. Columns 2 and 5 are based on the respective WLS-MRA models (Columns 1 and 4), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories is significant for α =0.01. * for p<0.1, ** for p<0.05 and *** for p<0.01.
	Wage elasticities				Employment elasticities			
Variable	WLS	WLS general to specific	PEESE	WLS general to specific - China	WLS	WLS general to specific	PEESE	WLS general to specific - China
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
FAT – PET – PEESE tests								
Intercept (β_0) (PET) or (γ_0) (PEESE)	1.331 [0.138]	1.792* [0.923]	0.885*** [0.114]	-0.121 [0.223]	0.505 [0.720]	-0.105*** [0.019]	-0.122*** [0.017]	0.322 [4.028]
St. error (β_1) (FAT) or (γ_1)	-0.032 [0.728]	0.430 [0.443]	0.054 [0.073]	-0.229 [0.902]	0.283 [0.738]	-0.141 [0.313]	-0.135 [0.597]	0.620 [2.180]
<i>Type of trade (ref. trade in final goods)</i>			_					
Trade in intermediates	-0.002** [0.0009]	-0.003*** [0.0006]	-0.003*** [0.0006]	-0.003** [0.0008]	-0.058 [0.064]	omitted	omitted	omitted
Data aggregation (ref. worker-level)								
Industry-level	-0.002 [0.004]	-0.002 [0.003]	-0.002 [0.003]	-0.049** [0.019]	-0.602 [1.002]	omitted	omitted	omitted
Firm-level	0.021 [0.020]	0.015 [0.019]	0.018 [0.019]	omitted	-0.302 [0.987]	omitted	omitted	omitted
Sectoral Ievel (ref. economy-wide)								
Manufacturing	0.016*** [0.004]	0.014*** [0.002]	0.015*** [0.00004]	0.016** [0.004]	-0.122*** [0.023]	-0.095*** [0.014]	-0.090*** [0.016]	-0.211*** [0.033]
Services	-0.0003 [0.009]	-0.003 [0.010]	-0.0004 [0.009]	-0.011 [0.045]	-0.091** [0.036]	-0.071** [0.020]	-0.065** [0.021]	0.025 [0.021]
Skill-level (ref. varied skill)								
Low-skill	0.005 [0.004]	0.004 [0.004]	0.004 [0.004]	-0.001 [0.0009]	-0.015*** [0.0009]	-0.015** [0.0008]	-0.015*** [0.0006]	0.027 [0.046]
Medium-skill	0.006 [0.004]	0.004 [0.004]	0.005 [0.003]	omitted	-0.048*** [0.011]	-0.043** [0.011]	-0.042** [0.012]	omitted
High-skill	0.014** [0.004]	0.012** [0.004]	0.012** [0.004]	0.003** [0.001]	-0.006*** [0.001]	-0.005** [0.001]	-0.006*** [0.001]	0.307** [0.053]

Table D1.3. Labor market effects from trade (Dependent variable: e_i)

Other methodological aspects								
Technology and trade	0.0005	omitted	omittad	omitted	-0.003	omitted	omitted	omitted
(ref. only trade)	[0.001]	omitieu	omitieu	omneu	[0.002]	omittea	omitieu	omilieu
Time series	-0.007	omitted	omitted	omitted	-0.229**	omitted	omitted	omitted
(ref. cross-sectional)	[0.018]	omineu	omitteu	omnieu	[0.067]	omittea	omneu	ommeu
Lagged independent variable	0.0002	omitted	omitted	omitted	0.002	omitted	omitted	omitted
(ref. contemporaneous ind. var.)	[0.0006]	omineu	omitteu	omnieu	[0.002]	omittea	omneu	omnieu
Dependent variable in levels	-0.022	omitted	omitted	omitted	0.003	omitted	omitted	omitted
(ref. first-differenced dep. var)	[0.039]	omineu	ommed	omnica	[0.004]	omitica	ommed	omnica
Controlling for endogeneity	0.004	omitted	omitted	omitted	-0.0002	omitted	omitted	omitted
(ref. no IV approach)	[0.004]	omineu	ommed	omnica	[0.003]	omitica	omnica	ommed
Wages in shares	-1.997	-2.657**	-1.497***	0.131**				
(ref. wages not in shares)	[1.337]	[1.903]	[0.123]	[0.046]				
Hours worked					-0.024	-0.058***	-0.082***	0.256
(ref. number of workers)					[0.019]	[0.001]	[0.019]	[1.836]
Publication characteristics (K-)								
Peer reviewed paper	-0.743	-1.002**	-0.565***	omitted	0.172***	0.213***	0.227***	-0.344
(ref. working paper)	[0.575]	[0.447]	[0.055]	omnica	[0.030]	[0.027]	[0.027]	[0.485]
Peer reviewed paper * st_error	0.345	omitted	omitted	omitted	-0.928	omitted	omitted	omitted
reer reviewed puper st. error	[0.786]	omineu	ommed	omnica	[0.928]	omitica	ommed	omnica
Publication year	0.160***	0.136***	0.129***	0.574**	0.011	-0.123***	-0.136***	0.157
r uonoution your	[0.037]	[0.008]	[0.003]	[0.202]	[0.041]	[0.023]	[0.020]	[3.278]
Publication year * st. error	-0.478	omitted	omitted	omitted	0.422	0.403	0.847	-5.051
i donoation your st. onor	[0.440]	ommed	ommed	ommed	[0.363]	[0.307]	[1.414]	[3.464]
10-Year impact factor	1.151	1.579*	0.756***	0.449***	-0.401	omitted	omitted	omitted
	[1.051]	[0.835]	[0.102]	[0.029]	[0.546]	omnea	ommed	ommed
10-Year impact factor * st_error	0.345	omitted	omitted	omitted	-1.224	omitted	omitted	omitted
To Tear impact factor st. error	[0.466]	ommed	ommed	ommed	[0.825]	omnea	ommed	ommed
Geography (ref. US-UK-Canada-Japan	2)							
Europe	-0.008**	-0.008**	-0.011***	0.003	0.100	omitted	omitted	omitted
Lurope	[0.002]	[0.003]	[0.0009]	[0.076]	[0.074]	ommod	01111100	5
Spatial unit (ref. national-level analysis	s)							

Regional – level analysis	-0.031 [0.063]	omitted	omitted	omitted	0.062 [0.071]	0.433*** [0.047]	0.470*** [0.034]	0.094 [1.854]
Development status (ref. not-specified)								
Developed	0.005* [0.003]	0.008** [0.004]	0.009* [0.005]	omitted	-0.015 [0.012]	omitted	omitted	omitted
Developing	-0.026 [0.019]	-0.025 [0.019]	-0.025 [0.019]	omitted	-0.019 [0.018]	omitted	omitted	omitted
<i>Time (ref. midpoint before 2001)</i>								
After 2001	-0.524 [0.599]	-0.759 [0.466]	-0.302*** [0.062]	-0.087*** [0.0004]	-0.077*** [0.018]	-0.091*** [0.014]	-0.092*** [0.019]	-0.023 [0.073]
Other controls								
Dataspan	-0.379*** [0.031]	-0.412*** [0.027]	-0.387*** [0.025]	-0.377** [0.054]	0.0533*** [0.009]	0.056*** [0.010]	0.058** [0.020]	0.071** [0.016]
Number of RHS variables	-0.0002 [0.0006]	omitted	omitted	omitted	0.001 [0.0009]	omitted	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important</i> controls or fixed effects)	0.019** [0.006]	0.018*** [0.002]	0.017*** [0.0002]	0.017*** [0.002]	0.002 [0.018]	omitted	omitted	omitted
Observations	390	390	390	62	533	533	533	41
Adjusted R^2	0.325	0.331	0.326	0.644	0.541	0.504	0.501	0.864
Primary study fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Country x Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes

Note: The table replicates the main analysis, after trimming the first and the last 5 percentiles from the distribution of the elasticities and the st. errors, to remove unreasonable values in the estimated effects and their st. errors (outliers). Robust st. errors, clustered by primary study are reported in brackets. All models are weighted by the inverse of the variance of the collected estimates. Columns 2 and 6 are based on the respective weighted-least-squares models (Columns 1 and 5), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories are significant at the 10% level. * for p<0.1, ** for p<0.05 and *** for p<0.01.

		Wage ela	asticities		Employment elasticities			
Variable	WLS	WLS general to specific	PEESE	WLS general to specific - China	WLS	WLS general to specific	PEESE	WLS general to specific - China
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
FAT – PET – PEESE tests								
Intercept (β_0) (PET) or (γ_0) (PEESE)	2.085* [0.834]	1.944** [0.470]	0.027** [0.012]	0.087 [0.090]	-2.029** [0.726]	0.689 [0.710]	0.158 [0.459]	1.177** [0.493]
St. error (β_1) (FAT) or (γ_1)	0.063 [0.250]	0.224 [0.221]	0.00008 [0.0001]	-0.665 [0.366]	-0.057 [0.345]	0.086 [0.210]	-0.047 [0.047]	-1.134** [0.264]
<i>Type of trade (ref. trade in final goods)</i>								
Trade in intermediates	-0.002** [0.0007]	-0.003*** [0.0005]	-0.003*** [0.0005]	-0.003** [0.003]	-0.007 [0.077]	omitted	omitted	omitted
Data aggregation (ref. worker-level)								
Industry-level	-0.003 [0.004]	-0.009 [0.007]	-0.008 [0.006]	-0.597 [0.361]	1.763** [0.477]	omitted	omitted	omitted
Firm-level	0.013* [0.005]	0.016** [0.006]	0.016** [0.006]	omitted	1.867** [0.494]	omitted	omitted	omitted
Sectoral Ievel (ref. economy-wide)								
Manufacturing	0.014*** [0.002]	0.014*** [0.0009]	0.015*** [0.0001]	0.018** [0.002]	-0.103*** [0.012]	-0.101*** [0.012]	-0.081*** [0.011]	0.137** [0.030]
Services	-0.0008 [0.009]	-0.002 [0.009]	-0.0006 [0.009]	-0.032 [0.018]	-0.084*** [0.017]	-0.089*** [0.014]	-0.066*** [0.014]	0.097** [0.031]
Skill-level (ref. varied skill)								
Low-skill	0.005 [0.003]	0.004 [0.004]	0.005 [0.004]	-0.0002 [0.0003]	-0.002*** [0.0004]	-0.001*** [0.0002]	-0.0007 [0.0005]	-0.041 [0.023]
Medium-skill	0.007** [0.003]	0.005 [0.003]	0.005* [0.003]	omitted	-0.004 [0.008]	-0.004 [0.008]	-0.0007 [0.007]	omitted
High-skill	0.014** [0.004]	0.012** [0.003]	0.013** [0.004]	0.003*** [0.00004]	0.0008 [0.0009]	0.002*** [0.0001]	0.001*** [0.0001]	0.059 [0.038]

Table D1.4. Labor market effects from trade (Dependent variable: e_i)

Other methodological aspects								
Technology and trade	0.0004	omitted	omitted	omitted	-0.001	omitted	omitted	omitted
(ref. only trade)	[0.001]	omineu	ommed	ommed	[0.002]	omitteu	omineu	omittea
Time series	0.009	omitted	omitted	omitted	-0.093	omitted	omitted	omitted
(ref. cross-sectional)	[0.006]	omned	ommed	omned	[0.077]	omnied	omitted	omnea
Lagged independent variable	0.0008	omitted	omitted	omitted	-0.0005	omitted	omitted	omitted
(ref. contemporaneous ind. var.)	[0.001]	ommed	ommed	ommed	[0.0006]	ommed	ommed	omnea
Dependent variable in levels	0.011	omitted	omitted	omitted	0.004	omitted	omitted	omitted
(ref. first-differenced dep. var)	[0.032]	0			[0.003]	0	0	0
Controlling for endogeneity	0.005	omitted	omitted	omitted	0.008	omitted	omitted	omitted
(ref. no IV approach)	[0.0054]			0.010	[0.006]			
Wages in shares	-2.389**	-2.261**	-0.046***	-0.019				
(ref. wages not in shares)	[0.971]	[0.611]	[0.011]	[0.019]		0.0.40	0.000	0.000
Hours worked					0.062**	0.060	0.009	0.009
(ref. number of workers)					[0.022]	[0.037]	[0.009]	[0.008]
Publication characteristics (K-)	1.050.44	0.0.01.144	0.04046464	0 61 54	0.000	0.104	0.000	0.107
Peer reviewed paper	-1.050**	-0.961**	-0.048***	0.615*	0.009	-0.134	0.029	-0.107
(ref. working paper)	[0.410]	[0.229]	[0.011]	[0.368]	[0.025]	[0.146]	[0.109]	[0.125]
Peer reviewed paper * st. error	0.218	omitted	omitted	omitted	-0.034	omitted	omitted	omitted
1 1	[0.355]	0.000	0.010	0.102	[0.423]	1.000	0.000	
Publication year	-0.014	-0.003	0.010**	-0.103	0.069**	1.009	0.289	-1.760**
ý	[0.026]	[0.005]	[0.005]	[0.082]	[0.028]	[1.013]	[0.650]	[0.618]
Publication year * st. error	0.00005	omitted	omitted	omitted	0.511**	0.596**	omitted	1.446**
·	[0.381]	1 720***	0.002	0.010	[0.121]	[0.239]		[0.364]
10-Year impact factor	1.869**	1./39***	0.002	-0.010	0.253	omitted	omitted	omitted
1	[0.764]	[0.423]	[0.006]	[0.012]	[0.403]			
10-Year impact factor * st. error	0.294	omitted	omitted	omitted	-0.430	omitted	omitted	omitted
	[0.355]				[0.378]			
Geography (ref. US-UK-Canada-Japan	<i>)</i>	0.000***	0.011***	0.051	0.546**			
Europe	-0.009***	-0.009*** [0.001]		-0.031	0.540**	omitted	omitted	omitted
Spatial unit (not national low of an alust	[0.002]	[0.001]	[0.0008]	[0.031]	[0.207]			
spanai unii (rej. naiionai-ievei analysis	S)							

Regional – level analysis	-0.004 [0.050]	omitted	omitted	omitted	0.363*** [0.143]	0.483** [0.173]	0.253*** [0.012]	-0.067* [0.047]
Development status (ref. not-specified)								
Developed	0.004 [0.003]	0.008** [0.004]	0.008* [0.004]	omitted	-0.010 [0.007]	omitted	omitted	omitted
Developing	-0.010** [0.004]	-0.008** [0.004]	-0.007 [0.004]	omitted	-0.012 [0.010]	omitted	omitted	omitted
<i>Time (ref. midpoint before 2001)</i>								
After 2001	-1.029** [0.442]	-0.959** [0.230]	-0.016 [0.011]	-0.015*** [0.0001]	0.001 [0.011]	-0.007 [0.014]	0.013*** [0.0002]	0.026*** [0.001]
Other controls								
Dataspan	-0.027 [0.025]	-0.030** [0.014]	-0.016*** [0.002]	0.019 [0.022]	0.009* [0.004]	0.011 [0.007]	0.0004* [0.0003]	0.003 [0.013]
Number of RHS variables	0.0003 [0.0003]	omitted	omitted	omitted	-0.0007 [0.001]	omitted	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important</i> <i>controls or fixed effects</i>)	0.014*** [0.003]	0.017*** [0.008]	0.016*** [0.0006]	0.015*** [0.0007]	0.0007 [0.021]	omitted	omitted	omitted
Observations	459	459	459	64	635	635	635	60
Adjusted R^2	0.925	0.926	0.926	0.567	0.997	0.996	0.996	0.912
Primary study fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Country x Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes

Note: The table replicates the main analysis, after winsorizing the first and the last 20 percentiles from the wage and the employment samples, to remove unreasonable values (outliers). Robust st. errors, clustered by primary study are reported in brackets. All models are weighted by the inverse of the variance of the collected estimates. Columns 2 and 5 are based on the respective WLS-MRA models (Columns 1 and 4), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories is significant for α =0.01. * for p<0.1, ** for p<0.05 and *** for p<0.01.

Appendix D2. Estimation outcomes excluding semi-elasticities

	Table D2.1.	Labor market	effects from	technology	(Dependent	variable: e_i)
--	-------------	--------------	--------------	------------	------------	-------------------

		Wage elasticities	5	Employment elasticities			
Variable	WLS WLS PEESE			WLS	WLS	PEESE	
		general to			general to		
		specific			specific		
	(1)	(2)	(3)	(4)	(5)	(6)	
FAT – PET – PEESE tests							
Intercept (β_2) (PET) or (γ_2) (PEESE)	-0.203	0.418	0.110	0.033	0.367	0.484^{**}	
intercept (p_0) (1 L1) of (p_0) (1 LLSL)	[0.339]	[0.348]	[0.190]	[0.328]	[0.426]	[0.166]	
St error (β_{\star}) or st error ² (γ_{\star}) (FAT)	-1.049	-0.963**	2.850	-0.583	-0.140	-0.498	
St. end (p_1) of st. end (y_1) (111)	[1.602]	[1.141]	[1.895]	[2.099]	[1.611]	[0.508]	
Type of innovation (ref. process innovation	n)						
Product innovation	0.006	omitted	omitted	0.003	0.001**	0.001**	
i foddet filliovation	[0.003]	omnica	omnica	[0.006]	[0.0005]	[0.0004]	
Data aggregation (ref. worker-level)							
Industry-level	0.011	omitted	omitted	-0.083	0.142*	0.062	
industry iever	[0.405]	omned	ommea	[0.062]	[0.080]	[0.158]	
Firm-level	0.065	omitted	omitted	-0.052	0.413*	0.520**	
	[0.088]	ommed	ommed	[0.070]	[0.224]	[0.249]	
Sectoral Ievel (ref. economy-wide)							
Manufacturing	-0.009	-0.402	-0.048	-0.001	0.005	0.004	
Thunst actual mig	[0.140]	[0.348]	[0.121]	[0.057]	[0.011]	[0.008]	
Services	-0.002	-0.395	-0.042	-0.010	0.027	0.026***	
	[0.141]	[0.348]	[0.120]	[0.052]	[0.030]	[0.006]	
Skill-level (ref. varied skill)							
Low-skill	-0.001	omitted	omitted	0.082	-0.122	-0.118	
	[0.004]	ommed	ommed	[0.058]	[0.332]	[0.275]	
Medium-skill	-0.0001	omitted	omitted	0.082	-0.111	-0.106	
incutum skin	[0.004]	ommed	Smillea	[0.067]	[0.325]	[0.275]	
High-skill	0.001	omitted	omitted	0.082	0.013	0.016	

	[0.003]			[0.100]	[0.339]	[0.277]
Other methodological aspects						
Technology and trade	0.138**	-0.001	-0.006	0.206*	anitta d	a mitta d
(ref. only technology)	[0.057]	[0.087]	[0.063]	[0.108]	omiliea	omiliea
Time series	-0.014**	-0.012**	-0.006***	-0.011	-0.159***	-0.160*
(ref. cross-sectional)	[0.004]	[0.005]	[0.001]	[0.024]	[0.014]	[0.096]
Lagged independent variable	0.0009	omittad	omittad	-0.011	omitted	omittad
(ref. contemporaneous ind. var.)	[0.001]	omitieu	omitieu	[0.025]	omitiea	omilieu
Dependent variable in levels	-0.026	-0.070*	-0.129**	0.021	0.160***	0.161*
(ref. first-differenced dep. var)	[0.242]	[0.039]	[0.048]	[0.026]	[0.011]	[0.096]
Controlling for endogeneity	0.008	-0.078	-0.007	0.019	omitted	omittad
(ref. no IV approach)	[0.092]	[0.067]	[0.029]	[0.018]	omilieu	omilieu
Wages in shares	0.120	omittad	omitted			
(ref. wages not in shares)	[0.404]	omitieu	ommeu			
Hours worked				0.072**	0.0001***	0.0001
(ref. number of workers)				[0.035]	[6.8*10 ⁻⁸]	[0.00006]
Publication characteristics (K-)						
Peer reviewed paper	0.119	omitted	omitted	0.095	-0.029	-0.196
(ref. working paper)	[0.394]	ommed	ommed	[0.076]	[0.297]	[0.276]
Peer reviewed paper * st error	-0.194	omitted	omitted	-0.208	omitted	omitted
r eer reviewed paper – st. error	[1.202]	omnica	ommed	[1.708]	omnica	omnica
Publication year	-0.083	-0.018	-0.036	-0.006	0.208	0.245
r ubheation year	[0.185]	[0.038]	[0.027]	[0.053]	[0.414]	[0.368]
Publication year * st error	-1.942***	-1.997***	4.131	1.328	0.052	0.153
i doneditori yedi - st. error	[0.219]	[0.442]	[3.028]	[0.991]	[0.871]	[0.966]
10-Year impact factor	-0.002	-0.160	-0.177**	0.059	omitted	omitted
10 Tear impact factor	[0.615]	[0.112]	[0.078]	[0.400]	omnica	omned
10-Year impact factor * st_error	-2.077	-1.944	-14.543	-2.532	omitted	omitted
	[2.039]	[2.044]	[10.331]	[2.751]	ommou	Similar
Geography (ref. US-UK-Canada-Japan)						
Europe	0.333**	omitted	omitted	-0.032	omitted	omitted
Latope	[0.104]	onnica	onninea	[0.028]	onnica	onninea

Spatial unit (ref. national-level analysis)						
Regional – level analysis	omitted	omitted	omitted	-0.073 [0.070]	-0.392 [0.360]	-0.379** [0.151]
Development status (ref. not-specified)						
Developed	omitted	omitted	omitted	omitted	omitted	omitted
Developing	omitted	omitted	omitted	-0.142 [0.215]	2.740* [1.491]	3.194*** [0.819]
Time (ref. midpoint before 2001)						
After 2001	-0.342*** [0.096]	0.211 [0.311]	2.842 [1.770]	0.073 [0.074]	omitted	omitted
Other controls						
Datasnan	-0.002	-0.002	-0.010***	0.029**	-0.004	0.023
Dataspan	[0.002]	[0.002]	[0.0001]	[0.012]	[0.010]	[0.024]
Number of RHS variables	0.00002 [0.00006]	omitted	omitted	0.011** [0.004]	omitted	omitted
Including important controls or fixed	ouritt od	a mittad	a mittad	-0.105	-0.214	-0.209***
fixed effects)	omiliea	omitiea	omitiea	[0.122]	[0.255]	[0.022]
Observations	163	163	163	285	285	285
Adjusted R^2	0.481	0.409	0.318	0.992	0.994	0.995
Primary study fixed effects	yes	yes	yes	no	yes	yes
Country fixed effects	yes	yes	yes	no	no	yes
Midpoint fixed effects	yes	yes	yes	no	no	yes
Country by Midpoint fixed effects	yes	yes	yes	no	no	yes

Note: The table replicates the main analysis, after excluding the semi-elasticities (log-linear models). Robust st. errors, clustered by primary study are reported in brackets, with the exception of Column 6, where they are robust, but not clustered (to achieve convergence). All models are weighted by the inverse of the variance of the collected estimates. Columns 2 and 6 are based on the respective weighted-least-squares models (Columns 1 and 5), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories are significant at the 10% level. * for p<0.1, ** for p<0.05 and *** for p<0.01.

		Wage elastici	ties	Emp	Employment elasticities			
Variable	WLS WLS PEESE			WLS	WLS	PEESE		
		general to			general to			
		specific			specific			
	(1)	(2)	(3)	(4)	(5)	(6)		
FAT – PET – PEESE tests								
Intercept (β_0) (PET) or (γ_0) (PEESE)	-9.950**	3.825***	3.288***	-1.354**	0.068	-2.686*		
	[2.594]	[0.411]	[0.082]	[0.611]	[1.579]	[1.407]		
St. error (β_1) (FAT) or (γ_1)	0.122	0.635	0.002***	1.511*	0.332	0.083		
	[0.474]	[0.477]	[0.0004]	[0.792]	[0.299]	[0.142]		
Type of trade (ref. trade in final goods)		0.000	0.004/55					
Trade in intermediates	-0.002*	-0.003**	-0.004***	-0.217***	omitted	omitted		
	[0.001]	[0.0008]	[0.0003]	[0.048]				
Data aggregation (ref. worker-level)	0.004	0.010	0.000	0.710*				
Industry-level	-0.004	-0.010	0.009	0./12*	omitted	omitted		
, s	[0.004]	[0.007]	[0.006]	[0.379]				
Firm-level	0.011*	0.015**	0.016**	0.848**	omitted	omitted		
	[0.006]	[0.007]	[0.006]	[0.348]				
Sectoral Ievel (ref. economy-wide)	1 207***	2 2 2 0 * * *	0.050***	0 104***	0 1/1***	0.007***		
Manufacturing	1.297	5.569	2.838	-0.184	-0.141	-0.08/****		
-	[0.131]	[0.397]	[0.034]	[0.041]	[0.024]	[0.013]		
Services	1.208^{+++}	5.502****	2.834	-0.13/***	-0.121^{++++}	-0.039***		
Skill lovel (ref varied skill)	[0.147]	[0.394]	[0.055]	[0.043]	[0.022]	[0.019]		
Skill-level (Tej. varlea-skill)	0.011**	0.007	0.008	0.004*	0.004**	0.002		
Low-skill	10.0011	0.007 [0.005]	0.008	-0.004°	-0.004	-0.002		
	0.013**	0.005	0.005	-0.047**	_0.0/11**	[0.002]		
Medium-skill	0.013	10.0041	0.008 [0.00/1]	-0.047	-0.041	-0.027		
	0.020**	0.015**	0.016**	0.0008	0.002**	0.001		
High-skill	[0 005]	[0 005]	[0 005]	[0 0007]	[0 0005]	[0 0008]		
Other methodological aspects	[0.000]	[0.000]	[0.000]	[0.0007]	[0.0002]	[0.0000]		

Table D2.2. Labor market effects from trade (Dependent variable: e_i)

Technology and trade	-0.001	omitted	omitted	-0.001	omitted	omitted
(ref. only trade)	[0.002]	omned	omnica	[0.003]	ommed	omned
Time series	-0.010	omitted	omitted	-0.097	omitted	omitted
(ref. cross – sectional)	[0.008]	omineu	omittea	[0.063]	omitteu	omilieu
Lagged independent variable	0.0009	omittad	omittad	-0.001	omittad	omitted
(ref. contemporaneous)	[0.002]	omitieu	omitiea	[0.001]	omitieu	omitieu
Dependent variable in levels	0.043	omittad	omittad	0.038***	amittad	amittad
(ref. first-differenced dep. var.)	[0.048]	omitiea	omittea	[0.005]	omiliea	omitiea
Controlling for endogeneity	0.017**	amitta d	a mitta d	0.008	amitta d	amittad
(ref. no-IV approach)	[0.005]	omiliea	omittea	[0.006]	omitiea	omiliea
Wages in shares	-0.716***	-3.040***	-2.382***			
(ref. wages not in shares)	[0.133]	[0.499]	[0.076]			
Hours worked				-0.089**	-0.581	4.185
(ref. number of workers)				[0.043]	[2.780]	[2.474]
Publication characteristics						
Peer reviewed paper	0.637**	-3.293***	2.774***	0.132**	0.531	-3.640***
(ref. working paper)	[0.168]	[0.391]	[0.061]	[0.056]	[2.427]	[2.164]
Peer reviewed paper * st error	0.985	omitted	omitted	-1.981**	omitted	omitted
i cer ieviewed paper st. enor	[0.837]	omineu	omnica	[0.887]	omnica	omitted
Publication year	-0.281***	1.847***	1.557***	-0.083**	0.169	-3.638*
I ublication year	[0.052]	[0.219]	[0.035]	[0.024]	[2.234]	[1.997]
Publication year * st arror	-0.292	omittad	omittad	1.503**	1.470***	-0.105
Fublication year * st. enor	[0.631]	omitieu	omitiea	[0.587]	[0.344]	[0.251]
10 Voor import factor	-0.715***	1.108^{***}	0.932***	-1.423*	omittad	omittad
10- Tear Impact factor	[0.152]	[0.135]	[0.031]	[0.772]	omitieu	omilieu
10 Voor impost factor * st orror	-0.268	omittad	omittad	-1.204	omittad	omittad
10-Tear impact factor + st. enor	[0.425]	omitieu	omitiea	[0.764]	omitieu	omilieu
Geography (ref. non-OECD)						
Furone	1.347***	-0.047**	-0.044**	0.305**	omittad	omitted
Europe	[0.231]	[0.021]	[0.021]	[0.126]	omitieu	omitieu
Spatial unit (ref. national-level analysis	s)					
Regional – level analysis	1.565***	omitted	omitted	0.985**	0.754	2.581**

	[0.200]			[0.072]	[1.042]	[0.931]
Development status (ref. not specified)						
Developed	-0.003 [0.006]	0.012*** [0.003]	0.015*** [0.001]	-0.024* [0.012]	omitted	omitted
Developing	-0.038 [0.025]	-0.022 [0.024]	-0.020 [0.023]	-0.027 [0.017]	omitted	omitted
Time (ref. midpoint before 2001)						
After 2001	7.781** [2.251]	-5.690*** [0.662]	-4.814*** [0.117]	-0.002 [0.045]	-0.552 [3.473]	5.402* [3.092]
Other controls						
Dataspan	-0.573*** [0.111]	-0.004 [0.006]	-0.002 [0.006]	-0.514*** [0.025]	-0.508*** [0.015]	-0.561*** [0.0009]
Number of RHS variables	-0.00002 [0.0001]	omitted	omitted	0.0003 [0.001]	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important</i> controls or fixed effects)	0.172 [0.138]	0.079* [0.041]	0.024*** [0.002]	-0.371 [0.228]	omitted	omitted
Observations	306	306	306	471	471	471
Adjusted R^2	0.904	0.905	0.905	0.993	0.992	0.993
Primary study fixed effects	no	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	no	no
Midpoint fixed effects	yes	yes	yes	yes	yes	yes
Country x Midpoint fixed effects	yes	yes	yes	yes	no	no

Note: The table replicates the main analysis, after excluding the semi-elasticities (log-linear models). Robust st. errors, clustered by primary study are reported in brackets. All models are weighted by the inverse of the variance of the collected estimates. Columns 2 and 5 are based on the respective weighted-least-squares models (Columns 1 and 4), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories are significant at the 10% level. In Columns 5 and 6 some categories of fixed effects are omitted to achieve convergence. The estimations for trade with China are not reported due to insufficient observations * for p<0.1, ** for p<0.05 and *** for p<0.01.

Table D3.1. Labor market effects from technology (Dependent variable: e_i)	
--	--

	Wage elasticities			Employment elasticities			
Variable	WLS	WLS	PEESE	WLS	WLS	PEESE	
		general to			general to		
		specific			specific		
	(1)	(2)	(3)	(4)	(5)	(6)	
FAT – PET – PEESE tests							
Intercept (β) (DET) or (γ) (DEESE)	0.605	0.206*	0.637***	-0.756***	0.221	0.453**	
intercept (p_0) (1 \pm 1) or (p_0) (1 \pm \pm \pm \pm	[0.740]	[0.115]	[0.067]	[0.152]	[0.239]	[0.157]	
St error (β) or st error ² (γ) (EAT)	-3.373***	-1.653**	-2.884	-1.618	-1.344	0.805	
St. end (p_1) of st. end (γ_1) (PA1)	[0.632]	[0.550]	[2.301]	[1.692]	[1.004]	[0.615]	
Type of innovation (ref. process innovation	n)						
Product innovation	0.004	omitted	omitted	0.001**	0.001**	0.001**	
I foudet innovation	[0.004]	omitieu	omilieu	[0.0003]	[0.0005]	[0.0004]	
Data aggregation (ref. worker-level)							
Industry level	0.213**	omitted	omitted	-0.155	0.067	0.146**	
industry-lever	[0.079]	ommed	ommeu	[0.141]	[0.126]	[0.067]	
Firm lavel	0.539	omitted	omitted	-0.360	0.601*	0.918***	
Тттт-теует	[1.027]	omitieu	omilieu	[0.298]	[0.345]	[0.117]	
Sectoral Ievel (ref. economy-wide)							
Manufacturing	-0.287**	-0.212*	-0.752***	-0.227	0.221	0.200	
Wandracturing	[0.081]	[0.106]	[0.117]	[0.152]	[0.140]	[0.130]	
Services	1.031	0.134	-0.675**	0.078***	0.068***	0.038***	
50111605	[0.916]	[0.423]	[0.281]	[0.017]	[0.013]	[0.005]	
Skill-level (ref. varied skill)							
I ow_ekill	-0.005	omitted	omitted	0.056	0.038	-0.020	
Low-skiii	[0.004]	ommed	ommed	[0.054]	[0.038]	[0.015]	
Medium_skill	-0.035	omitted	omitted	-0.114*	-0.093	-0.194***	
wiedrum-skin	[0.045]	ommeu	ommeu	[0.065]	[0.071]	[0.017]	
High-skill	0.003	omitted	omitted	0.197***	0.189***	0.166***	

	[0.003]			[0.017]	[0.011]	[0.002]
Other methodological aspects						
Technology and trade (<i>ref. only technology</i>)	-0.007** [0.003]	-0.007** [0.002]	-0.016*** [0.003]	0.009 [0.025]	omitted	omitted
Time series (ref. cross-sectional) Lagged independent variable (ref. contemporaneous ind. var.)	-0.006 [0.006] 0.001 [0.001]	-0.006 [0.006] omitted	-0.009*** [0.0004] omitted	-0.142*** [0.017] -0.006 [0.016]	-0.146*** [0.009] omitted	-0.160*** [0.00006] omitted
Dependent variable in levels (ref. first-differenced dep. var) Controlling for endogeneity (ref. no IV approach) Wages in shares	0.821 [0.942] -0.012** [0.004] 0.055 [0.308]	-0.015 [0.383] -0.016** [0.004] omitted	-0.764** [0.279] -0.041*** [0.001] omitted	0.147*** [0.016] 0.029 [0.019]	0.151*** [0.006] omitted	0.160* [0.000006] omitted
(ref. number of workers)	[0.300]			0.0001*** [1.1*10 ⁻⁷]	0.0001*** [6.5*10 ⁻⁸]	0.0001*** [9.5*10 ⁻⁸]
Publication characteristics (K-)						
Peer reviewed paper (ref. working paper)	-0.967** [0.356]	omitted	omitted	0.731*** [0.083]	-0.241 [0.315]	-0.473** [0.170]
Peer reviewed paper * st. error	1.500* [0.761]	omitted	omitted	0.668 [1.670]	omitted	omitted
Publication year	-0.150** [0.038]	-0.283 [0.405]	-0.959** [0.299]	-0.398** [0.155]	0.290 [0.232]	0.473*** [0.070]
Publication year * st. error	-1.802*** [0.384]	-1.958*** [0.435]	-1.036* [0.581]	-1.331** [0.583]	-1.258** [0.569]	-2.579 [2.745]
10-Year impact factor	0.548*** [0.109]	0.770* [0.396]	0.731** [0.197]	-0.402** [0.172]	omitted	omitted
10-Year impact factor * st. error	-5.592** [1.431]	-4.997** [1.383]	-4.776 [4.845]	1.008 [1.444]	omitted	omitted
Geography (ref. US-UK-Canada-Japan)						
Europe	-0.635 [0.437]	omitted	omitted	-0.0008 [0.005]	omitted	omitted

Spatial unit (ref. national-level analysis)						
Regional – level analysis	0.102 [0.107]	omitted	omitted	0.504** [0.231]	0.472** [0.178]	0.773*** [0.157]
Development status (ref. not-specified)						
Developed	omitted	omitted	omitted	0.531 [0.482]	0.897*** [0.213]	0.542*** [0.047]
Developing	0.540*** [0.067]	0.562*** [0.061]	0.993*** [0.059]	0.199** [0.080]	0.211** [0.074]	0.325*** [0.049]
Time (ref. midpoint before 2001)						
After 2001	-0.311** [0.086]	-0.284*** [0.053]	-0.326** [0.079]	0.022 [0.080]	omitted	omitted
Other controls						
Dataspan	-0.348*** [0.010]	-0.348*** [0.008]	-0.382*** [0.011]	0.241*** [0.005]	0.241*** [0.005]	0.252*** [0.0006]
Number of RHS variables	0.00002 [0.00008]	omitted	omitted	0.001 [0.005]	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important controls or</i> <i>fixed effects</i>)	-0.046 [0.060]	omitted	omitted	-0.096** [0.046]	-0.098** [0.038]	-0.118 [0.093]
Observations	153	153	153	355	355	355
Adjusted R^2	0.731	0.723	0.547	0.985	0.985	0.984
Primary study fixed effects	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes
Country by Midpoint fixed effects	yes	yes	yes	yes	yes	yes

Note: The table replicates the main analysis, after excluding calculated elasticities (semi-elasticities), based on log-log (log-linear) models. Robust st. errors, clustered by primary study are reported in brackets. All models are weighted by the inverse of the variance of the collected estimates. Columns 2 and 6 are based on the respective weighted-least-squares models (Columns 1 and 5), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories are significant at the 10% level. * for p<0.1, ** for p<0.05 and *** for p<0.01.

		Wage e	elasticities		Employment elasticities				
Variable	WLS	WLS	PEESE	WLS	WLS	WLS	PEESE	WLS	
		general to		general to		general to		general to	
		specific		specific -		specific		specific -	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
FAT – PET – PEESE tests	(-)	(_)	(0)	(-)	(0)	(0)	(.)	(0)	
Lateracent (Q) (DET) an (a) (DEESE)	1.597**	1.893***	1.535***	0.009	-12.777***	-0.512***	-0.480***	-2.312	
Intercept (β_0) (PE1) or (γ_0) (PEESE)	[0.450]	[0.275]	[0.153]	[0.243]	[1.145]	[0.057]	[0.065]	[6.614]	
St error (B) (EAT) or (y)	0.413	0.498	-0.700***	-0.752	0.866	0.010	-0.301	-5.068	
St. end (p_1) (FAT) of (r_1)	[0.612]	[0.399]	[0.689]	[0.982]	[0.825]	[0.297]	[0.205]	[3.249]	
Type of trade (ref. trade in final goods)									
Trade in intermediates	-0.002**	-0.003***	-0.003***	-0.003**	-0.081	omitted	omitted	omitted	
Determine the (not such as low 1)	[0.001]	[0.0006]	[0.0006]	[0.0008]	[0.078]				
Data aggregation (ref. worker-level)	0.004	0.000	0.009	1.017*	10 041***				
Industry-level	-0.004 [0.004]	-0.009	-0.008	1.91/*	10.941****	omitted	omitted	omitted	
	0.013*	0.016**	0.016**	[0.907]	11 382***				
Firm-level	[0.007]	[0.007]	[0.006]	omitted	[1.067]	omitted	omitted	omitted	
Sectoral Ievel (ref. economy-wide)	[01007]	[0.007]	[0.000]		[1007]				
Monufosturina	0.015***	0.013***	0.015***	0.019**	-0.148***	-0.123***	-0.089***	1.379*	
Manufacturing	[0.003]	[0.002]	[0.00005]	[0.004]	[0.034]	[0.019]	[0.013]	[0.612]	
Services	-0.001	-0.003	-0.0004	-0.037	-0.134**	-0.101***	-0.062**	1.616**	
bervices	[0.010]	[0.010]	[0.009]	[0.049]	[0.039]	[0.018]	[0.018]	[0.634]	
Skill-level (ref. varied-skill)									
Low-skill	0.006	0.004	0.005	-0.0001	-0.003	-0.003**	-0.001	-0.085	
	[0.004]	[0.004]	[0.004]	[0.001]	[0.002]	[0.001]	[0.002]	[0.048]	
Medium-skill	0.00/* [0.004]	0.004	U.UU6*	omitted	-0.04/***	-U.U44*** [0.007]	-0.035***	omitted	
	[U.UU4] 0.015**	[0.003] 0.013**	[0.005] 0.013**	0.003**	[0.012] 0.0006	[U.UU7] 0.002**	[U.UU8] 0.001*	0.175	
High-skill	[0.005]	[0.004]	[0.004]	[0.001]	[0.0006]	[0.0006]	[0.0007]	[0.100]	

Table D3.2. Labor market effects from trade (Dependent variable: e_i)

Other methodological aspects								
Technology and trade	-0.0001	omitted	omitted	omitted	-0.001	omitted	omitted	omitted
(ref. only trade)	[0.001]	ommed	omned	omnieu	[0.002]	ominea	omitteu	omittea
Time series	0.007	omitted	omitted	omitted	-0.417**	omitted	omitted	omitted
(ref. cross – sectional)	[0.008]	ommed	omineu	ommed	[0.163]	omitica	omnica	omnica
Lagged independent variable	0.0008	omitted	omitted	omitted	-0.0005	omitted	omitted	omitted
(ref. contemporaneous)	[0.001]	onnica	ommou	onniced	[0.0006]	ommed	ommed	ommed
Dependent variable in levels	0.006	omitted	omitted	omitted	0.004	omitted	omitted	omitted
(ref. first-differenced dep. var.)	[0.051]				[0.003]			
Controlling for endogeneity	0.007	omitted	omitted	omitted	0.007	omitted	omitted	omitted
(ref. no-IV approach)	[0.005]			0.105*	[0.006]			
Wages in shares	0.572***	0.670***	0.529***	0.105*				
(ref. wages not in shares)	[0.132]	[0.108]	[0.058]	[0.051]	0 072***	0 000***	0 0 0 0 + + +	0 1 47
Hours worked					$0.2/3^{***}$	0.282^{***}	0.239***	0.147
(ref. number of workers)					[0.044]	[0.039]	[0.044]	[0.160]
Publication characteristics	0.000***	1.040***	0.05(***	1.055***	0 100***	0.070**	0.050	1 296
Peer reviewed paper	-0.909^{++++}	-1.040****		-1.955***	-0.189****	-0.079^{**}	-0.050	1.380
(ref. working paper)		[0.142]	[0.078]	[0.988]	[0.045]	[0.023]	[0.030]	[1.901]
Peer reviewed paper * st. error	0.008	omitted	omitted	omitted	-1.490	omitted	omitted	omitted
	[0.043]	0 122***	0 111***	0 457*	[1.037]	0 746***	0 6/1***	1 202
Publication year	[0.129**	[0.003]	[0,006]	0.437	0.100^{11}	0.740°°°	-0.041	-1.295
	0.151	[0.003]	[0.000]	[0.220]	[0.042]	[0.088] 1 1/8**	[0.093]	[7.730] 5.604
Publication year * st. error	-0.131 [0.61/1]	omitted	omitted	omitted	[0.470]	[0 325]	0.333	[4 476]
	1 /28**	1 689***	1 370***	0.466***	1 730***	[0.323]	[0.362]	[4.470]
10-Year impact factor	1. 4 20	[0.242]	[0 136]	[0 032]	[0 237]	omitted	omitted	omitted
	0 304	[0.242]	[0.150]	[0.032]	-0.809			
10-Year impact factor * st. error	[0 471]	omitted	omitted	omitted	[0 917]	omitted	omitted	omitted
Geography (ref. non-OECD)	[0.171]				[0.717]			
	-0.008**	-0.007**	-0.011***	-0.041	3.575***	• •	• 7	• •
Europe	[0.003]	[0.002]	[0.0009]	[0.083]	[0.470]	omitted	omitted	omitted
Spatial unit (ref. national-level analysis	s)							

Regional – level analysis	0.004 [0.070]	omitted	omitted	omitted	2.814*** [0.325]	3.129*** [0.188]	2.773*** [0.025]	1.473* [0.709]
Development status (ref. not specified)								
Developed	0.004 [0.003]	0.008** [0.004]	0.009** [0.005]	omitted	-0.022* [0.012]	omitted	omitted	omitted
Developing	-0.025 [0.019]	-0.023 [0.018]	-0.022 [0.018]	omitted	-0.027 [0.018]	omitted	omitted	omitted
<i>Time (ref. midpoint before 2001)</i>								
After 2001	-0.680** [2.227]	-0.811*** [0.138]	-0.630*** [0.076]	-0.088*** [0.0005]	-0.109*** [0.024]	-0.131*** [0.014]	-0.099*** [0.001]	0.262** [0.070]
Other controls								
Dataspan	-0.398*** [0.037]	-0.416*** [0.024]	-0.373*** [0.013]	-0.345** [0.059]	0.052*** [0.010]	-0.060*** [0.008]	0.042*** [0.001]	0.042 [0.028]
Number of RHS variables	-0.0004 [0.0004]	omitted	omitted	omitted	0.0005 [0.001]	omitted	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important</i> controls or fixed effects)	0.015 [0.004]	0.018*** [0.002]	0.016*** [0.002]	0.015*** [0.002]	0.0006 [0.024]	omitted	omitted	omitted
Observations	384	384	384	64	590	590	590	60
Adjusted R^2	0.910	0.912	0.912	0.621	0.993	0.992	0.992	0.767
Primary study fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Country x Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes

Note: The table replicates the main analysis, after excluding calculated elasticities (semi-elasticities), based on log-log (log-linear) models. Robust st. errors, clustered by primary study are reported in brackets. All models are weighted by the inverse of the variance of the collected estimates. Columns 2 and 5 are based on the respective weighted-least-squares models (Columns 1 and 4), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories are significant at the 10% level. * for p<0.1, ** for p<0.05 and *** for p<0.01.

Appendix D4. Estimation outcomes excluding elasticities where the dependent variable is expressed in shares

	Wage elasticities			Employment elasticities			
Variable	WLS	WLS	PEESE	WLS	WLS	PEESE	
		general to			general to		
		specific			specific		
	(1)	(2)	(3)	(4)	(5)	(6)	
FAT – PET – PEESE tests							
Intercent (\mathcal{B}_{1}) (PET) or (\mathcal{U}_{1}) (PEESE)	1.175*	3.909	9.666	-0.236	-0.503	-0.623***	
Intercept (p_0) (PE1) or (γ_0) (PEESE)	[0.602]	[2.661]	[10.820]	[0.600]	[0.525]	[0.147]	
St error (β) or st error ² (γ) (EAT)	-4.059**	-2.181***	-2.623	-1.209	0.080	0.698	
St. chor (p_1) of st. chor (p_1) (PAT)	[1.222]	[0.309]	[2.778]	[1.131]	[0.956]	[0.224]	
Type of innovation (ref. process innovation	n)						
Product innovation	-0.0007	omitted	omitted	0.001**	0.001**	0.001**	
Tioduct milovation	[0.004]	0.004]	ommed	[0.0002]	[0.0004]	[0.0005]	
Data aggregation (ref. worker-level)							
Industry-level	2.294	omitted	omitted	13.953	-1.024**	-1.164***	
industry lever	[1.462]	omineu	ommed	[10.502]	[0.295]	[0.114]	
Firm-level	1.526	omitted	omitted	0.043	-0.125*	-0.125*	
1 1111-10/01	[1.695]	ominea	ommed	[0.067]	[0.072]	[0.061]	
Sectoral Ievel (ref. economy-wide)							
Manufacturing	-0.295**	-0.185*	-0.769***	0.232	0.207	0.202	
Wandracturing	[0.078]	[0.104]	[0.119]	[0.162]	[0.142]	[0.129]	
Services	4.404	-2.765	-7.333	0.073***	0.045**	0.038***	
501 11005	[3.161]	[2.403]	[8.060]	[0.018]	[0.016]	[0.005]	
Skill-level (ref. varied skill)							
I ow-skill	-0.005*	omitted	omitted	0.058	0.004	-0.010***	
Low-skiii	[0.004]	Unitied	Unitied	[0.041]	[0.037]	[0.0004]	
Medium_skill	0.107	omitted	omitted	omitted	omitted	omitted	
wiedrum-skin	[0.202]	ommeu	ommeu	ommeu	ommeu	ommen	
High-skill	-0.005*	omitted	omitted	0.195***	0.170***	0.164***	

Table D4.1. Labor market effects from technology (Dependent variable: e_i)

	[0.003]			[0.017]	[0.015]	[0.0003]
Other methodological aspects						
Technology and trade	-0.007**	-0.007**	-0.016***	7.865	omittad	omitted
(ref. only technology)	[0.002]	[0.002]	[0.003]	[6.905]	omiliea	omiliea
Time series	-0.006	-0.007	-0.008***	7.441	-0.290**	-0.308**
(ref. cross-sectional)	[0.007]	[0.006]	[0.0004]	[5.512]	[0.108]	[0.129]
Lagged independent variable	omittad	omittad	omittad	-0.023	omitted	omitted
(ref. contemporaneous ind. var.)	omitieu	omitieu	omitieu	[0.024]	omilieu	omilieu
Dependent variable in levels	4.192	-2.906	-7.422	7.440	0.292**	0.308**
(ref. first-differenced dep. var)	[3.161]	[2.354]	[8.060]	[5.510]	[0.110]	[0.129]
Controlling for endogeneity	-0.010*	-0.012**	-0.041***	0.022	omitted	omitted
(ref. no IV approach)	[0.004]	[0.003]	[0.001]	[0.019]	omilieu	omitteu
Hours worked				0.0001***	0.0001***	0.0001***
(ref. number of workers)				$[8.6*10^{-8}]$	$[6.7*10^{-10}]$	$[7.8*10^{-10}]$
Publication characteristics (K-)						
Peer reviewed paper	-3.595	omitted	omitted	-14.147	0.708	0.839***
(ref. working paper)	[2.111]	omitieu	omitieu	[10.600]	[0.458]	[0.087]
Peer reviewed paper * st error	1.784	omitted	omitted	3.049**	omitted	omitted
r eer reviewed paper - st. error	[1.272]	[1.272]		[1.198]	omineu	omittea
Publication year	0.688	-0.399	-0.977**	-0.351*	-0.041	-0.079***
r ubileation year	[0.586]	[0.442]	[0.315]	[0.181]	[0.057]	[0.001]
Publication year * st error	-1.554**	-1.663***	-1.029*	-0.471	-0695	-0.245
i doneation year st. error	[0.380]	[0.295]	[0.572]	[0.523]	[0.742]	[0.721]
10-Vear impact factor	3.933	8.242	18.700	-14.363	omitted	omitted
10 Tear impact factor	[2.413]	[5.850]	[21.748]	[9.476]	omnica	omitica
10-Vear impact factor * st_error	-6.389**	-5.781**	-4.046	3.081	omitted	omitted
To Tear impact factor st. error	[1.887]	[1.593]	[4.845]	[2.155]	omnea	omitica
Geography (ref. US-UK-Canada-Japan)						
Europe	0.048	omitted	omitted	0.008*	omitted	omitted
Luiope	[0.095]	ommed	ommed	[0.004]	omnea	Smirica
Spatial unit (ref. national-level analysis)						
Regional – level analysis	2.162	omitted	omitted	-21.518	0.933**	1.101***

	[16.074]	[0.415]	[0.058]			
Development status (ref. not-specified)						
Developed	omitted	omitted	omitted	-0.890 [0.899]	0.580** [0.216]	0.495*** [0.015]
Developing	0.559*** [0.067]	0.541*** [0.060]	1.004*** [0.058]	-8.128 [7.006]	-0.020 [0.164]	0.008 [0.016]
Time (ref. midpoint before 2001)						
After 2001	-0.311** [0.088]	-0.286*** [0.055]	-0.327** [0.081]	7.394 [5.407]	omitted	omitted
Other controls						
Dataspan	-0.347*** [0.010]	-0.347*** [0.007]	-0.383*** [0.011]	0.021* [0.012]	0.017 [0.015]	0.024*** [0.0001]
Number of RHS variables	2.8*10 ⁻⁶ [0.00003]	omitted	omitted	-0.022 [0.021]	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important controls or</i> <i>fixed effects</i>)	-0.043 [0.059]	omitted	omitted	-0.042** [0.018]	-0.006 [0.034]	0.006** [0.002]
Observations	124	124	124	276	276	276
Adjusted R^2	0.731	0.784	0.575	0.278	0.229	0.222
Primary study fixed effects	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes
Country by Midpoint fixed effects	yes	yes	yes	yes	yes	yes

Note: The table replicates the main analysis, after excluding elasticities expressed in shares. Robust st. errors, clustered by primary study are reported in brackets. All models are weighted by the inverse of the variance of the collected estimates. Columns 2 and 6 are based on the respective weighted-least-squares models (Columns 1 and 5), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories are significant at the 10% level. * for p<0.1, ** for p<0.05 and *** for p<0.01.

		Wage e	elasticities			Employment	elasticities	
Variable	WLS	WLS	PEESE	WLS	WLS	WLS	PEESE	WLS
		general to		general to		general to specific		general to
		specific		China		speeme		China
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
FAT – PET – PEESE tests								
Intercept (β_0) (PET) or (γ_0) (PEESE)	1.562** [0.460]	1.893*** [0.275]	1.535*** [0.153]	0.009 [0.243]	-10.744*** [0.982]	-0.136*** [0.26]	-0.165*** [0.023]	-2.450 [6.274]
St. error (β_1) (FAT) or (γ_1)	0.398 [0.606]	0.498 [0.401]	-0.700*** [0.689]	-0.752 [0.982]	0.976 [0.817]	0.077 [0.325]	-0.314 [0.208]	-5.054 [3.314]
Type of trade (ref. trade in final goods)								
Trade in intermediates	-0.002** [0.0009]	-0.003*** [0.0007]	-0.003*** [0.0006]	-0.003** [0.0008]	-0.085 [0.091]	omitted	omitted	omitted
Data aggregation (ref. worker-level)								
Industry-level	-0.004 [0.004]	-0.009 [0.007]	-0.008 [0.006]	1.917* [0.967]	11.303*** [1.074]	omitted	omitted	omitted
Firm-level	0.013* [0.007]	0.016** [0.007]	0.016** [0.006]	omitted	11.805*** [1.155]	omitted	omitted	omitted
Sectoral Ievel (ref. economy-wide)								
Manufacturing	0.015*** [0.003]	0.013*** [0.002]	0.015*** [0.00005]	0.019** [0.004]	-0.157*** [0.037]	-0.131*** [0.023]	-0.089*** [0.013]	1.360** [0.495]
Services	-0.001 [0.010]	-0.004 [0.010]	-0.0004 [0.009]	-0.037 [0.049]	-0.145** [0.040]	-0.109*** [0.022]	-0.062** [0.018]	1.597** [0.517]
Skill-level (ref. varied-skill)								
Low-skill	0.006 [0.004]	0.004 [0.004]	0.005 [0.004]	-0.0001 [0.001]	-0.003* [0.001]	-0.003** [0.001]	-0.001 [0.002]	-0.085 [0.048]
Medium-skill	0.008* [0.004]	0.004 [0.003]	0.006* [0.003]	omitted	-0.049*** [0.012]	-0.046*** [0.008]	-0.035*** [0.008]	omitted
High-skill	0.015** [0.005]	0.013** [0.004]	0.013** [0.004]	0.003** [0.001]	0.0008 [0.0006]	0.002** [0.0006]	0.001* [0.0008]	0.175 [0.101]

Table D4.2. Labor market effects from trade (Dependent variable: e_i)

Other methodological aspects								
Technology and trade	-0.0004	omitted	omitted	omitted	-0.022**	omitted	omitted	omitted
(ref. only trade)	[0.001]	omitica	omned	omnica	[0.010]	omnica	omineu	omnica
Time series	0.007	omitted	omitted	omitted	-0.473**	omitted	omitted	omitted
(ref. cross – sectional)	[0.008]				[0.170]			
Lagged independent variable	0.0007	omitted	omitted	omitted	-0.0006	omitted	omitted	omitted
(ref. contemporaneous)	[0.001]				[0.0006]			
Dependent variable in levels	0.005	omitted	omitted	omitted	0.005	omitted	omitted	omitted
(ref. first-alfferenced dep. var.)	[0.050]				[0.003]			
(ref. no-W approach)	0.007	omitted	omitted	omitted	0.007	omitted	omitted	omitted
(<i>rej. no-1v approach</i>) Hours worked	[0.000]				0 299***	0 305***	0 241***	0 148
(ref number of workers)					[0.050]	[0.049]	[0.045]	[0,161]
Publication characteristics					[0:000]	[01017]	[0:0:0]	[0.101]
Peer reviewed paper	-0.891***	-1.041***	-0.856***	-1.955***	-0.247***	-0.189***	-0.119	1.362
(ref. working paper)	[0.171]	[0.142]	[0.078]	[0.988]	[0.053]	[0.041]	[0.041]	[1.762]
Deer reviewed paper * st. orror	-0.005	omittad	amittad	amittad	-1.522	omittad	amittad	amittad
reel leviewed paper · st. enor	[0.641]	omiliea	omiliea	omiliea	[1.031]	omitiea	omitiea	omittea
Publication year	0.129**	0.133***	0.111***	0.457*	0.154***	-0.234***	-0.187***	-1.329
r ubileation year	[0.037]	[0.013]	[0.006]	[0.220]	[0.040]	[0.032]	[0.032]	[7.562]
Publication year * st_error	-0.141	omitted	omitted	omitted	1.452**	1.337**	0.577	5.590
r donoution your st. onor	[0.615]	omnica		omneu	[0.497]	[0.375]	[0.390]	[4.798]
10-Year impact factor	1.391**	1.689***	1.370***	0.466***	5.850***	omitted	omitted	omitted
1	[0.415]	[0.244]	[0.136]	[0.032]	[0.760]			
10-Year impact factor * st. error	0.308	omitted	omitted	omitted	-0.757	omitted	omitted	omitted
Casegraphy (not non OECD)	[0.466]				[0.896]			
Geography (rej. non-OECD)	-0.008**	-0.007**	_0.011***	-0.041	3 818***			
Europe	-0.008	[0.007	10.011	-0.041	[0 492]	omitted	omitted	omitted
Spatial unit (ref. national-level analysis	(0.005) ()	[0.002]	[0.0007]	[0.005]	[0.472]			
	0.001				2.668***	2.907***	2.538***	1.454**
Regional – level analysis	[0.070]	omitted	omitted	omitted	[0.276]	[0.187]	[0.021]	[0.593]

Development status (ref. not specified)								
Developed	0.004 [0.003]	0.008** [0.004]	0.009* [0.005]	omitted	-0.024* [0.012]	omitted	omitted	omitted
Developing	-0.025 [0.019]	-0.023 [0.018]	-0.022 [0.018]	omitted	-0.028 [0.019]	omitted	omitted	omitted
<i>Time (ref. midpoint before 2001)</i>								
After 2001	-0.659** [2.234]	-0.812*** [0.138]	-0.630*** [0.076]	-0.088*** [0.0005]	-0.125*** [0.025]	-0.140*** [0.018]	-0.099*** [0.001]	0.262** [0.071]
Other controls								
Dataspan Number of RHS variables	-0.397*** [0.038] 0.0003	-0.416*** [0.024] omitted	-0.373*** [0.013] omitted	-0.345** [0.059] omitted	0.057*** [0.011] 0.0004	0.064*** [0.010] omitted	0.043*** [0.001] omitted	0.043 [0.029] omitted
Including important controls or fixed effects (<i>ref. missing important</i> controls or fixed effects)	[0.0004] 0.015*** [0.004]	0.018*** [0.002]	0.016*** [0.002]	0.015*** [0.002]	0.0006 [0.024]	omitted	omitted	omitted
Observations	378	378	378	64	557	557	557	59
Adjusted R^2	0.910	0.912	0.912	0.621	0.993	0.993	0.992	0.761
Primary study fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Country x Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes

Note: The table replicates the main analysis, after excluding elasticities expressed in shares. Robust st. errors, clustered by primary study are reported in brackets. All models are weighted by the inverse of the variance of the collected estimates. Columns 2 and 5 are based on the respective weighted-least-squares models (Columns 1 and 4), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories are significant at the 10% level. * for p<0.1, ** for p<0.05 and *** for p<0.01.

Appendix D5. Estimation outcomes excluding semi-elasticities, calculated elasticities and elasticities where the dependent variable is expressed in shares and winsorizing.

	V	Vage elasticitie	S	Employment elasticities		
Variable	WLS	WLS	PEESE	WLS	WLS	PEESE
		general to			general to	
		specific			specific	
	(1)	(2)	(3)	(4)	(5)	(6)
FAT – PET – PEESE tests	0.000	0.044444	0.044	40 	0.4.50.1.1.1	0.477
Intercept (β_0) (PET) or (γ_0) (PEESE)	-0.092***	-0.046***	-0.041	48.673***	0.462***	0.657***
	[0.002]	[0.008]	[0.048]	[10.897]	[0.105]	[0.069]
St error (β_1) or st error ² (γ_1) (FAT)	0.103	0.073	-0.312	-0.631	0.834**	0.624
	[0.158]	[0.135]	[0.738]	[659]	[0.334]	[1.276]
Type of innovation (ref. process innovation	n)					
Product innovation	0.001	omitted	omitted	0.001***	0.001***	0.001***
1 Todact milovation	[0.007]	omned	omilieu	[0.0002]	[0.0002]	[0.0002]
Data aggregation (ref. worker-level)						
Industry_level	0.086*	omitted	omitted	-4.970***	-0.435***	-0.544***
industry-lever	[0.040]	ommeu	omilieu	[1.077]	[0.060]	[0.036]
Firm laval	0.094	omitted	omitted	-2.225***	-0.029***	-0.041***
Глин-неуст	[0.102]	omitieu	omitieu	[0.478]	[0.005]	[0.0004]
Sectoral Ievel (ref. economy-wide)						
Manufacturing	0.198	0.785***	0.885	0.005**	0.004	0.008^{***}
Wanutacturing	[0.118]	[0.186]	[1.077]	[0.002]	[0.002]	[0.001]
Samiaaa	amitta d	amitta d	amitta d	0.019**	0.015	0.027***
Services	omittea	omiliea	omiliea	[0.008]	[0.010]	[0.006]
Skill-level (ref. varied skill)						
L ovy shill	-0.005***	omitted.	omittad	-8.292***	-0.066***	-0.075***
Low-skiii	[0.0009]	omiliea	omiliea	[1.890]	[0.009]	[0.0004]
Madimu -1-11	-0.301**	: 44 - J		i44 - J		:44 - 1
Medium-skill	[0.077]	omittea	omitted	omitted	omittea	omittea

Table D5.1. Labor market effects from technology (Dependent variable: e_i)

High-skill	-0.006*** [0.0008]	omitted	omitted	-8.937*** [1.827]	-1.094** [0.473]	-1.838 [5.929]
Other methodological aspects						
Technology and trade (<i>ref. only technology</i>)	-0.199 [0.208]	-0.542** [0.169]	-0.636 [0.889]	3.260*** [0.713]	omitted	omitted
Time series	-0.010***	-0.011***	-0.009***	0.915***	-0.132***	-0.127***
(ref. cross-sectional)	[0.005]	[0.005]	[0.0005]	[0.218]	[0.005]	[0.002]
Lagged independent variable (<i>ref. contemporaneous ind. var.</i>)	omitted	omitted	omitted	-0.031 [0.022]	omitted	omitted
Dependent variable in levels	• • • • • •	-0.525**	-0.596	-0.915***	0.132**	0.127***
(ref. first-differenced dep. var)	omittea	[0.125]	[0.742]	[0.218]	[0.005]	[0.002]
Controlling for endogeneity	-0.312	-1.014**	-1.175	0.004	• • • •	• 1
(ref. no IV approach)	[0.275]	[0.271]	[1.494]	[0.003]	omitted	omitted
Hours worked				0.0001***	0.0001***	0.0001***
(ref. number of workers)				[6.1*10 ⁻¹¹]	$[1.6*10^{-12}]$	$[2.2*10^{-12}]$
Publication characteristics (K-)						
Peer reviewed paper	::44 - 1		: 44 - J	7.282***	-0.424***	-0.507***
(ref. working paper)	omittea	omiliea	omiliea	[1.672]	[0.048]	[0.035]
Peer reviewed paper * st. error	-0.174** [0.057]	omitted	omitted	0.909** [0.335]	omitted	omitted
Publication year	0.056	0.183**	0.208	-3.718***	-0.128***	-0.132***
i ubileation year	[0.086]	[0.060]	[0.329]	[0.818]	[0.003]	[0.002]
Publication year * st error	0.050**	-0.055**	-0.465	-0.115	0.098	-0.098
i doncation year st. citor	[0.019]	[0.021]	[1.180]	[0.185]	[0.194]	[0.955]
10-Year impact factor	-0.155	-0.779	-0.842	114.85**	omitted	omitted
10 Tour impact factor	[0.151]	[0.182]	[0.992]	[25.803]	omnea	omnica
10-Year impact factor * st. error	0.160 [0.310]	0.905** [0.277]	2.260 [3.980]	-0.886 [1.530]	omitted	omitted
Geography (ref. US-UK-Canada-Japan)						
Europe	0.059*** [0.007]	omitted	omitted	0.011*** [0.001]	omitted	omitted
Spatial unit (ref. national-level analysis)						

Regional – level analysis	omitted	omitted	omitted	-0.118* [0.059]	-0.424*** [0.048]	0.577*** [0.008]
Development status (ref. not-specified)						
Developed	omitted	omitted	omitted	omitted	omitted	omitted
Developing	omitted	omitted	omitted	-2.935*** [0.562]	-0.177 [0.402]	0.284 [1.294]
Time (ref. midpoint before 2001)						
After 2001	omitted	omitted	omitted	6.704*** [1.493]	omitted	omitted
Other controls						
Dataspan	-0.076** [0.032]	-0.114*** [0.021]	-0.152 [0.115]	0.029*** [0.003]	0.032*** [0.004]	0.024*** [0.0004]
Number of RHS variables	0.00003 [0.00001]	omitted	omitted	-0.007** [0.003]	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important controls or</i> <i>fixed effects</i>)	-0.043 [0.059]	omitted	omitted	0.025 [0.080]	0.109** [0.052]	0.010 [0.034]
Observations	58	58	58	198	198	198
Adjusted R^2	0.563	0.616	0.615	0.848	0.817	0.803
Primary study fixed effects	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes
Country by Midpoint fixed effects	yes	yes	yes	yes	yes	yes

Note: The table replicates the main analysis, after excluding semi-elasticities, calculated elasticities and elasticities expressed in shares and winsorizing using the 20^{th} and the 80^{th} percentile. Robust st. errors, clustered by primary study are reported in brackets. All models are weighted by the inverse of the variance of the collected estimates. Columns 2 and 6 are based on the respective weighted-least-squares models (Columns 1 and 5), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories are significant at the 10% level. * for p<0.1, ** for p<0.05 and *** for p<0.01.

	V	Vage elasticit	ies	Employment elasticities		
Variable	WLS	WLS	PEESE	WLS	WLS	PEESE
		general to			general to	
		specific			specific	
	(1)	(2)	(3)	(4)	(5)	(6)
FAT – PET – PEESE tests						
Intercept (\mathcal{B}_{0}) (PET) or (γ_{0}) (PEESE)	0.085	-1.779**	-0.571***	3.800	0.046**	0.018
	[4202.8]	[0.709]	[0.111]	[4.072]	[0.012]	[0.011]
St. error (β_1) (FAT) or (γ_1)	1.758	0.787*	0.207	-0.032	0.318	-0.070
	[2.239]	[0.431]	[0.252]	[0.638]	[0.267]	[0.091]
Type of trade (ref. trade in final goods)	0.000	0.000	0.000			
Trade in intermediates	-0.002	-0.002**	-0.003***	-0.095**	omitted	omitted
	[0.008]	[0.0009]	[0.0001]	[0.023]		
Data aggregation (ref. worker-level)	0.004	0.010	0.000	2 007		
Industry-level	-0.004	-0.010	-0.009	-2.887	omitted	omitted
	[0.010]	[0.007]	[0.006]	[3.381]		
Firm-level	0.007	0.015*	0.010***	-2.757	omitted	omitted
Sectoral Isual (ref. coor any wide)	[0.011]	[0.007]	[0.000]	[3.341]		
Sectoral Tevel (rej. economy-wide)	0.102	0 255**	0 107**	0 122***	0 11/***	0.001***
Manufacturing	0.195	0.555^{++}	0.107^{+1}	-0.122	-0.114	-0.081
	[26403.1]	[0.141]	[0.027]	0.000***	[0.017]	0.066***
Services	[28/65 1]	0.528 [0.139]	[0.026]	-0.090	-0.102	-0.000
Skill-lovel (ref varied-skill)	[20+03.1]	[0.137]	[0.020]	[0.021]	[0.010]	[0.014]
Skill-level (rej. variea-skill)	0.010	0.006	0.007	-0.002**	-0.002***	-0.0008
Low-skill	[0 011]	[0.005]	[0.007	[0 0004]	[0.0002]	[0 0005]
	0.011	0.005	0.007**	-0.012**	-0.013**	-0.005**
Medium-skill	[0.018]	[0.004]	[0.003]	[0.003]	[0.005]	[0.002]
	0.019	0.014**	0.015**	0.0007	0.002***	0.001***
High-skill	[0.012]	[0.004]	[0.004]	[0.0009]	[0.0002]	[0.0002]
Other methodological aspects	[]	[]	[]	[]	[]	[]

Table D5.2. Labor market effects from trade (Dependent variable: e_i)

Technology and trade (ref. only trade)	-0.0008 [0.014]	omitted	omitted	0.239*** [0.003]	omitted	omitted
Time series (<i>ref. cross – sectional</i>)	0.013 [0.015]	omitted	omitted	-0.120** [0.042]	omitted	omitted
Lagged independent variable (ref. contemporaneous)	-0.0008 [0.009]	omitted	omitted	-0.0008 [0.001]	omitted	omitted
Dependent variable in levels (<i>ref. first-differenced dep. var.</i>)	-0.041 [0.175]	omitted	omitted	0.038*** [0.006]	omitted	omitted
Controlling for endogeneity (<i>ref. no-IV approach</i>)	0.014 [0.012]	omitted	omitted	0.009 [0.007]	omitted	omitted
Hours worked				-0.064	-0.178***	-0.181***
(ref. number of workers)				[0.140]	[0.008]	[0.004]
Publication characteristics						
Peer reviewed paper	-0.142	0.035*	0.016	0.145	0.243***	0.252
(ref. working paper)	[1543.3]	[0.017]	[0.010]	[0.161]	[0.005]	[0.004]
Peer reviewed paper * st. error	-1.287 [2.587]	omitted	omitted	-0.122 [594]	omitted	omitted
Dublication year	0.071	0.108**	0.025**	0.094**	0.085***	0.097***
Fublication year	[1608.0]	[0.046]	[0.008]	[0.041]	[0.007]	[0.003]
Dublication year * at amon	-0.907	anitta d	amitta d	0.685**	0.819**	0.127
Publication year * st. error	[1.212]	omiliea	omitiea	[0.218]	[0.303]	[0.163]
10 Veer immed feater	0.389	-2.406**	-0.813***	-4.795		
10- Year Impact factor	[41063.2]	[0.934]	[0.149]	[5.515]	omittea	omittea
10-Year impact factor * st. error	0.218 [0.812]	omitted	omitted	-0.897 [0.551]	omitted	omitted
Geography (ref. non-OECD)						
Europe	-0.025 [806.5]	0.034 [0.021]	0.002 [0.007]	-3.574 [3.951]	omitted	omitted
Spatial unit (ref. national-level analysis	s)					
Regional – level analysis	0.110 [33863.2]	omitted	omitted	-3.383 [3.557]	0.137*** [0.011]	0.147*** [0.011]
Development status (ref. not specified)						

Developed	-0.002 [0.017]	0.010** [0.002]	0.014*** [0.001]	-0.013* [0.007]	omitted	omitted
Developing	-0.019 [0.027]	-0.008** [0.002]	-0.005*** [0.003]	-0.014 [0.009]	omitted	omitted
<i>Time (ref. midpoint before 2001)</i>						
After 2001	-0.170 [5205.9]	0.290** [0.107]	0.120*** [0.023]	-0.289*** [0.061]	-0.216*** [0.010]	-0.224*** [0.005]
Other controls						
Dataspan	0.003 [0.057]	0.002 [0.001]	0.004*** [0.0004]	-0.019** [0.007]	-0.018*** [0.004]	-0.005*** [0.0002]
Number of RHS variables	-0.0001 [0.002]	omitted	Omitted	-0.00006 [0.0009]	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important</i> controls or fixed effects)	0.153 [0.355]	0.064 [0.037]	0.006 [0.004]	-0.097 [0.112]	omitted	omitted
Observations	225	225	225	408	408	408
Adjusted R^2	0.920	0.922	0.913	0.997	0.997	0.997
Primary study fixed effects	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes
Country x Midpoint fixed effects	ves	ves	ves	ves	ves	ves

Country x Midpoint fixed effectsyesyesyesyesyesyesNote: The table replicates the main analysis, after excluding semi-elasticities, calculated elasticities and elasticities expressed in shares and winsorizing using the 20^{th} and
the 80^{th} percentile. Robust st. errors, clustered by primary study are reported in brackets (except for Column 1, where, to achieve convergence, the estimation is based on
simple st. errors). All models are weighted by the inverse of the variance of the collected estimates. Columns 2 and 5 are based on the respective weighted-least-squares
models (Columns 1 and 4), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the
reported categories are significant at the 10% level. The estimations for trade with China are dropped due to insufficient observations. * for p<0.1, ** for p<0.05 and ***
for p<0.01.</td>

Appendix E. Random and fixed effects analysis

Appendix E1. Random effects meta-regression analysis

Table E1.1.	Labor market	effects from	technology	(Dependent	variable: e_i)
			01	`	L/

	W	age elasticities	5	Employment elasticities				
Variable	Random effects	Random effects general to specific	Random effects PEESE	Random effects	Random effects general to specific	Random effects PEESE		
	(1)	(2)	(3)	(4)	(5)	(6)		
FAT – PET – PEESE tests								
Intercept (β_0) (PET) or (γ_0) (PEESE)	2.178 [3.247]	3.109 [3.833]	0.897 [5.970]	-0.107 [0.248]	0.159 [0.123]	-0.122 [0.315]		
St. error (β_1) or st. error ² (γ_1) (FAT)	3.688 [3.211]	2.365 [3.702]	0.677 [6.222]	-1.820 [1.356]	-0.158 [0.353]	-1.286 [1.509]		
Type of innovation (ref. process innovation	ı)							
Product innovation	-1.391 [1.642]	omitted	omitted	0.320 [0.216]	0.282 [0.201]	0.278 [0.194]		
Data aggregation (ref. worker-level)								
Industry-level	0.682** [0.183]	omitted	omitted	3.830** [1.700]	2.217** [0.611]	3.946* [2.195]		
Firm-level	-10.139 [6.657]	omitted	omitted	8.305*** [2.108]	6.811*** [1.001]	8.594** [2.574]		
Sectoral Ievel (ref. economy-wide)								
Manufacturing	0.108 [0.078]	-0.068 [0.136]	0.006 [0.058]	0.012 [0.520] 0.204*	-0.031 [0.516] 0.258	-0.016 [0.513]		
Services	[0.090]	[0.182]	[0.040]	[0.173]	[0.177]	[0.172]		
Skill-level (ref. varied skill)								
Low-skill	-0.496 [0.496]	omitted	omitted	-1.153 [0.972]	-1.232 [0.962]	-1.202 [0.962]		

Medium-skill	-0.535	omitted	omitted	-1.159	-1.172	-1.180
	[0.476]			[0.990]	[0.970]	[0.978]
High-skill	0.498 [0.387]	omitted	omitted	2.133** [0.957]	2.212** [0.956]	2.092** [0.948]
Other methodological aspects						
Technology and trade	-0.032***	-0.035**	-0.032***	0.0009	amitta d	amitta d
(ref. only technology)	[0.003]	[0.009]	[0.004]	[0.127]	omiliea	omiliea
Time series	0.259	0.159	-0.042	-0.362***	-0.443***	-0.430***
(ref. cross-sectional)	[0.376]	[0.558]	[0.352]	[0.070]	[0.032]	[0.024]
Lagged independent variable	-0.013	omittad	omittad	-0.559*	amittad	omittad
(ref. contemporaneous ind. var.)	[0.021]	omiliea	omiliea	[0.300]	omiliea	omiliea
Dependent variable in levels	0.149	-0.159	-0.127**	0.373***	0.450***	0.432***
(ref. first-differenced dep. var)	[0.167]	[0.096]	[0.059]	[0.067]	[0.032]	[0.024]
Controlling for endogeneity	-0.149	-0.125	-0.157	0.012	amittad	omittad
(ref. no IV approach)	[0.132]	[0.117]	[0.106]	[0.072]	omiliea	omiliea
Wages in shares	-3.446	omittad	omittad			
(ref. wages not in shares)	[5.723]	omiliea	omiliea			
Hours worked				-0.004***	0.0004^{***}	0.0004^{***}
(ref. number of workers)				[2 0*10-12]	[2 5×10-14]	$r = \pi + 10^{-1/1}$
(10) 11111001 09 (10) 1101 10)				$[3.8*10^{-1}]$	$[2.5*10^{-1}]$	$[5.7*10^{-14}]$
Publication characteristics (K-)				[3.8*10]	[2.5*10**]	$[5.7*10^{-14}]$
Publication characteristics (K-) Peer reviewed paper	-7.891	omitted	omittad	-0.088	-2.460***	[5.7*10 ⁻¹⁴] -2.751***
Publication characteristics (K-) Peer reviewed paper (ref. working paper)	-7.891 [5.829]	omitted	omitted	-0.088 [1.623]	-2.460*** [0.600]	[5.7*10 ⁻¹⁴] -2.751*** [0.657]
Publication characteristics (K-) Peer reviewed paper (ref. working paper)	-7.891 [5.829] -3.091**	omitted	omitted	-0.088 [1.623] 0.724	-2.460*** [0.600]	[5.7*10 ⁻¹⁴] -2.751*** [0.657]
Publication characteristics (K-)Peer reviewed paper (ref. working paper)Peer reviewed paper * st. error	-7.891 [5.829] -3.091** [1.217]	omitted omitted	omitted omitted	-0.088 [1.623] 0.724 [1.131]	-2.460*** [0.600] omitted	[5.7*10 ^{-1*}] -2.751*** [0.657] <i>omitted</i>
Publication characteristics (K-) Peer reviewed paper (ref. working paper) Peer reviewed paper * st. error Publication veer	-7.891 [5.829] -3.091** [1.217] -2.715*	omitted omitted 23.720	omitted omitted 6.429	-0.088 [1.623] 0.724 [1.131] 1.373**	-2.460*** [0.600] <i>omitted</i> 1.663**	[5.7*10 ^{-1*}] -2.751*** [0.657] <i>omitted</i> 1.650**
Publication characteristics (K-)Peer reviewed paper (ref. working paper)Peer reviewed paper * st. errorPublication year	-7.891 [5.829] -3.091** [1.217] -2.715* [1.528]	omitted omitted 23.720 [30.191]	omitted omitted 6.429 [41.265]	-0.088 [1.623] 0.724 [1.131] 1.373** [0.572]	-2.460*** [0.600] <i>omitted</i> 1.663** [0.591]	[5.7*10 ^{-1*}] -2.751*** [0.657] <i>omitted</i> 1.650** [0.630]
Publication characteristics (K-) Peer reviewed paper (ref. working paper) Peer reviewed paper * st. error Publication year	-7.891 [5.829] -3.091** [1.217] -2.715* [1.528] 0.485	<i>omitted</i> <i>omitted</i> 23.720 [30.191] -0.270	<i>omitted</i> <i>omitted</i> 6.429 [41.265] 0.207	-0.088 [1.623] 0.724 [1.131] 1.373** [0.572] 0.360	-2.460*** [0.600] <i>omitted</i> 1.663** [0.591] -0.444	[5.7*10 ^{-1*}] -2.751*** [0.657] <i>omitted</i> 1.650** [0.630] 0.397**
Publication characteristics (K-)Peer reviewed paper (ref. working paper)Peer reviewed paper * st. errorPublication yearPublication year * st. error	-7.891 [5.829] -3.091** [1.217] -2.715* [1.528] 0.485 [0.668]	<i>omitted</i> <i>omitted</i> 23.720 [30.191] -0.270 [0.558]	<i>omitted</i> <i>omitted</i> 6.429 [41.265] 0.207 [0.442]	-0.088 [1.623] 0.724 [1.131] 1.373** [0.572] 0.360 [0.269]	-2.460*** [0.600] <i>omitted</i> 1.663** [0.591] -0.444 [1.682]	[5.7*10 ^{-1*}] -2.751*** [0.657] <i>omitted</i> 1.650** [0.630] 0.397** [0.184]
Publication characteristics (K-)Peer reviewed paper (ref. working paper)Peer reviewed paper * st. errorPublication yearPublication year * st. error10 Year impact factor	-7.891 [5.829] -3.091** [1.217] -2.715* [1.528] 0.485 [0.668] -12.422	<i>omitted</i> <i>omitted</i> 23.720 [30.191] -0.270 [0.558] -16.502	<i>omitted</i> <i>omitted</i> 6.429 [41.265] 0.207 [0.442] -5.025	-0.088 [1.623] 0.724 [1.131] 1.373** [0.572] 0.360 [0.269] 6.628**	-2.460*** [0.600] <i>omitted</i> 1.663** [0.591] -0.444 [1.682] <i>omitted</i>	[5.7*10 ⁻¹⁴] -2.751*** [0.657] omitted 1.650** [0.630] 0.397** [0.184] omitted
Publication characteristics (K-)Peer reviewed paper (ref. working paper)Peer reviewed paper * st. errorPublication yearPublication year * st. error10-Year impact factor	-7.891 [5.829] -3.091** [1.217] -2.715* [1.528] 0.485 [0.668] -12.422 [11.086]	<i>omitted</i> <i>omitted</i> 23.720 [30.191] -0.270 [0.558] -16.502 [20.023]	<i>omitted</i> <i>omitted</i> 6.429 [41.265] 0.207 [0.442] -5.025 [27.199]	-0.088 [1.623] 0.724 [1.131] 1.373** [0.572] 0.360 [0.269] 6.628** [3.004]	-2.460*** [0.600] <i>omitted</i> 1.663** [0.591] -0.444 [1.682] <i>omitted</i>	[5.7*10 ^{-1*}] -2.751*** [0.657] <i>omitted</i> 1.650** [0.630] 0.397** [0.184] <i>omitted</i>
Publication characteristics (K-) Peer reviewed paper (ref. working paper) Peer reviewed paper * st. error Publication year Publication year * st. error 10-Year impact factor * st. error	-7.891 [5.829] -3.091** [1.217] -2.715* [1.528] 0.485 [0.668] -12.422 [11.086] 0.354	<i>omitted</i> <i>omitted</i> 23.720 [30.191] -0.270 [0.558] -16.502 [20.023] 1.849	<i>omitted</i> <i>omitted</i> 6.429 [41.265] 0.207 [0.442] -5.025 [27.199] 3.816	-0.088 [1.623] 0.724 [1.131] 1.373** [0.572] 0.360 [0.269] 6.628** [3.004] -0.486	-2.460*** [0.600] <i>omitted</i> 1.663** [0.591] -0.444 [1.682] <i>omitted</i>	[5.7*10 ⁻¹⁴] -2.751*** [0.657] omitted 1.650** [0.630] 0.397** [0.184] omitted amitted

Geography (ref. US-UK-Canada-Japan)						
Europe	0.041 [0.758]	omitted	omitted	0.073** [0.023]	omitted	omitted
Spatial unit (ref. national-level analysis)						
Regional – level analysis	0.826*** [0.135]	omitted	omitted	0.635 [0.640]	0.582 [0.436]	0.807** [0.304]
Development status (ref. not-specified)						
Developed	omitted	omitted	omitted	0.383** [0.055]	0.349 [0.234]	0.419*** [0.055]
Developing	0.333 [0.309]	0.435 [0.319]	0.326 [0.325]	0.203 [0.171]	0.606* [0.347]	0.265* [0.153]
Time (ref. midpoint before 1995)						
After 2001	-0.764** [0.293]	-0.473* [0.263]	-0.450 [0.293]	0.312 [0.190]	omitted	omitted
Other controls						
Dataspan	-0.599*** [0.022]	-0.596*** [0.032]	-0.587*** [0.014]	1.333*** [0.150]	1.365*** [0.081]	1.340*** [0.080]
Number of RHS variables	-0.0004 [0.003]	omitted	omitted	0.015 [0.0026]	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important controls or</i> <i>fixed effects</i>)	-0.473** [0.151]	omitted	omitted	-0.489** [0.238]	-0.542** [0.207]	-0.497** [0.217]
Observations	231	231	231	392	392	392
Adjusted R ²	0.504	0.368	0.372	0.662	0.655	0.656
Primary study fixed effects	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes
Country by Midpoint fixed effects	yes	yes	yes	yes	yes	ves

Note: St. errors clustered by primary study are reported in brackets. The st. errors are corrected for the random effects estimates, based on Doucouliagos et al. (2022). Columns 2 and 5 are based on the respective WLS-MRA models (Columns 1 and 4), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories is significant at the 10% level. * for p<0.1, ** for p<0.05 and *** for p<0.01.

	Wage elasticities				Employment elasticities			
Variable	Random effects	Random effects general to specific	Random effects PEESE	Random Effects general to specific - China	Random effects	Random effects general to specific	Random effects PEESE	Random Effects general to specific - China
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
FAT – PET – PEESE tests								
Intercept (β_0) (PET) or (γ_0) (PEESE)	0.006 [0.085]	0.004 [0.066]	0.038 [0.140]	-0.375 [0.366]	-0.095 [0.197]	-0.082 [0.199]	-0.136 [0.535]	-0.494 [0.570]
St. error (β_1) (FAT) or (γ_1)	0.447 [1.164]	0.322 [1.196]	0.945 [2.500]	-5.394 [4.504]	-0.972 [0.807]	-0.623 [0.715]	-0.773 [1.971]	-13.945** [5.581]
Type of trade (ref. trade in final goods)								
Trade in intermediates	-0.100 [0.170]	-0.184 [0.173]	-0.182 [0.175]	-0.145** [0.039]	-0.557 [0.448]	omitted	omitted	omitted
Data aggregation (ref. worker-level)								
Industry-level	-0.485 [0.484]	0.018 [0.364]	0.005 [0.361]	7.554 [5.707]	-3.305* [1.787]	omitted	omitted	omitted
Firm-level	0.119 [0.315]	0.198 [0.488]	0.197 [0.484]	omitted	-2.525 [1.632]	omitted	omitted	omitted
Sectoral Ievel (ref. economy-wide)								
Manufacturing Services	0.420** [0.163] 0.103 [0.203]	0.305** [0.098] -0.039 [0.189]	0.292** [0.087] -0.046 [0.164]	0.533** [0.198] 1.831 [2.535]	-0.415*** [0.107] 0.039 [0.276]	-0.449** [0.127] -0.003 [0.168]	-0.421** [0.155] -0.016 [0.182]	9.667 [10.522] 10.522 [10.299]
Skill-level (ref. varied skill)	[0.200]	[01105]	[01101]	[2:000]	[0.270]	[01100]	[01102]	[10.299]
Low-skill	0.075 [0.138]	0.025 [0.118]	0.025 [0.117]	-0.252 [0.266]	-0.485* [0.253]	-0.455* [0.247]	-0.469* [0.247]	-0.742* [0.333]
Medium-skill	-0.431 [0.372]	-0.502 [0.348]	-0.503 [0.350]	omitted	-0.591 [0.455]	-0.568 [0.468]	-0.580 [0.462]	omitted
High-skill	0.400	0.334	0.336	-0.143	0.251	0.264	0.248	1.035**

Table E1.2. Labor market effects from trade (Dependent variable: e_i)

	[0.260]	[0.240]	[0.239]	[0.599]	[0.212]	[0.207]	[0.204]	[0.386]
Other methodological aspects								
Technology and trade (ref. only trade)	0.071 [0.092]	omitted	omitted	omitted	-0.144 [0.116]	omitted	omitted	omitted
Time series (<i>ref. cross-sectional</i>)	-0.620* [0.345]	omitted	omitted	omitted	-0.546 [0.757]	omitted	omitted	omitted
Lagged independent variable (<i>ref. contemporaneous ind. var.</i>)	0.033 [0.069]	omitted	omitted	omitted	0.068 [0.105]	omitted	omitted	omitted
Dependent variable in levels (<i>ref. first-differenced dep. var</i>)	-0.306 [0.505]	omitted	omitted	omitted	0.203 [0.151]	omitted	omitted	omitted
Controlling for endogeneity (ref. no IV approach)	0.138 [0.151]	omitted	omitted	omitted	0.073 [0.145]	omitted	omitted	omitted
Wages in shares (ref. wages not in shares)	-2.283 [3.760]	-5.263** [2.054]	-6.060** [2.954]	omitted				
Hours worked					-0.114***	-0.115***	-0.114***	-0.100***
(ref. number of workers)					[0.006]	[0.006]	[0.016]	[0.018]
Publication characteristics (K-)								
Peer reviewed paper	-10.318	3.124	3.301	-7.312	-0.654	-0.161	-0.171	0.913
(ref. working paper)	[15.852]	[6.405]	[6.027]	[5.677]	[0.694]	[0.288]	[0.237]	[2.522]
Peer reviewed paper * st. error	-0.435 [1.127]	omitted	omitted	omitted	0.749 [0.954]	omitted	omitted	omitted
Publication year	0.454	0.450*	0.425	4.445	1.730**	-2.492	-2.713	-13.845
i donounon you	[1.284]	[0.255]	[0.261]	[3.038]	[0.549]	[1.574]	[4.069]	[9.140]
Publication year * st. error	0.022	omitted	omitted	omitted	0.426	0.249	0.152	15.897**
	[0.886]		~ - 1 1		[0.387]	[0.216]	[0.410]	[5.925]
10-Year impact factor	-16.291 [29.540]	8.076 [11.215]	9.714 [11.087]	0.298 [3.442]	-0.461 [1.351]	omitted	omitted	omitted
10-Year impact factor * st. error	0.385 [0.384]	omitted	omitted	omitted	0.555 [0.591]	omitted	omitted	omitted
Geography (ref. US-UK-Canada-Japan)								
Europe	0.016 [0.155]	0.131 [0.092]	0.135 [0.085]	0.515 [1.824]	7.123*** [1.700]	omitted	omitted	omitted

Spatial unit (ref. national-level analysis	5)							
Regional – level analysis	-0.656 [0.482]	omitted	omitted	omitted	3.902*** [0.464]	3.520*** [0.330]	3.479*** [0.0.411]	0.927 [1.243]
Development status (ref. not-specified)								
Developed	0.095 [0.372]	0.117 [0.354]	0.134 [0.322]	omitted	-0.283 [0.339]	omitted	omitted	omitted
Developing	0.132 [0.482]	0.118 [0.392]	0.155 [0.375]	5.957 [4.661]	-0.935 [0.571]	omitted	omitted	omitted
Time (ref. midpoint before 1995)								
After 2001	-6.724 [15.967]	5.562 [6.317]	5.695 [5.949]	-1.075** [0.389]	-0.814*** [0.174]	-0.592*** [0.117]	-0.551** [0.179]	1.484*** [0.251]
Other controls								
Dataspan	-1.926*** [0.201]	-1.974*** [0.151]	-1.938*** [0.216]	-2.924** [0.727]	-0.070 [0.053]	-0.055 [0.040]	-0.063 [0.041]	0.063 [0.083]
Number of RHS variables	-0.016 [0.028]	omitted	omitted	omitted	0.008 [0.017]	omitted	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important</i> <i>controls or fixed effects</i>)	0.708 [0.421]	0.415** [0.203]	0.415** [0.193]	0.344* [0.162]	-0.277 [0.684]	omitted	omitted	omitted
Observations	459	459	459	64	635	635	635	60
Adjusted R ²	0.174	0.190	0.191	0.665	0.275	0.261	0.259	0.793
Primary study fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Country x Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes

Note: St. errors clustered by primary study are reported in brackets. The st. errors are corrected for the random effects estimates, based on Doucouliagos et al. (2022). Columns 2 and 6 are based on the respective weighted-least-squares models (Columns 1 and 5), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories are significant at the 10% level. * for p<0.05 and *** for p<0.01.
Appendix E2. Fixed effects meta-regression analysis (MRA)

	W	age elasticitie	S	Employment elasticities			
Variable	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	
	(1)	general to specific (2)	PEESE (3)	(4)	general to specific (5)	PEESE (6)	
FAT – PET – PEESE tests							
Intercept (β_0) (PET) or (γ_0) (PEESE)	-1.815 [2.465]	0.551** [0.243]	0.678** [0.242]	0.751*** [0.150]	0.706*** [0.141]	0.231 [0.141]	
St. error (β_1) or st. error ² (γ_1) (FAT)	-1.807*** [0.397]	-1.474*** [0.157]	2.018*** [0.321]	-1.540*** [0.244]	-1.296*** [0.097]	0.780*** [0.213]	
Type of innovation (ref. process innovation	n)						
Product innovation	0.005 [0.007]	omitted	omitted	0.001 [0.0008]	0.001* [0.0008]	0.001 [0.0008]	
Data aggregation (ref. worker-level)							
Industry-level	0.200 [0.145]	omitted	omitted	0.100 [0.110]	0.167** [0.084]	0.190** [0.084]	
Firm-level	0.925 [1.335]	omitted	omitted	0.982*** [0.194]	[0.183]	[0.183]	
Sectoral Ievel (ref. economy-wide)							
Manufacturing	-0.277 [0.248]	-0.225 [0.164]	-0.790*** [0.162]	0.227*** [0.007]	0.221*** [0.007]	0.200*** [0.007]	
Services	-0.270 [0.248]	-0.218 [0.164]	-0.783*** [0.162]	0.076*** [0.005]	0.068*** [0.005]	0.038*** [0.005]	
Skill-level (ref. varied skill)							
Low-skill	-0.003 [0.005]	omitted	omitted	0.061*** [0.011]	0.050*** [0.011]	0.012 [0.011]	
Medium-skill	-0.002 [0.008]	omitted	omitted	0.069*** [0.012]	0.060*** [0.012]	0.029** [0.012]	

Table E2.1. Labor market effects from technology (Dependent variable: e_i)

High-skill	-0.001 [0.005]	omitted	omitted	0.195*** [0.007]	0.187*** [0.007]	0.160*** [0.007]
Other methodological aspects						
Technology and trade	-0.008	-0.008	-0.017	0.009	omitted	omitted
(ref. only technology)	[0.103]	[0.103]	[0.104]	[0.052]	omitted	ommed
Time series	-0.010	-0.010	-0.008	-0.142***	-0.146***	-0.160***
(ref. cross-sectional)	[0.012]	[0.012]	[0.012]	[0.003]	[0.003]	[0.003]
Lagged independent variable	0.001*	omitted	omitted	-0.007***	omitted	omitted
(ref. contemporaneous ind. var.)	[0.0005]	omitieu	omilieu	[0.001]	omittea	omilieu
Dependent variable in levels	-0.445*	-0.391**	-0.872***	0.148***	0.152***	0.161***
(ref. first-differenced dep. var)	[0.254]	[0.171]	[0.170]	[0.003]	[0.003]	[0.003]
Controlling for endogeneity	-0.015**	-0.019**	-0.040***	0.029***	omitted	omitted
(ref. no IV approach)	[0.007]	[0.007]	[0.06]	[0.001]	omnieu	ommed
Wages in shares	2.864	omitted	omitted			
(ref. wages not in shares)	[2.778]	ommea	ommed			
Hours worked				0.0001	0.0001	0.0001
(ref. number of workers)				[0.297]	[0.297]	[0.297]
Publication characteristics (K-)						
Peer reviewed paper	1.546	omitted	omitted	-0.782**	-0.848***	-0.307**
(ref. working paper)	[1.799]	ommed	ommed	[0.280]	[0.135]	[0.135]
Peer reviewed paper * st. error	0.155	omitted	omitted	0.534**	omitted	omitted
	[0.396]	0 4 0 -		[0.246]		0.470.111
Publication year	-0.781*	-0.127	-0.934***	0.602***	0.676***	0.650***
	[0.456]	[0.220]	[0.228]	[0.091]	[0.073]	[0.073]
Publication year * st. error	-1.896***	-2.045***	-1.070***	-1.355***	-1.282***	-2.682***
	[0.123]	[0.108]	[0.154]	[0.099]	[0.096]	[0.651]
10-Year impact factor	-2.031	1.130**	0.713**	0.244	omitted	omitted
10 1000 000000	[2.158]	[0.354]	[0.345]	[0.548]		0
10-Year impact factor * st. error	-4.030**	-3.764**	-2.793**	0.881***	omitted	omitted
	[0.568]	[0.514]	[0.903]	[0.231]		
Geography (ref. US-UK-Canada-Japan)	0.075	• •	• 7	0.0000	• 7	• •
Europe	-0.375	omitted	omitted	-0.0008	omitted	omitted

	[0.379]			[0.011]		
Spatial unit (ref. national-level analysis)						
Regional – level analysis	0.114 [0.235]	omitted	omitted	0.492*** [0.082]	0.474*** [0.082]	0.769*** [0.082]
Development status (ref. not-specified)						
Developed	omitted	omitted	omitted	0.580 [0.355]	0.888** [0.346]	0.547 [0.346]
Developing	0.557*** [0.143]	0.575*** [0.142]	1.014*** [0.140]	0.198** [0.063]	0.211*** [0.036]	0.325*** [0.035]
Time (ref. midpoint before 1995)						
After 2001	-0.307* [0.158]	-0.279* [0.156]	-0.330** [0.154]	0.025 [0.060]	omitted	omitted
Other controls						
Dataspan	-0.349*** [0.174]	-0.349** [0.174]	-0.385** [0.175]	0.241** [0.073]	0.240** [0.073]	0.252** [0.073]
Number of RHS variables	1.1*10 ⁻⁶ [0.00009]	omitted	omitted	0.0008*** [0.0002]	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important controls or</i> <i>fixed effects</i>)	-0.049 [0.045]	omitted	omitted	-0.095*** [0.015]	-0.098*** [0.014]	-0.118*** [0.014]
Observations	231	231	231	392	392	392
χ^2 p-value	0.000***	0.000***	0.000***	0.000***	0.000***	0.000***
Primary study fixed effects	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes
Country by Midpoint fixed effects	yes	yes	yes	yes	yes	yes

Note: St. errors are reported in brackets. All models are weighted by the inverse of the variance of the collected elasticities. Columns 2 and 5 are based on the respective WLS-MRA models (Columns 1 and 4), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories is significant at the 10% level. * for p<0.05 and *** for p<0.01.

		Wage elasticities				Employment elasticities			
Variable	Fixed effects	Fixed effects general to specific	PEESE	Fixed effects general to specific - China	Fixed effects	Fixed effects general to specific	PEESE	Fixed effects general to specific - China	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
FAT – PET – PEESE tests									
Intercept (β_0) (PET) or (γ_0) (PEESE)	2.064** [0.957] 0.176	1.701** [0.853] 0.375***	0.004 [0.005] 0.013	0.009 [0.273] 0.752	-13.987*** [1.950] 0.963***	3.561** [1.296]	1.339 [1.354]	-2.312 [7.164] 5.068***	
St. error (β_1) (FAT) or (γ_1)	[0.113]	[0.082]	[0.013]	[0.545]	[0.154]	[0.062]	[0.051]	[1.061]	
Type of trade (ref. trade in final goods)									
Trade in intermediates	-0.002** [0.001]	-0.003** [0.001]	-0.003*** [0.0006]	-0.003* [0.002]	-0.040 [0.048]	omitted	omitted	omitted	
Data aggregation (ref. worker-level)									
Industry-level	-0.004** [0.002]	-0.009*** [0.001]	0.0003 [0.0008]	1.917** [0.913]	10.793*** [1.711]	omitted	omitted	omitted	
Firm-level	0.014*** [0.002]	0.016*** [0.001]	0.026*** [0.0008]	omitted	11.139*** [1.712]	omitted	omitted	omitted	
Sectoral Ievel (ref. economy-wide)									
Manufacturing	0.014 [0.009] -0.002	0.014 [0.009] -0.003	-0.001 [0.001] -0.016***	0.019** [0.009] -0.037	-0.149*** [0.027] -0.136**	-0.121*** [0.027] -0.099***	-0.088** [0.027] -0.060**	1.379 [2.093] 1.616	
Services	[0.009]	[0.009]	[0.001]	[0.071]	[0.027]	[0.027]	[0.027]	[2.093]	
Skill-level (ref. varied skill)		L]					L]		
Low-skill	0.006*** [0.001]	0.004** [0.001]	-0.002* [0.001]	-0.0001 [0.003]	-0.003** [0.001]	-0.003** [0.001]	-0.001 [0.001]	-0.085 [0.191]	
Medium-skill	0.007** [0.003]	0.005* [0.003]	-0.006** [0.002]	omitted	-0.040** [0.035]	-0.036 [0.035]	-0.028 [0.035]	omitted	
High-skill	0.015***	0.013***	0.005***	0.003	0.0006	0.002**	0.001	0.175	

Table E2.2. Labor market effects from trade (Dependent variable: e_i)

	[0.002]	[0.002]	[0.001]	[0.003]	[0.0008]	[0.0008]	[0.0008]	[0.195]
Other methodological aspects								
Technology and trade	0.0001	omitted	omitted	omitted	-0.001	omitted	omitted	omitted
(ref. only trade)	[0.002]				[0.002]			
(ref cross-sectional)	0.007**	omitted	omitted	omitted	-0.322***	omitted	omitted	omitted
I agged independent variable					0.0005			
(raf contamporaneous ind var)		omitted	omitted	omitted	-0.0003	omitted	omitted	omitted
(<i>rej. contemportaneous ind. vur.</i>)	0.018				[0.0004]			
(ref first-differenced dep var)	0.018 [0.027]	omitted	omitted	omitted	[0.004]	omitted	omitted	omitted
Controlling for endogeneity	0.007***				0.007***			
(ref. no IV approach)	[0.007	omitted	omitted	omitted	[0.007	omitted	omitted	omitted
(<i>vej. no iv approach</i>) Wages in shares	_2 987**	-2 613**	0 114***		[0.0000]			
(ref wages not in shares)	[1 027]	[0 899]	[0 001]	omitted				
Hours worked	[1.027]	[0.077]	[0.001]		0 272	0.275	0 199***	0 147
(ref number of workers)					[2 028]	[2 028]	[2 029]	[2 029]
Publication characteristics (K-)					[2.020]	[2.020]	[2.027]	[2.027]
Peer reviewed paper	-1.146**	-0.948**	0.013***	-1.945**	-0.180	-0.970	-0.411	1.386
(ref working paper)	[0.451]	[0.395]	[0 008]	[0.918]	[2,029]	[2.047]	[2.049]	[3,185]
(rej: working paper)	0.275	[0.090]	[0.000]	[0.910]	-1.556***	[[2:0:12]	[0.100]
Peer reviewed paper * st. error	[0.183]	omitted	omitted	omitted	[0,175]	omitted	omitted	omitted
	0.114**	0.130**	-0.009***	0.457**	0.181	5.048**	1.950	-1.293
Publication year	[0.052]	[0.048]	[0.0005]	[0.152]	[0.204]	[1.836]	[1.920]	[7.295]
	-0.011			•	1.195***	1.148***		5.603***
Publication year * st. error	[0.125]	omitted	omitted	omitted	[0.070]	[0.063]	omitted	[1.674]
10 Vereninger of forstern	1.851**	1.517**	0.0004	0.466**	-0.567		· · · · · · · · · · · · · · · · · · ·	
10-Year impact factor	[0.872]	[0.774]	[0.0004]	[0.183]	[0.431]	omittea	omittea	omittea
10 Veen immed featon * at amon	0.243**				-0.769***	:44 - 1		: 44 - 1
10- i ear impact factor * st. error	[0.110]	omitted	omitted	omitted	[0.124]	omitted	omitted	omitted
Geography (ref. US-UK-Canada-Japan	ı)							
Europa	-0.008**	-0.008*	-0.005***	-0.041	3.334***	omittad	omittad	omittad
Europe	[0.004]	[0.004]	[0.001]	[0.091]	[0.573]	ommed	omilied	omitted

Spatial unit (ref. national-level analysis	s)							
Regional – level analysis	0.018 [0.161]	omitted	omitted	omitted	2.829*** [0.467]	3.090*** [0.466]	2.780*** [0.464]	1.472** [0.498]
Development status (ref. not-specified)								
Developed	0.004*** [0.001]	0.008*** [0.001]	0.005*** [0.001]	omitted	-0.022*** [0.003]	omitted	omitted	omitted
Developing	-0.025*** [0.003]	-0.023*** [0.004]	-0.009*** [0.0008]	omitted	-0.026*** [0.004]	omitted	omitted	omitted
Time (ref. midpoint before 1995)								
After 2001	-0.915* [0.514]	-0.716* [0.432]	0.006*** [0.001]	-0.088 [0.150]	-0.109 [0.111]	-0.128 [0.111]	-0.098 [0.111]	0.262 [0.164]
Other controls								
Dataspan	-0.407** [0.118]	-0.408** [0.117]	0.007*** [0.0005]	-0.345** [0.122]	0.052 [0.124]	0.058 [0.124]	0.042 [0.124]	0.042 [0.159]
Number of RHS variables	0.0004** [0.0001]	omitted	omitted	omitted	-0.0004** [0.0002]	omitted	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important</i> <i>controls or fixed effects</i>)	0.015** [0.007]	0.018** [0.007]	0.015** [0.005]	0.015** [0.007]	0.008 [0.025]	omitted	omitted	omitted
Observations	459	459	459	64	635	635	635	60
χ^2 p-value	0.000***	0.000***	0.000***	0.000***	0.000***	0.000***	0.000***	0.000***
Primary study fixed effects	yes	yes	no	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	no	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	no	yes	yes	yes	yes	yes
Country x Midpoint fixed effects	yes	yes	no	yes	yes	yes	yes	yes

Note: St. errors are reported in brackets. All models are weighted by the inverse of the variance of the collected elasticities. Columns 2 and 6 are based on the respective weighted-least-squares models (Columns 1 and 5), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories are significant at the 10% level. * for p<0.1, ** for p<0.05 and *** for p<0.01.

Appendix F. Partial correlation coefficients. Theoretical principles and empirical results.

Appendix F1. Construction and summary statistics of partial correlation coefficients

Partial correlation coefficients (pcc's) are unitless statistical measures of the strength and the direction of association between two variables, holding other factors constant (Stanley & Doucouliagos, 2012). We calculate pcc's from the reported regression outcomes, based on the following formula (Eq. F1.1):

$$pcc = \frac{t}{\sqrt{t^2 + df}} \tag{F1}$$

where pcc is the partial correlation coefficient, t is the associated t-statistic of the reported regression outcome and df reports the degrees of freedom of the t-statistic. The standard error of the partial correlation coefficient is calculated based on Eq. (F2):

$$se_{pcc} = \sqrt{(1 - pcc^2)/df}$$
(F2)

Partial correlation coefficients are particularly useful since they allow the direct comparison between the reported outcomes from different studies; therefore, they enable the compilation of the most comprehensive datasets within a particular field of study. In contrast, they have two main drawbacks. First, they are just statistical and not economic measures of the investigated effects. Secondly, their distribution is not normal when their values are close to -1 and +1. Although this is hardly a problem for the current analysis, we follow Stanley and Doucouliagos (2012) and perform the Fischer's z-transformation, as illustrated in Eq. (F3):

$$r_{\rm z} = \frac{1}{2} ln \left(\frac{1+r}{1-r}\right) \tag{F3}$$

Furthermore, calculating partial correlation coefficients requires detailed information about the degrees of freedom of the estimated model in the primary study. Such information is rarely directly available from the authors and has to be approximated based on the total number of observations and the number of variables, including the various types of fixed effects. However, those calculations are not always accurate, due to lack of information about the total number of fixed effects groups included in the estimation. Therefore, the calculated partial correlation coefficients introduce 'noise' to our analysis.

The above drawbacks explain our choice to use elasticities in the main analysis and use the partial correlation coefficients in a supplementary way.

	Obs.	Mean	St. Dev.	IQR	Skewn.	Kyrt.	I ²	Q-test
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Technology pcc's	611	0.015	0.457	0.100	-10.67	210.2	100%	0.000***
• Wage pcc's	230	0.017	0.178	0.048	-0.982	13.68	99.9%	0.000***
• Employment pcc's	381	0.013	0.562	0.124	-9.175	147.3	100%	0.000***
Trade pcc's	1092	-0.005	0.152	0.052	-0.773	10.97	100%	0.000***
• Wage pcc's	457	0.005	0.124	0.026	0.531	10.19	100%	0.000***
• Employment pcc's	635	-0.012	0.169	0.077	-1.064	10.01	99.8%	0.000***

Table F1.1. Summary statistics

Note: Column 4 reports the interquartile range (the difference between the 75th and the 25th percentile). Column 7 reports the percentage of the between-estimate heterogeneity, besides the one attributed to sampling error. Column 8 reports the p-value for the χ^2 Cochran's test under the null hypothesis of no between-estimate heterogeneity. * for p < 0.1, ** for p < 0.05 and ** * for p < 0.0

	W	age elasticities	5	Employment elasticities			
Variable	WLS	WLS	PEESE	WLS	WLS	PEESE	
		general to			general to		
		specific	(2)		specific		
	(1)	(2)	(3)	(4)	(5)	(6)	
FAT – PET – PEESE tests		0.007	1 0 0 0	1.051	4 40011	0.000	
Intercept (β_0) (PET) or (γ_0) (PEESE)	11.592	-0.095	1.020	-1.354	1.400**	-0.038	
	[13.497]	[0.831]	[1.663]	[3.098]	[0.550]	[0.085]	
St error (β_{i}) or st error ² (γ_{i}) (FAT)	-9.129***	-7.645**	7.584	-5.939**	-13.905	0.091	
St. end (p_1) of st. end (p_1) (TT)	[2.050]	[2.080]	[6.081]	[2.552]	[3.167]	[0.120]	
Type of innovation (ref. process innovatio	n)						
D roduct innovation	0.020	omittad	omitted	-0.001	0.009	-0.032*	
FIOUUCT IIIIOVATION	[0.022]	omilieu	omitieu	[0.012]	[0.0008]	[0.017]	
Data aggregation (ref. worker-level)							
In deseting langel	0.114	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	0.154	-0.516	0.660***	
Industry-level	[0.162]	omittea	omittea	[0.608]	[0.425]	[0.063]	
	-5.693		• •	-3.851	-2.417**	-1.212***	
Firm-level	[6.879]	omitted	omitted	[4.781]	[0.716]	[0.144]	
Sectoral Ievel (ref. economv-wide)							
	-0.417	0.940	-1.635	0.020**	0.017**	-0.018	
Manufacturing	[0.433]	[0.880]	[2.105]	[0.009]	[0.008]	[0.021]	
	0 364	0.918	-1 673	0.054***	0.046***	0.030	
Services	[0 437]	[0.893]	[2,111]	[0 012]	[0 011]	[0.044]	
Skill-level (ref_varied skill)	[0.157]	[0.095]	[2,111]	[0:012]	[0.011]	[0.011]	
Skill-level (rej. varied skill)	0.032			-0 107***	-0.027	-0 571***	
Low-skill	[0.032 [0.029]	omitted	omitted	[0.028]	[0.027	[0 070]	
	$\begin{bmatrix} 0.027 \end{bmatrix}$			0.000**	0.017	0.524***	
Medium-skill	[0.079	omitted	omitted	-0.033	-0.017	[0.062]	
		amitta d	amitta d	[0.031]	[0.002]	[0.002]	
High-skill	0.072	omittea	omitted	-0.004	0.025	-0.308	

Table F2.1. Labor market effects from technology (Dependent variable: pcc_i)

	[0.054]			[0.015]	[0.051]	[0.009]
Other methodological aspects						
Technology and trade	-0.001	0.0006***	0.0006**	0.028***	:44 - 1	i44 - J
(ref. only technology)	[0.0008]	$[2.2*10^{-7}]$	$[2.2*10^{-7}]$	[0.005]	omittea	omittea
Time series	0.003	0.003	-0.009	0.604	-0.017	-0.088***
(ref. cross-sectional)	[0.003]	[0.002]	[0.008]	[0.778]	[0.011]	[0.001]
Lagged independent variable	-0.014**			-0.645	:44 - 1	: 44 - 1
(ref. contemporaneous ind. var.)	[0.004]	omitted	omittea	[0.766]	omittea	omittea
Dependent variable in levels	0.300	0.902	-1.780	0.606	0.016	0.064***
(ref. first-differenced dep. var)	[0.418]	[0.892]	[2.094]	[0.770]	[0.010]	$[2.7*10^{-6}]$
Controlling for endogeneity	-0.012	-0.014	-0.008	0.007	· · · · 1	0
(ref. no IV approach)	[0.019]	[0.021]	[0.011]	[0.005]	omittea	Omittea
Wages in shares	-11.379	• 1	•			
(ref. wages not in shares)	[14.285]	omitted	omitted			
Hours worked				-0.006	5.6*10 ⁻⁶	0.0002***
(ref. number of workers)				[0.016]	$[4.6*10^{-6}]$	$[1.7*10^{-13}]$
Publication characteristics (K-)						
Peer reviewed paper	-7.851	· · · · · · · · · · · · · · · · · · ·	· · · · 1	1.973	-0.315	0.606***
(ref. working paper)	[9.298]	omitted	omittea	[3.951]	[0.338]	[0.047]
D	-1.209	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	-5.522	1	1
Peer reviewed paper * st. error	[6.759]	omitted	omitted	[3.220]	omitted	omitted
	2.413	0.400	-0.851	-1.022	-1.564**	-1.431***
Publication year	[2.739]	[0.454]	[1.072]	[1.386]	[0.561]	[0.150]
	2.741	3.654	-3.329	-3.184	-1.218	0.721
Publication year * st. error	[4.544]	[2.208]	[4.662]	[3.125]	[3.281]	[1.850]
	9.576	1.018**	-0.452	-1.644**		• 1
10-Year impact factor	[10.998]	[0.419]	[0.877]	[0.583]	omitted	omitted
	-4.499	-8.776**	25.004	9.391	• 1	• •
10-Year impact factor * st. error	[7.167]	[4.211]	[16.254]	[5.941]	omitted	omitted
Geography (ref. US-UK-Canada-Japan)						
	-0.027	•,, 7	•,, 1	0.051***	• 1	· 1
Europe	[0.020]	omitted	omitted	[0.011]	omitted	omitted

Spatial unit (ref. national-level analysis)						
Regional – level analysis	-2.450 [2.649]	omitted	omitted	-6.469** [2.281]	-6.423*** [1.050]	0.600 [0.548]
Development status (ref. not-specified)						
Developed	omitted	omitted	omitted	0.006** [0.002]	0.007 [0.005]	0.009 [0.006]
Developing	0.003 [0.003]	0.004 [0.003]	0.003 [0.003]	0.003 [0.003]	0.008 [0.009]	0.011 [0.013]
Time (ref. midpoint before 1995)						
After 2001	0.0006 [0.018]	0.005 [0.022]	-0.003 [0.011]	0.276 [0.320]	omitted	omitted
Other controls						
Dataspan	-0.005*** [0.00005]	-0.005*** [0.0001]	-0.004*** [0.0006]	-1.810 [2.137]	-0.108** [0.031]	0.130*** [0.017]
Number of RHS variables	0.0006 [0.0004]	omitted	omitted	-2.7*10 ⁻⁷ [0.00004]	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important controls or</i> <i>fixed effects</i>)	-0.007 [0.007]	omitted	omitted	-0.195** [0.067]	-0.202*** [0.023]	-0.014 [0.022]
Observations	229	229	231	370	370	370
Adjusted R ²	0.533	0.502	0.456	0.999	0.999	0.995
Primary study fixed effects	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes
Country by Midpoint fixed effects	yes	yes	yes	yes	yes	yes

Note: Robust st. errors, clustered by primary study are reported in brackets. All models are weighted by the inverse of the variance of the partial correlation coefficients. Columns 2 and 5 are based on the respective WLS-MRA models (Columns 1 and 4), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories is significant at the 10% level. * for p<0.1, ** for p<0.05 and *** for p<0.01.

		Wage e	lasticities		Employment elasticities			
Variable	WLS	WL	PEESE	WLS	WLS	WLS	PEESE	WLS
		general to		general to		general to		general to
		specific		Specific - China		specific		Specific - China
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
FAT – PET – PEESE tests								
Intercept (β_0) (PET) or (γ_0) (PEESE)	-7.215 [4.504]	2.138 [1.755]	-0.015 [0.221]	-1.033 [0.591]	-5.402* [2.699]	7.719 [6.659]	2.787 [1.962]	0.597 [0.798]
St. error (β_1) (FAT) or (γ_1)	-13.119 [8.926]	-7.729 [7.250]	1.207 [0.893]	-3.738 [3.720]	-54.025** [19.156]	-23.194 [20.614]	-15.535 [11.776]	13.079 [22.173]
Type of trade (ref. trade in final goods)								
Trade in intermediates	-0.005 [0.006]	-0.001*** [0.0001]	-0.001*** [0.0002]	-0.001*** [5.4*10 ⁻⁶]	-0.004 [0.010]	omitted	omitted	omitted
Data aggregation (ref. worker-level)								
Industry-level	0.113 [0.071]	0.063 [0.039]	0.020** [0.008]	0.564 [0.410]	0.764 [0.461]	omitted	omitted	omitted
Firm-level	0.070 [0.063]	0.041 [0.033]	0.006 [0.007]	omitted	1.441** [0.687]	omitted	omitted	omitted
Sectoral Ievel (ref. economy-wide)								
Manufacturing	-0.276 [0.334]	-0.104 [0.312]	0.237*** [0.007] 0.231	0.115 [0.134]	0.607** [0.241] 0.555**	0.110 [0.111] 0.106	0.188 [0.154] 0.105	-0.223 [0.220]
Services	-0.282 [0.334]	[0.312]	[0.007]	[0.082]	[0.206]	[0.091]	[0.141]	[0.235]
Skill-level (ref. varied-skill)								
Low-skill	0.009 [0.006]	0.009 [0.007]	0.001 [0.001]	-0.006 [0.008]	-0.022** [0.010]	-0.018* [0.010]	-0.036** [0.011]	-0.022* [0.011]
Medium-skill	0.001	0.003	-0.002 [0.002]	omitted	0.017 [0.028]	-0.003 [0.024]	-0.031** [0.013]	omitted
High-skill	0.011* [0.006]	0.013 [0.008]	0.003** [0.001]	0.004 [0.016]	-0.020*** [0.005]	-0.010** [0.004]	-0.015** [0.007]	0.035** [0.010]

Table F2.2. Labor market effects from trade (Dependent variable: pcc_i)

Other methodological aspects								
Technology and trade	0.005	omitted	omitted	omitted	-0.062	omitted	omitted	omitted
(ref. only trade)	[0.004]	omitieu	omilieu	omilieu	[0.063]	omitieu	omilieu	omilieu
Time series	-0.002	omitted	omitted	omitted	-0.632**	omitted	omitted	omitted
(ref. cross – sectional)	[0.003]	omnea	ominea	omilieu	[0.309]	omitted	ominea	omnieu
Lagged independent variable	0.009	omitted	omitted	omitted	0.015	omitted	omitted	omitted
(ref. contemporaneous)	[0.006]	omnica	omitica	omineu	[0.012]	omnieu	omitica	omnica
Dependent variable in levels	-0.005	omitted	omitted	omitted	0.002	omitted	omitted	omitted
(ref. first-differenced dep. var.)	[0.006]	omnica	omned	omneu	[0.020]	omnieu	omnea	omnied
Controlling for endogeneity	-0.061	omitted	omitted	omitted	-0.244***	omitted	omitted	omitted
(ref. no-IV approach)	[0.054]	ommea	omned	ommed	[0.054]	ommed	ommed	ommed
Wages in shares	7.651	-1.158	-0.108	omitted				
(ref. wages not in shares)	[4.557]	[0.864]	[0.104]	ommed				
Hours worked					0.013	0.004	0.003	-0.001
(ref. number of workers)					[0.010]	[0.011]	[0.006]	[0.002]
Publication characteristics								
Peer reviewed paper	3.727	-0.627**	-0.359***	0.029	-6.864**	-0.869	-0.896**	-0.771***
(ref. working paper)	[2.439]	[0.188]	[0.057]	[0.128]	[2.692]	[0.517]	[0.433]	[0.086]
Peer reviewed paper * st_error	6.994	omitted	omitted	omitted	81.209**	omitted	omitted	omitted
reer tevrewed puper st. error	[8.727]	omnica	omned	omneu	[31.516]	omnieu	omnea	omnica
Publication year	0.536*	0.184	0.052**	0.069	0.792	11.202	3.440	0.588
i donoution your	[0.312]	[0.191]	[0.023]	[0.038]	[0.564]	[9.968]	[2.682]	[0.978]
Publication year * st_error	13.566*	omitted	omitted	omitted	-15.393	-0.045***	-16.143	-24.205
	[8.006]	ommed	onnicea		[9.819]	[0.010]	[11.795]	[32.088]
10-Year impact factor	-5.306	1.301	-0.131	-0.870	-6.549**	omitted	omitted	omitted
	[3.325]	[1.165]	[0.144]	[0.850]	[3.059]	ommou	ommed	ommed
10-Year impact factor * st. error	6.487	omitted	omitted	omitted	78.691**	omitted	omitted	omitted
	[4.647]			0	[32.530]	0		o
Geography (ref. non-OECD)	0.10.5	0.100	0.000	0.101				
Europe	-0.106	-0.120	-0.008	0.184	8.200**	omitted	omitted	omitted
	[0.066]	[0.089]	[0.011]	[0.185]	[3.232]			
Spatial unit (ref. national-level analysis	s)							

Regional – level analysis	0.190 [0.144]	omitted	omitted	omitted	0.026 [0.088]	0.176 [0.112]	0.213* [0.116]	-0.004 [0.032]
Development status (ref. not specified)								
Developed	0.020 [0.017]	0.002*** [0.0004]	0.001** [0.0003]	omitted	-0.013 [0.011]	omitted	omitted	omitted
Developing	0.010 [0.010]	-0.001** [0.0005]	-0.002** [0.0005]	omitted	-0.016* [0.009]	omitted	omitted	omitted
Time (ref. midpoint before 2001)								
After 2001	2.664 [1.726]	-0.468 [0.395]	-0.019 [0.048]	0.032 [0.062]	0.004 [0.069]	0.026 [0.016]	0.028 [0.018]	0.059 [0.076]
Other controls								
Dataspan	-0.027 [0.031]	-0.053** [0.023]	-0.078*** [0.0005]	-0.071*** [0.011]	0.016 [0.030]	0.0006 [0.014]	0.009 [0.005]	0.001** [0.0003]
Number of RHS variables	-0.0007 [0.0007]	omitted	omitted	omitted	-0.007 [0.006]	omitted	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important</i> controls or fixed effects)	0.914 [0.628]	-0.571 [0.533]	0.057 [0.070]	0.128** [0.038]	0.407 [0.332]	omitted	omitted	omitted
Observations	457	457	457	64	635	635	635	60
Adjusted R^2	0.665	0.548	0.545	0.575	0.540	0.325	0.169	0.767
Primary study fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes
Country x Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes

Note: Robust st. errors, clustered by primary study are reported in brackets. All models are weighted by the inverse of the variance of the partial correlation coefficients. Columns 2 and 5 are based on the respective weighted-least-squares models (Columns 1 and 4), following a general-to-specific approach, by sequentially eliminating the least significant explanatory variables, until at least one variable in the reported categories are significant at the 10% level. * for p<0.1, ** for p<0.01.

Appendix G. Collective robustness analysis (summary tables, only WLS general-to-specific estimation outcomes)

.	Table G1.	Wage	effects	from	techno	logy
----------	-----------	------	---------	------	--------	------

	Wage elasticities from technology										
Variable	WLS	WLS	WLS	WLS	WLS	WLS	WLS	Random	Partial		
	(main	trimming	winsorizing	excluding	excluding	excluding	excluding	effects	correlation		
	analysis)	outners	outners	semi-		in shares	(4)-(6) and winsorizing		coefficients		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)		
FAT – PET – PEESE tests											
Interport $(\boldsymbol{\theta})$ (DET) or (\boldsymbol{u}) (DEESE)	0.551**	0.421*	0.437*	0.418	0.206*	3.909	-0.046***	3.109	-0.095		
Intercept (p_0) (PE1) or (γ_0) (PEESE)	[0.161]	[0.212]	[0.212]	[0.348]	[0.115]	[2.661]	[0.008]	[3.833]	[0.831]		
St error (β) (EAT) or (γ)	-1.474**	-1.936**	-1.237*	-0.963**	-1.653**	-2.181***	0.073	2.365	-7.645**		
St. end (p_1) (PAT) of (p_1)	[0.567]	[0.824]	[0.648]	[1.141]	[0.550]	[0.309]	[0.135]	[3.702]	[2.080]		
Type of trade (ref. trade in final goods)											
Product innovation	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted		
Data aggregation (ref. worker-level)											
Industry-level	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted		
Firm-level	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted		
Sectoral Ievel (ref. economy-wide)											
Manufacturing	-0.225*	-0.287	-0.268*	-0.402	-0.212*	-0.185*	0.785***	-0.068	0.940		
	[0.111]	[0.200]	[0.133]	[0.348]	[0.106]	[0.104]	[0.186]	[0.136]	[0.880]		
Services	-0.218*	-0.279	-0.259*	-0.395	0.134	-2.765	omitted	-0.062	0.918		
	[0.112]	[0.201]	[0.133]	[0.348]	[0.423]	[2.403]		[0.182]	[0.893]		
Skill-level (ref. varied-skill)											
Low-skill	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted		
Medium-skill	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted		
High-skill	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted		
Other methodological aspects	0.000**	0.000**	0.00 (**	0.001	0.007**	0.007**	0.540**	0.025**	0.000 6 * * *		
rechnology and trade	-0.008**	-0.009**	0.006**	-0.001	-0.00/**	-0.00/**	-0.542**	-0.035**	0.0006^{***}		
(ref. only trade)	[0.002]	[0.004]	[0.002]	[0.087]	[0.002]	[0.002]	[0.169]	[0.009]	$[2.2*10^{7}]$		
Time series	-0.010*	-0.013**	-0.013**	-0.012**	-0.006	-0.00/	-0.011***	0.159	0.003		

(ref. cross – sectional)	[0.005]	[0.005]	[0.006]	[0.005]	[0.006]	[0.006]	[0.005]	[0.558]	[0.002]
Lagged independent variable (<i>ref. contemporaneous</i>)	omitted	omitted	omitted						
Dependent variable in levels	-0.391***	-0.462**	-0.392**	-0.070*	-0.015	-2.906	-0.525**	-0.159	0.902
(ref. first-differenced dep. var.)	[0.086]	[0.176]	[0.120]	[0.039]	[0.383]	[2.354]	[0.125]	[0.096]	[0.892]
Controlling for endogeneity	-0.019**	-0.002	-0.025***	-0.078	-0.016**	-0.012**	-1.014**	-0.125	-0.014
(ref. no-IV approach)	[0.005]	[0.002]	[0.005]	[0.067]	[0.004]	[0.003]	[0.271]	[0.117]	[0.021]
Wages in shares (ref. wages not in shares)	omitted	omitted	omitted	omitted	omitted	omitted	-0.542** [0.169]	omitted	omitted
Publication characteristics									
Peer reviewed paper (<i>ref. working paper</i>)	omitted	omitted	omitted						
Peer reviewed paper * st. error	omitted	omitted	omitted						
Dublication year	-0.127	-0.298	0.088	-0.018	-0.283	-0.399	0.183**	23.720	0.400
Fublication year	[0.389]	[0.405]	[0.361]	[0.038]	[0.405]	[0.442]	[0.060]	[30.191]	[0.454]
Publication year * st arror	-2.045***	-1.427*	-1.273**	-1.997***	-1.958***	-1.663***	-0.055**	-0.270	3.654
r ublication year st. error	[0.423]	[0.724]	[0.583]	[0.442]	[0.435]	[0.295]	[0.021]	[0.558]	[2.208]
10-Vear impact factor	1.130**	0.686	0.118	-0.160	0.770*	8.242	-0.779	-16.502	1.018**
10-1 car impact factor	[0.478]	[0.654]	[0.485]	[0.112]	[0.396]	[5.850]	[0.182]	[20.023]	[0.419]
10-Year impact factor * st_error	-3.764**	-3.485*	-1.843	-1.944	-4.997**	-5.781**	0.905**	1.849	-8.776**
To Tear impact factor st. error	[1.407]	[1.973]	[1.563]	[2.044]	[1.383]	[1.593]	[0.277]	[4.042]	[4.211]
Geography (ref. non-OECD)			_						
Europe	omitted	omitted	omitted						
Spatial unit (ref. national-level analysis	•)								
Regional – level analysis	omitted	omitted	omitted						
Development status (ref. not specified)	• •	• •		• 1	• •	•	• 1	• •	•
Developed	omitted	omitted	omitted						
Developing	0.5/5***	0.631***	0.405***	omitted	0.362***	0.541***	omitted	0.435	0.004
Time (ref midnoint hefore 2001)	[0.005]	[0.126]	[0.069]		[0.001]	[0.000]		[0.319]	[0.005]
	-0 279***	-0 292***	0.083**	0.211	-0 284***	-0.286***		-0 473*	0.005
After 2001	[0.047]	[0.044]	[0.022]	[0.311]	[0.053]	[0.055]	omitted	[0.263]	[0.022]

Other controls									
Detector	-0.349***	-0.354***	0.036**	-0.002	-0.348***	-0.347***	-0.114***	-0.596***	-0.005***
Dataspan	[0.008]	[0.014]	[0.009]	[0.002]	[0.008]	[0.007]	[0.021]	[0.032]	[0.0001]
Number of RHS variables	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Including important controls or fixed									
effects (ref. missing important	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
controls or fixed effects)									
Observations	231	193	231	163	153	124	58	231	229
Adjusted R^2	0.622	0.634	0.676	0.409	0.723	0.784	0.616	0.368	0.502
Primary study fixed effects	yes	yes	yes	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes	yes
Country x Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes	yes

Note: All columns (except Column 8) report WLS-MRA models, following the general-to-specific approach. Column 1 reports the main analysis outcomes. Column 2 trims the first and last 5 percentiles of the distributions of the elasticities and the st. errors to remove outliers. Column 3 winsorizes the first and the last 20 percentiles of the distribution of the elasticities. Column 4 excludes semi-elasticities (from log-linear models), Column 5 excludes calculated elasticities (as reported in the primary studies), Column 6 excludes elasticities expressed in shares. Column 7 excludes the elasticities from Columns 4 to 6 and winsorizes the first and last 20 percentiles of the remaining sample. Column 7 reports the random effects model (fixed effects model is not reported here for brevity since the point estimates are the same with the WLS-MRA in Column 1). Column 8 reports the WLS-MRA estimation results based on partial correlation coefficients. All models are weighted by the inverse of the variance of the elasticities. * for p<0.1, ** for p<0.05 and *** for p<0.01.

 Table G2. Employment effects from technology

	Employment elasticities from technology										
Variable	WLS (main analysis)	WLS trimming outliers	WLS winsorizing outliers	WLS excluding semi-	WLS excluding calculated	WLS excluding elasticities in shares	WLS excluding (4)-(6) and winsorizing	Random effects	Partial correlation coefficients		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)		
FAT – PET – PEESE tests											
Intercept (β_0) (PET) or (γ_0) (PEESE)	0.706* [0.358]	-0.298 [0.265]	0.125*** [0.032]	0.367 [0.426]	0.221 [0.239]	-0.503 [0.525]	0.462*** [0.105]	0.159 [0.123]	1.400** [0.550]		
St. error (β_1) (FAT) or (γ_1)	-1.296 [0.951]	-0.544 [0.740]	0.170 [0.252]	-0.140 [1.611]	-1.344 [1.004]	0.080 [0.956]	0.834** [0.334]	-0.158 [0.353]	-13.905 [3.167]		
Type of trade (ref. process innovation)											
Product innovation	0.001** [0.0005]	0.003** [0.001]	0.001*** [0.0002]	0.001** [0.0005]	0.001** [0.0005]	0.001** [0.0004]	0.001*** [0.0002]	0.282 [0.201]	0.009 [0.0008]		
Data aggregation (ref. worker-level)											
Industry-level	0.167** [0.078] 1.117***	0.184** [0.074] 0.985***	-0.021 [0.029] 0.239***	0.142* [0.080] 0.413*	0.067 [0.126] 0.601*	-1.024** [0.295] -0.125*	-0.435*** [0.060] -0.029***	2.217** [0.611] 6.811***	-0.516 [0.425] -2.417**		
Firm-level	[0.066]	[0.062]	[0.024]	[0.224]	[0.345]	[0.072]	[0.005]	[1.001]	[0.716]		
Sectoral Ievel (ref. economy-wide)											
Manufacturing	0.221 [0.139] 0.068**	0.243 [0.161] 0.104***	0.075 [0.048] 0.031***	0.005 [0.011] 0.027	0.221 [0.140] 0.068***	0.207 [0.142] 0.045**	0.004 [0.002] 0.015	-0.031 [0.516] 0.258	0.017** [0.008] 0.046***		
Services	[0.013]	[0.019]	[0.004]	[0.030]	[0.013]	[0.016]	[0.010]	[0.177]	[0.011]		
Skill-level (ref. varied-skill)											
Low-skill	0.050** [0.023]	0.093*** [0.022]	0.016 [0.017]	-0.122 [0.332]	0.038 [0.038]	0.004 [0.037]	-0.066*** [0.009]	-1.232 [0.962]	-0.027 [0.065]		
Medium-skill	0.060** [0.018]	0.108*** [0.021]	0.033** [0.011]	-0.111 [0.325]	-0.093 [0.071]	omitted	omitted	-1.172 [0.970]	-0.017 [0.062]		
High-skill	0.187*** [0.014]	0.225*** [0.023]	0.139*** [0.005]	0.013 [0.339]	0.189*** [0.011]	0.170*** [0.015]	-1.094** [0.473]	2.212** [0.956]	0.023 [0.051]		

Other methodological aspects									
Technology and trade (ref. only trade)	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Time series	-0.146***	0.780***	-0.045***	-0.159***	-0.146***	-0.290**	-0.132***	-0.443***	-0.088***
(ref. cross – sectional)	[0.008]	[0.222]	[0.002]	[0.014]	[0.009]	[0.108]	[0.005]	[0.032]	[0.001]
Lagged independent variable (ref. contemporaneous)	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Dependent variable in levels	0.152***	0.048***	0.046***	0.160***	0.151***	0.292**	0.132**	0.450***	0.064***
(ref. first-differenced dep. var.)	[0.006]	[0.003]	[0.001]	[0.011]	[0.006]	[0.110]	[0.005]	[0.032]	$[2.7*10^{-6}]$
Controlling for endogeneity (ref. no-IV approach)	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Hours worked	0.0001***	0.0001***	0.0001***	0.0001***	0.0001***	0.0001***	0.0001***	0.0004***	0.0002***
(ref. number of workers)	[6.80*10 ⁻¹¹]	$[2.4*10^{-10}]$	$[3.5*10^{-12}]$	[6.8*10 ⁻⁸]	[6.5*10 ⁻⁸]	$[6.7*10^{-10}]$	$[1.6*10^{-12}]$	$[2.5*10^{-14}]$	$[1.7*10^{-13}]$
Publication characteristics									
Peer reviewed paper	-0.848**	-0.771***	-0.161***	-0.029	-0.241	0.708	-0.424***	-2.460***	0.606***
(ref. working paper)	[0.406]	[0.178]	[0.005]	[0.297]	[0.315]	[0.458]	[0.048]	[0.600]	[0.047]
Peer reviewed paper * st. error	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Publication year	0.676***	0.614***	0.187***	0.208	0.290	-0.041	-0.128***	1.663**	-1.431***
r doneation year	[0.021]	[0.030]	[0.015]	[0.414]	[0.232]	[0.057]	[0.003]	[0.591]	[0.150]
Publication year * st error	-1.282**	-1.212*	-0.378*	0.052	-1.258**	-0695	0.098	-0.444	0.721
r doneation year st. error	[0.565]	[0.659]	[0.194]	[0.871]	[0.569]	[0.742]	[0.194]	[1.682]	[1.850]
10-Year impact factor	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
10-Year impact factor * st. error	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Geography (ref. non-OECD)									
Europe	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Spatial unit (ref. national-level analysis	s)								
Regional – level analysis	0.474** [0.173]	-0.171 [0.208]	0.068 [0.050]	-0.392 [0.360]	0.472** [0.178]	0.933** [0.415]	-0.424*** [0.048]	0.582 [0.436]	0.600 [0.548]
Development status (ref. not specified)									
Developed	0.888*** [0.202]	omitted	-0.018 [0.052]	omitted	0.897*** [0.213]	0.580** [0.216]	omitted	0.349 [0.234]	0.009 [0.006]
Developing	0.211**	0.228**	0.055**	2.740*	0.211**	-0.020	-0.177	0.606*	0.011

	[0.072]	[0.065]	[0.017]	[1.491]	[0.074]	[0.164]	[0.402]	[0.347]	[0.013]
Time (ref. midpoint before 2001)									
After 2001	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Other controls									
Dataspan	0.240*** [0.006]	0.256*** [0.005]	0.012*** [0.001]	-0.004 [0.010]	0.241*** [0.005]	0.017 [0.015]	0.032*** [0.004]	1.365*** [0.081]	0.130*** [0.017]
Number of RHS variables	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Including important controls or fixed effects (<i>ref. missing important</i> controls or fixed effects)	-0.098** [0.038]	-0.099* [0.051]	-0.011* [0.006]	-0.214 [0.255]	-0.098** [0.038]	-0.006 [0.034]	0.109** [0.052]	-0.542** [0.207]	-0.014 [0.022]
Observations	392	328	392	285	355	276	198	392	370
Adjusted R^2	0.985	0.624	0.944	0.994	0.985	0.229	0.817	0.655	0.995
Primary study fixed effects	yes	yes	yes	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	no	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	no	yes	yes	yes	yes	yes
Country x Midpoint fixed effects	ves	ves	ves	no	ves	ves	ves	ves	ves

Country x Midpoint fixed effectsyesyesyesyesyesyesyesyesyesyesNote: All columns (except Column 8) report WLS-MRA models, following the general-to-specific approach. Column 1 reports the main analysis outcomes. Column 2 trims the first and last 5 percentiles of the distributions
of the elasticities and the st. errors to remove outliers. Column 3 winsorizes the first and the last 20 percentiles of the distribution of the elasticities. Column 4 excludes semi-elasticities (from log-linear models), Column
5 excludes calculated elasticities (as reported in the primary studies), Column 6 excludes elasticities expressed in shares. Column 7 excludes the elasticities from Columns 4 to 6 and winsorizes the first and last 20
percentiles of the remaining sample. Column 7 reports the random effects model (fixed effects model is not reported here for brevity since the point estimates are the same with the WLS-MRA in Column 1). Column 8
reports the WLS-MRA estimation results based on partial correlation coefficients. All models are weighted by the inverse of the variance of the elasticities. * for p<0.1, ** for p<0.05 and *** for p<0.01.</th>

Table G3. Wage effects from trade

	Wage elasticities from trade										
Variable	WLS (main analysis)	WLS trimming outliers	WLS winsorizing outliers	WLS excluding semi-	WLS excluding calculated	WLS excluding elasticities	WLS excluding (4)-(6) and winceriging	Random effects	Partial correlation coefficients		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)		
FAT – PET – PEESE tests											
Intercept (β_0) (PET) or (γ_0) (PEESE)	1.701** [0.665]	1.792* [0.923]	1.944** [0.470]	3.825*** [0.411]	1.893*** [0.275]	1.893*** [0.275]	-1.779** [0.709]	0.004 [0.066]	2.138 [1.755]		
St. error (β_1) (FAT) or (γ_1)	0.375 [0.310]	0.430 [0.443]	0.224 [0.221]	0.635 [0.477]	0.498 [0.399]	0.498 [0.401]	0.787* [0.431]	0.322 [1.196]	-7.729 [7.250]		
Type of trade (ref. trade in final goods)											
Trade in intermediates	-0.002*** [0.0006]	-0.003*** [0.0006]	-0.003*** [0.0005]	-0.003** [0.0008]	-0.003*** [0.0006]	-0.003*** [0.0007]	-0.002** [0.0009]	-0.184 [0.173]	-0.001*** [0.0001]		
Data aggregation (ref. worker-level)											
Industry-level	-0.009 [0.007]	-0.002 [0.003]	-0.009 [0.007]	-0.010 [0.007]	-0.009 [0.007]	-0.009 [0.007]	-0.010 [0.007]	0.018 [0.364]	0.063 [0.039]		
Firm-level	0.016** [0.006]	0.015 [0.019]	0.016** [0.006]	[0.015** [0.007]	[0.016**	[0.007]	0.015* [0.007]	0.198 [0.488]	[0.033]		
Sectoral Ievel (ref. economy-wide)											
Manufacturing	0.014*** [0.001]	0.014*** [0.002]	0.014*** [0.0009]	3.389*** [0.397]	0.013*** [0.002]	0.013*** [0.002]	0.355** [0.141]	0.305** [0.098]	-0.104 [0.312]		
Services	-0.003 [0.009]	-0.003 [0.010]	-0.002 [0.009]	3.362*** [0.394]	-0.003 [0.010]	-0.004 [0.010]	0.328** [0.139]	-0.039 [0.189]	-0.109 [0.312]		
Skill-level (ref. varied-skill)											
Low-skill	0.004 [0.001]	0.004 [0.004]	0.004 [0.004]	0.007 [0.005]	0.004 [0.004]	0.004 [0.004]	0.006 [0.005]	0.025 [0.118]	0.009 [0.007]		
Medium-skill	0.005	0.004 [0.004]	0.005	0.006 [0.004]	0.004 [0.003]	0.004 [0.003]	0.005 [0.004]	-0.502 [0.348]	0.003 [0.004]		
High-skill	0.013** [0.004]	0.012** [0.004]	0.012** [0.003]	0.015** [0.005]	0.013** [0.004]	0.013** [0.004]	0.014** [0.004]	0.334 [0.240]	0.013 [0.008]		

Other methodological aspects									
Technology and trade (<i>ref. only trade</i>)	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Time series (<i>ref. cross – sectional</i>)	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Lagged independent variable (ref. contemporaneous)	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Dependent variable in levels (ref. first-differenced dep. var.)	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Controlling for endogeneity (ref. no-IV approach)	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Wages in shares (ref. wages not in shares)	-2.613** [0.867]	-2.657** [1.903]	-2.261** [0.611]	-3.040*** [0.499]	0.670*** [0.108]	omitted	0.006 [0.005]	-5.263** [2.054]	-1.158 [0.864]
Publication characteristics	[]	[]		[]	[]		[]	[]	[]
Peer reviewed paper	-0.948**	-1.002**	-0.961**	-3.293***	-1.040***	-1.041***	0.035*	3.124	-0.627**
(ref. working paper)	[0.325]	[0.447]	[0.229]	[0.391]	[0.142]	[0.142]	[0.017]	[6.405]	[0.188]
Peer reviewed paper * st. error	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Publication year	0.130*** [0.005]	0.136*** [0.008]	-0.003 [0.005]	1.847*** [0.219]	0.133*** [0.003]	0.133*** [0.013]	0.108** [0.046]	0.450* [0.255]	0.184 [0.191]
Publication year * st. error	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
10-Year impact factor	1.517** [0.598]	1.579* [0.835]	1.739*** [0.423]	1.108*** [0.135]	1.689*** [0.242]	1.689*** [0.244]	-2.406** [0.934]	8.076 [11.215]	1.301 [1.165]
10-Year impact factor * st. error	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Geography (ref. non-OECD)									
Europe	-0.008*** [0.002]	-0.008** [0.003]	-0.009*** [0.001]	-0.047** [0.021]	-0.007** [0.002]	-0.007** [0.002]	0.034 [0.021]	0.131 [0.092]	-0.120 [0.089]
Spatial unit (ref. national-level analysis	s)								
Regional – level analysis	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Development status (ref. not specified)									
Developed	0.008** [0.004]	0.008** [0.004]	0.008** [0.004]	0.012*** [0.003]	0.008** [0.004]	0.008** [0.004]	0.010** [0.002]	0.117 [0.354]	0.002*** [0.0004]
Developing	-0.023	-0.025	-0.008**	-0.022	-0.023	-0.023	-0.008**	0.118	-0.001**

	[0.018]	[0.019]	[0.004]	[0.024]	[0.018]	[0.018]	[0.002]	[0.392]	[0.0005]
<i>Time (ref. midpoint before 2001)</i>									
After 2001	-0.716**	-0.759	-0.959**	-5.690***	-0.811***	-0.812***	0.290**	5.562	-0.468
	[0.328]	[0.466]	[0.230]	[0.662]	[0.138]	[0.138]	[0.107]	[6.317]	[0.395]
Other controls									
Dataanan	-0.408***	-0.412***	-0.030**	-0.004	-0.416***	-0.416***	0.002	-1.974***	-0.053**
Dataspan	[0.019]	[0.027]	[0.014]	[0.006]	[0.024]	[0.024]	[0.001]	[0.151]	[0.023]
Number of RHS variables	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Including important controls or fixed	0 018***	0 018***	0.017***	0.079*	0 018***	0 018***	0.064	0 415**	-0 571
effects (ref. missing important	[0.002]	[0.002]	[0.008]	[0.041]	[0.002]	[0.002]	[0.037]	[0.203]	[0.533]
controls or fixed effects)	[0:002]	[0:002]	[0:000]	[0:011]	[0:002]	[0:002]	[0:007]	[0:200]	[0.000]
Observations	459	390	459	306	384	378	225	459	457
Adjusted R^2	0.906	0.331	0.926	0.905	0.912	0.912	0.922	0.190	0.548
Primary study fixed effects	yes	yes	yes	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	yes	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes	yes
Country x Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes	yes

Note: All columns (except Column 8) report WLS-MRA models, following the general-to-specific approach. Column 1 reports the main analysis outcomes. Column 2 trims the first and last 5 percentiles of the distributions of the elasticities and the st. errors to remove outliers. Column 3 winsorizes the first and the last 20 percentiles of the distribution of the elasticities. Column 4 excludes semi-elasticities (from log-linear models), Column 5 excludes calculated elasticities (as reported in the primary studies), Column 6 excludes elasticities expressed in shares. Column 7 excludes the elasticities from Columns 4 to 6 and winsorizes the first and last 20 percentiles of the remaining sample. Column 7 reports the random effects model (fixed effects model is not reported here for brevity since the point estimates are the same with the WLS-MRA in Column 1). Column 8 reports the WLS-MRA estimation results based on partial correlation coefficients. All models are weighted by the inverse of the variance of the elasticities. * for p<0.1, ** for p<0.05 and *** for p<0.01.

Table G4. Employment effects from trade

	Employment elasticities from trade										
Variable	WLS (main analysis)	WLS trimming outliers	WLS winsorizing outliers	WLS excluding semi-	WLS excluding calculated	WLS excluding elasticities	WLS excluding (4)-(6) and	Random effects	Partial correlation coefficients		
	(1)	(2)	(3)	(4)	(5)	in snares (6)	(7)	(8)	(9)		
FAT – PET – PEESE tests											
Intercept (β_0) (PET) or (γ_0) (PEESE)	3.561** [1.558]	-0.105*** [0.019]	0.689 [0.710]	0.068 [1.579]	-0.512*** [0.057]	-0.136*** [0.26]	0.046** [0.012]	-0.082 [0.199]	7.719 [6.659]		
St. error (β_1) (FAT) or (γ_1)	-0.058 [0.314]	-0.141 [0.313]	0.086 [0.210]	0.332 [0.299]	0.010 [0.297]	0.077 [0.325]	0.318 [0.267]	-0.623 [0.715]	-23.194 [20.614]		
Type of trade (ref. trade in final goods)											
Trade in intermediates	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted		
Data aggregation (ref. worker-level)											
Industry-level	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted		
Sectoral level (ref. economy_wide)	omitiea	omilieu	omitieu	omittea	omilieu	omitieu	omnea	omilieu	ommed		
Manufacturing	-0.121*** [0.020]	-0.095*** [0.014]	-0.101*** [0.012]	-0.141*** [0.024]	-0.123*** [0.019]	-0.131*** [0.023]	-0.114*** [0.017]	-0.449** [0.127]	0.110 [0.111]		
Services	-0.099 [0.019]	-0.071** [0.020]	-0.089*** [0.014]	-0.121*** [0.022]	-0.101*** [0.018]	-0.109*** [0.022]	-0.102*** [0.018]	-0.003 [0.168]	0.106 [0.091]		
Skill-level (ref. varied-skill)											
Low-skill	-0.003* [0.001]	-0.015** [0.0008]	-0.001*** [0.0002]	-0.004** [0.001]	-0.003** [0.001]	-0.003** [0.001]	-0.002*** [0.0003]	-0.455* [0.247]	-0.018* [0.010]		
Medium-skill	-0.036** [0.016]	-0.043** [0.011]	-0.004 [0.008]	-0.041** [0.015]	-0.044*** [0.007]	-0.046*** [0.008]	-0.013** [0.005]	-0.568 [0.468]	-0.003 [0.024]		
High-skill	0.002** [0.0006]	-0.005** [0.001]	0.002*** [0.0001]	0.002** [0.0005]	0.002** [0.0006]	0.002** [0.0006]	0.002*** [0.0002]	0.264 [0.207]	-0.010** [0.004]		
Other methodological aspects											
Technology and trade (ref. only trade)	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted		

Time series	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Lagged independent variable									
(ref. contemporaneous)	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Dependent variable in levels	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
(ref. first-differenced dep. var.)	ommed	omilieu	omnica	ominea	omineu	omnieu	omnea	omnica	omittea
Controlling for endogeneity	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
(ref. no-IV approach)	0 275***	0 05 8***	0.060	0 5 9 1	0 202***	0 205***	0 179***	0 115***	0.004
(ref number of workers)	[0.273****	-0.038****	0.000	-0.381	0.282	0.303	-0.178	-0.113	0.004
Publication characteristics	[0.041]	[0.001]	[0.037]	[2.780]	[0.039]	[0.049]	[0.008]	[0.000]	[0.011]
Peer reviewed naper	-0.970**	0 213***	-0 134	0.531	-0 079**	-0 189***	0 243***	-0.161	-0.869
(ref working paper)	[0 337]	[0 027]	[0 146]	[2 427]	[0.023]	[0 041]	[0 005]	[0 288]	[0 517]
Peer reviewed paper * st error	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
i con icvictived puper st. enter	5 048**	-0.123***	1 009	0 169	0 746***	-0 234***	0.085***	-2.492	11 202
Publication year	[2.214]	[0.023]	[1.013]	[2.234]	[0.088]	[0.032]	[0.007]	[1.574]	[9,968]
Publication year * st. error	1.148**	0.403	0.596**	1.470***	1.148**	1.337**	0.819**	0.249	-0.045***
	[0.326]	[0.307]	[0.239]	[0.344]	[0.325]	[0.375]	[0.303]	[0.216]	[0.010]
10-Year impact factor	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
10-Year impact factor * st. error	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Geography (ref. US-UK-Canada)									
Europe	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Spatial unit (ref. national-level analysis	s)								
Regional level analysis	3.090***	0.433***	0.483**	0.754	3.129***	2.907***	0.137***	3.520***	0.176
Regional – level analysis	[0.197]	[0.047]	[0.173]	[1.042]	[0.188]	[0.187]	[0.011]	[0.330]	[0.112]
Development status (ref. not specified)									
Developed	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Developing	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Time (ref. midpoint before 2001)									
After 2001	-0.128***	-0.091***	-0.007	-0.552	-0.131***	-0.140***	-0.216***	-0.592***	0.026
	[0.016]	[0.014]	[0.014]	[3.473]	[0.014]	[0.018]	[0.010]	[0.117]	[0.016]
Other controls									

Dataspan	0.058***	0.056***	0.011	-0.508***	-0.060***	0.064***	-0.018***	-0.055	0.0006
	[0.008]	[0.010]	[0.007]	[0.015]	[0.008]	[0.010]	[0.004]	[0.040]	[0.014]
Number of RHS variables	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
Including important controls or fixed									
effects (ref. missing important	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted	omitted
controls or fixed effects)									
Observations	635	533	635	471	590	557	408	635	635
Adjusted R^2	0.992	0.504	0.996	0.992	0.992	0.993	0.997	0.261	0.325
Primary study fixed effects	yes	yes	yes	yes	yes	yes	yes	yes	yes
Country fixed effects	yes	yes	yes	no	yes	yes	yes	yes	yes
Midpoint fixed effects	yes	yes	yes	yes	yes	yes	yes	yes	yes
Country x Midpoint fixed effects	yes	yes	yes	no	yes	yes	yes	yes	yes

Note: All columns report WLS-MRA models, following the general-to-specific approach. Column 1 reports the main analysis outcomes. Column 2 trims the first and last 5 percentiles of the distributions of the elasticities and the st. errors to remove outliers. Column 3 winsorizes the first and the last 20 percentiles of the distribution of the elasticities. Column 4 excludes semi-elasticities (from log-linear models), Column 5 excludes calculated elasticities (as reported in the primary studies), Column 6 excludes elasticities expressed in shares. Column 7 excludes the elasticities from Columns 4 to 6 and winsorizes the first and last 20 percentiles of the remaining sample. Column 7 reports the random effects model is not reported here for brevity since the point estimates are the same with the WLS-MRA in Column 1). Column 8 reports the WLS-MRA estimation results based on partial correlation coefficients. All models are weighted by the inverse of the variance of the elasticities. * for p<0.0, ** for p<0.01.

Appendix H. Best practice estimates

Sample	Reasoning	Outcome
Wage elasticities from technology	Constant + time series analysis + first-differenced dependent variable + model includes both technology and trade + estimation method addresses endogeneity + more recent estimate (after 2001)	-0.104
Employment elasticities from technology	Constant + analysis at the macro- level + firm-level evidence + varied skill-level of the workforce + time series analysis + first differenced dependent variable + elasticities reported in peer-reviewed journals + the estimated model includes important control variables	0.796
Wage elasticities from trade	Constant + trade refers to both intermediate and final products + manufacturing sector + firm level analysis + in Europe + elasticities reported in peer-reviewed journals	0.775
Employment elasticities from trade	Constant + manufacturing sector + more recent effect (after 2001) + elasticities published in peer-reviewed journals	2.167

 Table G1. Best practice estimates

Reference list

Stanley, T. D., & Doucouliagos, H. (2012). *Meta-regression analysis in economics and business*. Routledge Advances in Research Methods.